UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, DC 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of report (Date of earliest event reported) April 22, 2005

Matrix Service Company

(Exact Name of Registrant as Specified in Its Charter)

Delaware

(State or Other Jurisdiction of Incorporation)

001-15461 (Commission File Number) 73-1352174 (IRS Employer Identification No.)

74116 (Zip Code)

10701 E. Ute Street Tulsa, Oklahoma (Address of Principal Executive Offices)

918-838-8822

(Registrant's Telephone Number, Including Area Code)

Not Applicable

(Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):			
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)		
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)		
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))		
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))		

Item 1.01 Entry into a Material Definitive Agreement.

Securities Purchase Agreement

On April 22, 2005, Matrix Service Company (the "Company") completed a private placement of \$30 million aggregate principal amount of its 7% Senior Unsecured Convertible Notes due 2010 (the "Notes").

The Notes were issued under a Securities Purchase Agreement (the "Securities Purchase Agreement") among the Company and certain investors (including any transferees or assignees thereof, the "Investors"). The Securities Purchase Agreement and form of note, which is attached as an exhibit to the Securities Purchase Agreement, provide, among other things, that the Notes will bear interest at a rate of 7% per year. Interest is payable in advance on April 25, 2005 for the period to and including April 25, 2007, which advance payment is \$4.2 million. After April 25, 2007, interest is payable in arrears on each March 31, June 30, September 30 and December 31 (each, an "Interest Payment Date"), beginning on June 30, 2007, at the rate of 7% per year. If the Company maintains a specified leverage ratio of total debt to EBITDA (as those terms are defined in the Securities Purchase Agreement) and no default or event of default exists on the Interest Payment Date, the interest rate will be 5% per annum after April 25, 2007. If, however, the Company fails to refinance its senior credit facility on or before September 30, 2005, the interest rate will be increased by 5% per annum beginning October 1, 2005 until the date the senior credit facility is refinanced.

The Notes are convertible into shares of the Company's common stock at an initial conversion price of \$4.69 per share. The conversion price is subject to adjustment in certain circumstances.

Under the Securities Purchase Agreement and the Subordination Agreement entered into simultaneously therewith and described below, certain obligations of the Company to the Investors are subordinated to the Company's existing and future "Senior Obligations," which includes all existing and future indebtedness of the Company and its subsidiaries under its credit facility in an amount not to exceed \$90 million. As long as any of the Notes are outstanding, the Company and its subsidiaries will not be permitted to incur any debt, except for, among other things:

- Senior Obligations permitted under the senior credit facility or the refinancing or replacement thereof, including new and replacement letters of credit, all in an amount not to exceed \$90 million;
- capital leases not to exceed \$1 million outstanding at any time;
- operating leases not to exceed \$15 million outstanding at any time;
- purchase money financing in an amount not to exceed \$1 million; and
- debt under the Company's performance and bonding line, not to exceed \$150 million.

At any time following the occurrence of an event of default under the Notes, the Investors have a right to require the Company to purchase all or any part of the outstanding principal amount of the Notes at a purchase price in cash equal to the greater of: (A) 100% of such outstanding principal amount (except that such amount shall equal 110% in the case of any change of control transaction), plus all accrued but unpaid interest thereon and any unpaid liquidated damages and other amounts then owing to the Investors, through the date of purchase, or (B) the "Event Equity Value" (as defined in the Securities Purchase Agreement) of the underlying shares of common stock that would be issuable upon conversion of such principal amount and payment in the Company's common stock of all such accrued but unpaid interest thereon.

Pursuant to the Securities Purchase Agreement, the Company has agreed to indemnify the Investors, their affiliates and agents against certain liabilities. Other than in respect of this transaction, there are no material relationships between the Company, the Investors or their respective affiliates.

A copy of the Securities Purchase Agreement, including the form of Note attached thereto, is attached to this Current Report on Form 8-K as Exhibit 10.1 and is incorporated by reference as though fully set forth herein. The foregoing summary description of the Securities Purchase Agreement and the transactions contemplated therein is not intended to be complete and is qualified in its entirety by the complete text of the Securities Purchase Agreement and related exhibits.

A copy of the press release announcing the completion of the private placement is attached as Exhibit 99.1 to this Current Report on Form 8-K and is incorporated by reference herein.

Subordination Agreement

In connection with the private placement of the Notes, on April 22, 2005, the Company entered into a Subordination Agreement (the "Subordination Agreement") with the Investors as Subordinated Creditors and JPMorgan Chase Bank, N.A., as a Lender, LC Issuer and as Agent (the "Agent") for the Lenders under the senior credit facility, and in favor of the Lenders now or hereafter party to the Company's senior credit facility. The Subordination Agreement sets forth terms of the (i) subordination of payments under the Notes to the prior payment in full of obligations under the Company's senior credit facility and (ii) subordination of any security interests or other liens securing payment of any indebtedness under the Notes to all liens in favor of the Senior Obligations under the senior credit facility.

Other than in respect of this transaction, there are no material relationships between the Company, the Lenders, the Investors or their respective affiliates, except that some of the Lenders and their affiliates have engaged in and may engage in commercial banking transactions and may engage in investment banking transactions with the Company in the ordinary course of business and also have provided or may provide advisory and financial services to the Company.

A copy of the Subordination Agreement is attached to this Current Report on Form 8-K as Exhibit 10.2 and is incorporated by reference herein. The foregoing summary description of the Subordination Agreement and the transactions contemplated therein is not intended to be complete and is qualified in its entirety by the complete text of the Subordination Agreement.

Registration Rights Agreement

In connection with the private placement of the Notes, on April 22, 2005, the Company entered into a Registration Rights Agreement (the "Registration Rights Agreement") with the Investors. The Registration Rights Agreement requires the Company to file a registration statement with respect to the shares of the Company's common stock issuable upon conversion of the Notes within 30 days after the closing date and to cause the registration statement to be declared effective by the Securities and Exchange Commission (the "Commission") no later than the earlier of (i) 120 days after the closing, and (ii) five trading days after the Company is notified by the Commission that the registration statement will not be reviewed or is no longer subject to further review and comments. The Registration Rights Agreement also requires the Company to use its best efforts to keep the registration statement continuously effective until the earlier of (a) the date on which all of the Company's common stock covered by such registration statement have been sold or may be sold without volume restrictions pursuant to Rule 144(k) under the Securities Act of 1933, as amended, or (b) the fifth anniversary of the closing date. If the Company fails to satisfy its obligations under the Registration Rights Agreement, the Company will owe the note holders as partial liquidated damages an amount in cash equal to 1% of the aggregate amount paid for the Notes for each such event, and thereafter on each monthly anniversary of each such event (if the applicable failure shall not have been cured by such date) until the applicable failure is cured, the Company will owe the note holders an amount in cash equal to an additional 1% of the aggregate amount paid for the Notes.

Other than in respect of this transaction, there are no material relationships between the Company, the Investors or their respective affiliates.

A copy of the Registration Rights Agreement is attached to this Current Report on Form 8-K as Exhibit 10.3 and is incorporated by reference herein. The foregoing summary description of the Registration Rights Agreement and the transactions contemplated therein is not intended to be complete and is qualified in its entirety by the complete text of the Registration Rights Agreement.

Credit Agreement Amendment

On April 25, 2005, Matrix Service Company ("the Company") announced that it has entered into Amendment Nine to its credit facility ("Amendment Nine") that increases the revolving commitment from \$32 million to \$35 million and establishes a \$10 million Revolving Loan B commitment. The revolving commitments bear cash pay interest at a rate of prime plus 1% and accrued interest at 1.0% escalating fifty basis points monthly until capped at 5.0% after November 30, 2005. The term loan now bears cash pay and accrued interest at these same rates. The Revolving Loan B commitment expires on October 31, 2005. A copy of the press release and Amendment Nine are attached to this Current Report on Form 8-K as Exhibits 10.5 and 99.2, respectively, and are incorporated by reference herein. The foregoing summary description of Amendment Nine and the transactions contemplated therein is not intended to be complete and is qualified in its entirety by the complete text of Amendment Nine.

Amendment to Rights Agreement

Effective as of November 2, 1999, the Company entered into a Rights Agreement (the "Rights Agreement") with UMB Bank, N.A., the Company's transfer agent (the "Rights Agent"). In connection with the private placement described above, effective as of April 21, 2005, the Company entered into Amendment No. One to the Rights Agreement with the Rights Agent (the

"Rights Amendment"). In general, the Rights Amendment renders the provisions of the Rights Agreement inapplicable to the private placement by exempting the Investors from the definition of "Acquiring Person" as a result of the purchase of the Notes pursuant to the Securities Purchase Agreement.

A copy of the Rights Amendment is attached hereto as Exhibit 10.4 to this Current Report on Form 8-K and is incorporated herein by reference. The foregoing description of the Rights Amendment is qualified in its entirety by reference to the full text of the Rights Amendment.

Item 2.03 Creation of a Direct Financial Obligation under an Off-Balance Sheet Arrangement of a Registrant.

The description of the Securities Purchase Agreement, Notes, Subordination Agreement and Amendment No. 9 set forth under in Item 1.01 of this Current Report on Form 8-K is incorporated by reference herein.

Item 3.02 Unregistered Sales of Equity Securities.

On April 22, 2005, the Company concluded a private placement offering of \$30 million of Senior Unsecured Convertible Notes (the "Notes") at a price equal to \$1.00 for every \$1.00 of Notes purchased. The Notes are convertible into shares of the Company's common stock at an initial conversion price of \$4.69 per share. The conversion price is subject to adjustment in certain circumstances.

A total of \$30 million in principal amount of Notes were sold resulting in gross proceeds of \$30 million. After payment of cash selling commission fees in the amount of \$1.2 million and prepaid interest of \$4.2 million, the Company received net proceeds of \$24.6 million.

The securities were offered pursuant to exemptions from registration under Section 4(2) of the Securities Act of 1933, as amended, and Rule 506 of Regulation D. There were a total of ten purchasers, all of whom were accredited. A legend was placed on each certificate indicating that the Notes have not been registered and are restricted from resale.

Item 9.01 Financial Statements and Exhibits.

Exhibit No.	Description
	
10.1	Securities Purchase Agreement, including form of Note.
10.2	Subordination Agreement.
10.3	Registration Rights Agreement.
10.4	Rights Amendment.
10.5	Senior Credit Facility Amendment 9 dated April 22, 2005.
99.1	Press Release, dated April 25, 2005, issued by the Company.
99.2	Press Release, dated April 25, 2005, issued by the Company.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

By:

Matrix Service Company

Dated: April 25, 2005

/s/ George L. Austin

George L. Austin

Chief Financial Officer and Principal Accounting Officer

EXHIBIT INDEX

Exhibit No.	Description
10.1	Securities Purchase Agreement, including form of Note.
10.2	Subordination Agreement.
10.3	Registration Rights Agreement.
10.4	Rights Amendment.
10.5	Senior Credit Facility Amendment 9 dated April 22, 2005.
99.1	Press Release, dated April 25, 2005, issued by the Company.
99.2	Press Release, dated April 25, 2005, issued by the Company.

SECURITIES PURCHASE AGREEMENT

This Securities Purchase Agreement (this "Agreement") is dated as of April 22, 2005, among Matrix Service Company, a Delaware corporation (the "Company"), and the investors identified on the signature pages hereto (each an "Investor" and, collectively, the "Investors").

WHEREAS, subject to the terms and conditions set forth in this Agreement the Company desires to sell certain securities to each of the Investors and each Investor, severally and not jointly, desires to purchase from the Company certain securities of the Company, as more fully described in this Agreement.

NOW, THEREFORE, IN CONSIDERATION of the mutual covenants contained in this Agreement, and for other good and valuable consideration the receipt and adequacy of which are hereby acknowledged, the Company and the Investors agree as follows:

ARTICLE I. DEFINITIONS

- 1.1 <u>Definitions</u>. In addition to the terms defined elsewhere in this Agreement, for all purposes of this Agreement, the following terms shall have the meanings indicated in this Section 1.1:
 - "Action" means any action, suit, inquiry, notice of violation, proceeding (including any partial proceeding such as a deposition) or investigation pending or threatened in writing against or affecting the Company, any Subsidiary or any of their respective properties before or by any court, arbitrator, governmental or administrative agency, regulatory authority (federal, state, county, local or foreign), stock market, stock exchange or trading facility.
 - "Affiliate" means any Person that, directly or indirectly through one or more intermediaries, controls or is controlled by or is under common control with a Person, as such terms are used in and construed under Rule 144.

"Bankruptcy Event" means any of the following events: (a) the Company or any Subsidiary commences a proceeding under any bankruptcy, reorganization, arrangement, adjustment of debt, relief of debtors, dissolution, insolvency or liquidation or similar law of any jurisdiction relating to the Company or any Subsidiary thereof; (b) there is commenced against the Company or any Subsidiary any such case or proceeding that is not dismissed within 60 days after commencement; (c) the Company or any Subsidiary is adjudicated by a court of competent jurisdiction insolvent or bankrupt or any order of relief or other order approving any such case or proceeding is entered; (d) the Company or any Subsidiary suffers any appointment of any custodian or the like for it or any substantial part of its property that is not discharged or stayed within 60 days; (e) under applicable law the Company or any Subsidiary makes a general

assignment for the benefit of creditors; (f) the Company or any Subsidiary fails to pay, or states that it is unable to pay or is unable to pay, its debts generally as they become due; (g) the Company or any Subsidiary calls a meeting of its creditors with a view to arranging a composition, adjustment or restructuring of its debts; or (h) the Company or any Subsidiary, by any act or failure to act, expressly indicates its consent to, approval of or acquiescence in any of the foregoing or takes any corporate or other action for the purpose of effecting any of the foregoing.

"Benefit Arrangement" means at any time an employee benefit plan within the meaning of Section 3(3) of ERISA which is not a Plan or Multiemployer Plan and which is maintained or otherwise contributed by the Company.

"Benefit Plan" has the meaning set forth in Section 3.1(aa)(ii).

"Business Day" means any day except Saturday, Sunday and any day that is a federal legal holiday or a day on which banking institutions in the State of Delaware, Oklahoma or Texas are authorized or required by law or other governmental action to close.

"Cash Interest Coverage Ratio" means, as of any fiscal quarter end of the Company, the ratio of (i) EBITDA for the period of the four consecutive fiscal quarters ending thereon to (ii) Cash Interest Expense for such period.

"Cash Interest Expense" means, for any period, interest expense of the Company and the Subsidiaries for such period (excluding interest expense that has been capitalized and not paid in cash), determined on a consolidated basis in accordance with GAAP.

"Closing" means the closing of the purchase and sale of the Securities pursuant to Article II.

"Closing Date" means the Business Day immediately following the date on which all of the conditions set forth in Sections 5.1 and 5.2 hereof are satisfied, or such other date as the parties may agree.

"Code" means the Internal Revenue Code of 1986, as amended.

"Commission" means the Securities and Exchange Commission.

"Common Stock" means the common stock of the Company, par value \$.01 per share, and any securities into which such common stock may hereafter be reclassified.

"Common Stock Equivalents" means any securities of the Company or any Subsidiary which entitle the holder thereof to acquire Common Stock at any time, including without limitation, any debt, preferred stock, rights, options, warrants or other instrument that is at any time convertible into or exchangeable for, or otherwise entitles the holder thereof to

receive, Common Stock or other securities that entitle the holder to receive, directly or indirectly, Common Stock.

- "Company Counsel" means Conner & Winters, LLP.
- "Company Deliverables" has the meaning set forth in Section 2.2(a).
- "Contingent Liability" means, as to any Person, any obligation, contingent or otherwise, of such Person guaranteeing or having the economic effect of guaranteeing any Debt of any other Person in any manner, whether directly or indirectly, including without limitation any obligation of such Person, direct or indirect, (a) to purchase or pay (or advance or supply funds for the purchase or payment of) such Debt or to purchase (or to advance or supply funds for the purchase of) any security for the payment of such Debt, (b) to purchase property or services for the purpose of assuring the owner of such Debt of its payment, or (c) to maintain the solvency, working capital, equity, cash flow, fixed charge or other coverage ratio, or any other financial condition of the primary obligor so as to enable the primary obligor to pay any Debt or to comply with any agreement relating to any Debt or obligation.

"Debt" of any Person shall mean, without duplication, (a) all obligations of such Person for borrowed money or with respect to deposits or advances of any kind, (b) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (c) all obligations of such Person upon which interest charges are customarily paid, (d) all obligations of such Person under conditional sale or other title retention agreements relating to property or assets purchased by such Person, (e) all obligations of such Person issued or assumed as the deferred purchase price of property or services (other than unsecured accounts payable incurred in the ordinary course of business), (f) all Debt of others secured by (or for which the holder of such Debt has an existing right, contingent or otherwise, to be secured by) any Lien on property owned or acquired by such Person, whether or not the obligations secured thereby have been assumed, (g) all obligations of such Person in respect of interest rate protection agreements, foreign currency exchange agreements or other interest or exchange rate hedging arrangements that exceed amounts necessary to hedge the Company's cross-currency exposure, (h) all obligations of such Person as an account party in respect of letters of credit and bankers' acceptances. The Debt of any Person shall include the Debt of any partnership in which such Person is a general partner, (i) all obligations of such Person as lessee which (y) are capitalized in accordance with GAAP or (z) arise pursuant to sale-leaseback transactions and (j) all Debt of others guaranteed by such Person.

- "Delaware Courts" means the state and federal courts sitting in the State of Delaware.
- "Disclosure Materials" has the meaning set forth in Section 3.1(h).
- **"EBITDA"** means consolidated net income determined in accordance with GAAP plus, to the extent deducted in determining consolidated net income, (i) consolidated

interest expense, (ii) expense for taxes paid or accrued, (iii) depreciation, amortization and other non-cash charges, including non-cash charges related to the implementation of the Company's restructuring plan, and cash charges for professional fees associated therewith, (iv) losses on sale of fixed assets, and (v) extraordinary losses incurred other than in the ordinary course of business, minus, to the extent included in consolidated net income, (i) gains on sales of fixed assets, and (ii) extraordinary gains realized other than in the ordinary course of business, all calculated for the Company and its Subsidiaries on a consolidated basis for the then most recently ended four fiscal quarters.

"Effective Date" means the date that the Registration Statement required by Section 2(a) of the Registration Rights Agreement is first declared effective by the Commission.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, or any successor statute.

"ERISA Group" means the Company and each Subsidiary and all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control which, together with the Company or any Subsidiary, are treated as a single employer under the Code.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"First Notice" has the meaning set forth in Section 4.11.

"GAAP" means U.S. generally accepted accounting principles.

"Intellectual Property Rights" has the meaning set forth in Section 3.1(p).

"Interest Coverage Ratio" means, as of any fiscal quarter end of the Company, the ratio of (i) EBITDA for the period of the four consecutive fiscal quarters ending thereon to (ii) Interest Expense for such period.

"Interest Expense" means, for any period, interest expense of the Company and the Subsidiaries for such period, determined on a consolidated basis in accordance with GAAP.

"Investment Amount" means, with respect to each Investor, the investment amount indicated below such Investor's signature page to this Agreement.

"Investor Deliverables" has the meaning set forth in Section 2.2(b).

"Investor Party" has the meaning set forth in Section 4.10.

"Leverage Ratio" means, as of any date, the ratio of (i) Total Debt as of such date to (ii) EBITDA for the period of four consecutive fiscal quarters ending on, or most recently before, such date.

"Lien" means any lien, charge, encumbrance, security interest, right of first refusal or other restrictions of any kind.

"Losses" has the meaning set forth in Section 4.10.

"Material Adverse Effect" means any of (i) a material and adverse effect on the legality, validity or enforceability of any Transaction Document, (ii) a material and adverse effect on the results of operations, assets, prospects, business or condition (financial or otherwise) of the Company and the Subsidiaries, taken as a whole, or (iii) an adverse impairment to the Company's ability to perform on a timely basis its obligations under any Transaction Document

"New Issue Securities" has the meaning set forth in Section 4.11.

"Notes" means the 7% senior unsecured convertible promissory notes issuable by the Company to the Investors pursuant to terms hereof, in the Form of Exhibit A, due five years from the Original Issue Date (as defined therein) which, among other things, give the Holders thereof the right to acquire shares of Common Stock on the terms thereof.

"Outside Date" means May 6, 2005.

"Permitted Indebtedness" has the meaning set forth in Section 6.3.

"Permitted Liens" means: (a) liens for taxes, assessments or governmental charges not delinquent or being contested in good faith and by appropriate proceedings and for which adequate reserves in accordance with GAAP are maintained on the books of the Company or the applicable Subsidiary; (b) liens arising out of deposits in connection with workers' compensation, unemployment insurance, old age pensions or other social security or retirement benefits legislation; (c) deposits or pledges to secure bids, tenders, contracts (other than contracts for the payment of money), leases, statutory obligations, surety and appeal bonds, and other obligations of like nature arising in the ordinary course of business of the Company or a Subsidiary; (d) liens imposed by law, such as mechanics', workers', materialmens', carriers' or other like liens arising in the ordinary course of business of the Company or a Subsidiary which secure the payment of obligations which are not past due or which are being diligently contested in good faith by appropriate proceedings and for which adequate reserves in accordance with GAAP are maintained on the books of the Company or the applicable Subsidiary; (e) liens existing on the Closing Date, and described on Schedule 3.1(o); (f) purchase money security interests or liens for the purchase of fixed assets to be used in the business of the Company or a Subsidiary, securing solely the fixed assets so purchased and the proceeds thereof; (g) capitalized leases which do not violate any provision of this Agreement; (h) liens of commercial depository institutions, arising in the ordinary course of business, constituting a statutory or common law right of setoff against amounts on deposit with such institution; and (i) rights of way, zoning restrictions, easements and similar encumbrances affecting the Company's real property which do not materially interfere with the use of such property.

"Person" means an individual or corporation, partnership, trust, incorporated or unincorporated association, joint venture, limited liability company, joint stock company, government (or an agency or subdivision thereof) or other entity of any kind.

"PBGC" means the Pension Benefit Guarantee Corporation or any entity succeeding to any or all of its functions under ERISA.

"Plan" means at any time an employee pension plan benefit plan which is covered by Title IV of ERISA or subject to the minimum funding standards under the Code and either (i) is maintained, or contributed to, by any member of the ERISA group for employees of any member of the ERISA group or (ii) has at any time within the preceding five years been maintained, or contributed to, by any Person which was at such time a member of the ERISA Group for employees of any Person which was at such time a member of the ERISA group.

"Prepaid Interest" means an amount in United States Dollars equal to the full amount of interest that would accrue on the original principal amount of the Notes (assuming no payment, prepayment, redemption, or repurchase) pursuant to Section 2 of the Notes for the two year period commencing on the Original Issue Date of such Notes through the 2nd anniversary of such Original Issue Date.

"Proceeding" means an action, claim, suit, investigation or proceeding (including, without limitation, an investigation or partial proceeding, such as a deposition), whether commenced or threatened.

"Purchase Money Financing" has the meaning set forth in Section 6.3.

"Registration Statement" means a registration statement meeting the requirements set forth in the Registration Rights Agreement and covering the resale by the Investors of the Underlying Shares.

"Registration Rights Agreement" means the Registration Rights Agreement, dated as of the date of this Agreement, among the Company and the Investors, in the form of Exhibit B hereto.

"Required Investors" means one or more Investors representing greater than 50% of the aggregate principal amount of all Notes then outstanding.

"Required Minimum" means, as of any date, the maximum aggregate number of shares of Common Stock then issued or potentially issuable in the future pursuant to the Transaction Documents that the Company is obligated to issue, whether contingently or otherwise, including, without limitation, any Underlying Shares issuable upon conversion in full of all Notes.

"Restricted Payment" means, with respect to any Person, (a) any direct or indirect distribution, dividend or other payment on account of any equity interest in, or shares of

capital stock or other securities of, such Person and (b) any management, consulting or other similar fees, or any interest thereon, payable by such Person to any affiliate of such Person (other than the Company), or to any other Person other than an unrelated third party; <u>provided</u>, however, that Restricted Payments shall not include any arms length consulting agreements with consultants of the Company which are approved by the Board of Directors of the Company.

"Rule 144" means Rule 144 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"SEC Reports" has the meaning set forth in Section 3.1(h).

"Securities" means the Notes and the Underlying Shares issuable thereunder.

"Securities Act" means the Securities Act of 1933, as amended.

"Senior Credit Agreement" means that certain Credit Agreement dated as of March 7, 2003, among the Company, the Agent designated therein and the various Lenders party thereto, as amended by that certain Amendment One to Credit Agreement dated as of May 22, 2003, that certain Amendment Two to Credit Agreement dated as of August 27, 2003, that certain Amendment Three to Credit Agreement dated as of December 19, 2003, that certain Amendment Four to Credit Agreement dated as of March 11, 2004, that certain Amendment Five to Credit Agreement dated as of May 6, 2004, that certain Amendment Six to Credit Agreement dated as of August 5, 2004, that certain Amendment Seven to Credit Agreement dated as of October 6, 2004, that certain Amendment Eight to Credit Agreement dated as of November 30, 2004, that certain letter agreement dated March 23, 2005, and that certain letter agreement dated April 8, 2005, and that certain Amendment Nine to Credit Agreement dated as of April 22, 2005 and as may hereafter be amended from time to time.

"Senior Credit Facility" means the "Loans" described in the Senior Credit Agreement.

"Senior Documents" means the Senior Credit Agreement and the other Loan Documents, as defined in the Senior Credit Agreement, each as may be amended, restated, supplemented, modified, extended, renewed, refinanced or replaced from time to time, and any replacement thereof in connection with any refinancing of the Senior Obligations.

"Senior Obligations" means all indebtedness, obligations and other liabilities of the Company and its Subsidiaries, or any of them, now or hereafter existing in favor of the Lenders and the Agent (as defined under the Senior Documents), whether for principal (including, without limitation, protective advances), reimbursement obligations, interest (including interest accruing subsequent to the filing of any petition initiating any Reorganization (as defined under the Senior Documents), whether or not a claim for such interest is allowed in any such proceeding), guaranteed obligations, fees, premiums, indemnities, costs, expenses (including, without limitation, auditor, legal and other professional fees, costs and expenses), or

otherwise, provided that the principal amount of extensions of credit under the Senior Documents that shall constitute Senior Obligations for purposes of this Agreement shall under no circumstances exceed \$90 million.

"Short Sales" include, without limitation, all "short sales" as defined in Rule 200 promulgated under Regulation SHO under the Exchange Act and all types of direct and indirect stock pledges, forward sale contracts, options, puts, calls, short sales, swaps and similar arrangements (including on a total return basis), and sales and other transactions through non-US broker dealers or foreign regulated brokers.

"Steelhead" means Steelhead Investments Ltd., a Cayman Islands corporation.

"Subordination Agreement" means that certain Subordination Agreement dated April 22, 2005, among the Investors, the Company, the various obligors listed on the signature pages thereto and JPMorgan Chase Bank, N.A., as a lender, letter of credit issuer and as agent for the lenders under the Credit Agreement, in the form of Exhibit C hereto.

"Subsidiary" means any subsidiary of the Company included in the SEC Reports.

"Total Debt" means, as of any date, the aggregate principal amount of all Debt of the Company and the Subsidiaries as of such date determined on a consolidated basis in accordance with GAAP.

"Trading Day" means (i) a day on which the Common Stock is traded on a Trading Market, or (ii) if the Common Stock is not quoted on the OTC Bulletin Board, a day on which the Common Stock is quoted in the over-the-counter market as reported by the National Quotation Bureau Incorporated (or any similar organization or agency succeeding to its functions of reporting prices); provided, that in the event that the Common Stock is not listed or quoted as set forth in (i) and (ii) hereof, then Trading Day shall mean a Business Day.

"Trading Market" means whichever of the New York Stock Exchange, the American Stock Exchange, the NASDAQ National Market, the NASDAQ SmallCap Market or OTC Bulletin Board on which the Common Stock is listed or quoted for trading on the date in question.

"Transaction Documents" means this Agreement, the Notes, the Registration Rights Agreement and any other documents or agreements executed in connection with the transactions contemplated hereunder, but excluding the Subordination Agreement.

"Underlying Shares" means the shares of Common Stock issuable upon conversion of the Notes.

ARTICLE II. PURCHASE AND SALE

- 2.1 <u>Closing</u>. Subject to the terms and conditions set forth in this Agreement, at the Closing the Company shall issue and sell to each Investor, and each Investor shall, severally and not jointly, purchase from the Company, the Notes representing such Investor's Investment Amount. The Closing shall take place at the offices of Bryan Cave LLP, 1290 Avenue of the Americas, New York, NY 10104 on the Closing Date or at such other location or time as the parties may agree.
- 2.2 <u>Closing Deliveries</u>. (a) At the Closing, the Company shall deliver or cause to be delivered to each Investor the following (the "Company Deliverables"):
 - (i) Notes in the aggregate principal amount of the Investment Amount indicated below such Investor's name on its signature page of this Agreement, registered in the name of such Investor;
 - (ii) the legal opinion of Company Counsel, in agreed form, addressed to the Investors;
 - (iii) the Registration Rights Agreement, duly executed by the Company;
 - (iv) the Subordination Agreement, duly executed by the Company; and
 - (v) any other documents reasonably requested by such Investor.
 - (b) At the Closing, each Investor shall deliver or cause to be delivered to the Company the following (the "Investor Deliverables"):
 - (i) its Investment Amount indicated below such Investor's name on the signature page of this Agreement, in United States dollars and in immediately available funds, by wire transfer to an account designated in writing by the Company for such purpose less an amount equal to the Prepaid Interest on the principal amount of the Notes issuable to such Investor and, in the case of Steelhead, less the amount contemplated in Section 7.1 (adjusted for payments previously made);
 - (ii) the Registration Rights Agreement, duly executed by such Investor; and
 - (iii) the Subordination Agreement, duly executed by the Investor.

ARTICLE III. REPRESENTATIONS AND WARRANTIES

- 3.1 Representations and Warranties of the Company. The Company hereby makes the following representations and warranties to each Investor:
- (a) <u>Subsidiaries</u>. The Company has no direct or indirect Subsidiaries other than as specified in <u>Schedule 3.1(a)</u>. Except as disclosed in <u>Schedule 3.1(a)</u>, the Company owns, directly or indirectly, all of the capital stock of each Subsidiary free and clear of any and all Liens (other than Permitted Liens), and all the issued and outstanding shares of capital stock of each Subsidiary are validly issued and are fully paid, non-assessable and free of preemptive and similar rights.
- (b) <u>Organization and Qualification</u>. The Company and each Subsidiary are duly incorporated or otherwise organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation or organization (as applicable), with the requisite power and authority to own and use its properties and assets and to carry on its business as currently conducted. Neither the Company nor any Subsidiary is in violation of any of the provisions of its respective certificate or articles of incorporation, bylaws or other organizational or charter documents. The Company and each Subsidiary are duly qualified to conduct its respective businesses and are in good standing as a foreign corporation or other entity in each jurisdiction in which the nature of the business conducted or property owned by it makes such qualification necessary, except where the failure to be so qualified or in good standing, as the case may be, could not, individually or in the aggregate, have or reasonably be expected to result in a Material Adverse Effect.
- (c) <u>Authorization; Enforcement</u>. The Company has the requisite corporate power and authority to enter into and to consummate the transactions contemplated by each of the Transaction Documents and otherwise to carry out its obligations thereunder. The execution and delivery of each of the Notes and the other Transaction Documents by the Company and the consummation by it of the transactions contemplated thereby have been duly authorized by all necessary action on the part of the Company and no further action is required by the Company in connection therewith. Each Transaction Document has been (or upon delivery will have been) duly executed by the Company and, when delivered in accordance with the terms hereof, will constitute the valid and binding obligation of the Company enforceable against the Company in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, liquidation or similar laws relating to, or affecting generally the enforcement of, creditors' rights and remedies or by other equitable principles of general application.
- (d) No Conflicts. The execution, delivery and performance of the Transaction Documents by the Company and the consummation by the Company of the transactions contemplated thereby (i) do not and will not conflict with or violate any provision of the Company's or any Subsidiary's certificate or articles of incorporation, bylaws or other

organizational or charter documents, or (ii) upon execution and delivery of Amendment Nine to the Senior Credit Agreement, will not conflict with, or constitute a default (or an event that with notice or lapse of time or both would become a default) under, or give to others any rights of termination, amendment, acceleration or cancellation (with or without notice, lapse of time or both) of, any agreement, credit facility, debt or other instrument (evidencing a Company or Subsidiary debt or otherwise) or other understanding to which the Company or any Subsidiary is a party or by which any property or asset of the Company or any Subsidiary is bound or affected, or (iii) result in a violation of any law, rule, regulation, order, judgment, injunction, decree or other restriction of any court or governmental authority to which the Company or a Subsidiary is subject (including federal and state securities laws and regulations), or by which any property or asset of the Company or a Subsidiary is bound or affected; except in the case of each of clauses (ii) and (iii), such as could not, individually or in the aggregate, have or reasonably be expected to result in a Material Adverse Effect. Except to the extent prohibited by the Subordination Agreement, payments of cash on account of principal of or interest under the Notes, upon any Event of Default under the Notes, as a result of liquidated damages under any Transaction Document or upon a Buy In under and as such term is defined in the Notes will not require the consent of, any payment to, or the springing of any Lien in favor of any lender to or creditor of the Company or any Subsidiary (under a credit facility, loan agreement or otherwise) and will not result in a default under any such credit facilities, loans or other agreements.

- (e) <u>Filings, Consents and Approvals</u>. The Company is not required to obtain any consent, waiver, authorization or order of, give any notice to, or make any filing or registration with, any court or other federal, state, local or other governmental authority or other Person in connection with the execution, delivery and performance by the Company of the Transaction Documents, other than (i) the filing with the Commission of one or more Registration Statements in accordance with the requirements Registration Rights Agreement, (ii) filings required by state securities laws, (iii) the filing of a Notice of Sale of Securities on Form D with the Commission under Regulation D of the Securities Act (iv) the filings required in accordance with Section 4.8, (v) the filing of notification forms regarding listing of additional shares and change in the number of shares outstanding with the Nasdaq National Market and (vi) those that have been made or obtained prior to the date of this Agreement.
- (f) <u>Issuance of the Securities</u>. The Underlying Shares have been duly authorized and, when issued and paid for in accordance with the Notes, will be duly and validly issued, fully paid and nonassessable, free and clear of all Liens. The Company has reserved from its duly authorized capital stock a number of shares of Common Stock issuable upon conversion of the Notes, which number of reserved shares is not less than the Required Minimum calculated as of the date hereof.
- (g) <u>Capitalization</u>. The number of shares and type of all authorized, issued and outstanding capital stock of the Company, and all shares of Common Stock reserved for issuance under the Company's various option and incentive plans, is specified in the SEC Reports. Except as specified in the SEC Reports, no securities of the Company are entitled to preemptive or similar rights, and no Person has any right of first refusal, preemptive right, right

of participation, or any similar right to participate in the transactions contemplated by the Transaction Documents. Except as specified in the SEC Reports, there are no outstanding options, warrants, scrip rights to subscribe to, calls or commitments of any character whatsoever relating to, or securities, rights or obligations convertible into or exchangeable for, or giving any Person any right to subscribe for or acquire, any shares of Common Stock, or contracts, commitments, understandings or arrangements by which the Company or any Subsidiary is or may become bound to issue additional shares of Common Stock, or securities or rights convertible or exchangeable into shares of Common Stock. The issue and sale of the Securities will not, immediately or with the passage of time, obligate the Company to issue shares of Common Stock or other securities to any Person (other than the Investors or any subsequent holders of the Securities) and will not result in a right of any holder of Company securities to adjust the exercise, conversion, exchange or reset price under such securities.

- (h) SEC Reports; Financial Statements. The Company has filed all reports required to be filed by it under the Securities Act and the Exchange Act, including pursuant to Section 13(a) or 15(d) thereof, for the twelve months preceding the date hereof (or such shorter period as the Company was required by law to file such reports) (the foregoing materials being collectively referred to herein as the "SEC Reports" and, together with the Schedules to this Agreement (if any), the "Disclosure Materials") on a timely basis or has timely filed a valid extension of such time of filing and has filed any such SEC Reports prior to the expiration of any such extension. As of their respective dates, the SEC Reports complied in all material respects with the requirements of the Securities Act and the Exchange Act and the rules and regulations of the Commission promulgated thereunder, and none of the SEC Reports, when filed, contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The financial statements of the Company included in the SEC Reports comply in all material respects with applicable accounting requirements and the rules and regulations of the Commission with respect thereto as in effect at the time of filing. Such financial statements have been prepared in accordance with GAAP applied on a consistent basis during the periods involved, except as may be otherwise specified in such financial statements or the notes thereto, and fairly present in all material respects the financial position of the Company and its consolidated Subsidiaries as of and for the dates thereof and the results of operations and cash flows for the periods then ended, subject, in the case of unaudited statements, to normal, immaterial, year-end audit adjustments.
- (i) <u>Press Releases</u>. The press releases disseminated by the Company during the twelve months preceding the date of this Agreement taken as a whole do not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made and when made, not misleading.
- (j) <u>Material Changes</u>. Since the date of the latest audited financial statements included within the SEC Reports, except as specifically disclosed in the SEC Reports, (i) there has been no event, occurrence or development that has had or that could reasonably be expected

to result in a Material Adverse Effect, (ii) the Company has not incurred any liabilities (contingent or otherwise) other than (A) trade payables, accrued expenses and other liabilities incurred in the ordinary course of business consistent with past practice and (B) liabilities not required to be reflected in the Company's financial statements pursuant to GAAP or required to be disclosed in filings made with the Commission, (iii) the Company has not altered its method of accounting or the identity of its auditors, (iv) the Company has not declared or made any dividend or distribution of cash or other property to its stockholders or purchased, redeemed or made any agreements to purchase or redeem any shares of its capital stock other than the Company's share buyback plan, and (v) the Company has not issued any equity securities to any officer, director or Affiliate, except pursuant to existing Company stock option plans.

- (k) <u>Litigation</u>. There is no Action which (i) adversely affects or challenges the legality, validity or enforceability of any of the Transaction Documents or the Securities or (ii) except as specifically disclosed in the SEC Reports, would, if there were an unfavorable decision, individually or in the aggregate, have or reasonably be expected to result in a Material Adverse Effect. Neither the Company nor any Subsidiary, nor any director or officer thereof (in his or her capacity as such), is or has been the subject of any Action involving a claim of violation of or liability under federal or state securities laws or a claim of breach of fiduciary duty, except as specifically disclosed in the SEC Reports. There has not been, and to the knowledge of the Company, there is not pending any investigation by the Commission involving the Company or any current or former director or officer of the Company (in his or her capacity as such). The Commission has not issued any stop order or other order suspending the effectiveness of any registration statement filed by the Company or any Subsidiary under the Exchange Act or the Securities Act.
- (l) <u>Labor Relations</u>. Except as disclosed in <u>Schedule 3.1(l)</u>, no material labor dispute exists or, to the knowledge of the Company, is imminent with respect to any of the employees of the Company.
- (m) <u>Compliance</u>. Except as disclosed in <u>Schedule 3.1(m)</u>, neither the Company nor any Subsidiary (i) is in default under or in violation of (and no event has occurred that has not been waived that, with notice or lapse of time or both, would result in a default by the Company or any Subsidiary under), nor has the Company or any Subsidiary received notice of a claim that it is in default under or that it is in violation of, any indenture, loan or credit agreement or any other agreement or instrument to which it is a party or by which it or any of its properties is bound (whether or not such default or violation has been waived), (ii) is in violation of any order of any court, arbitrator or governmental body, or (iii) is or has been in violation of any statute, rule or regulation of any governmental authority, including without limitation all foreign, federal, state and local laws relating to taxes, environmental protection, occupational health and safety, product quality and safety and employment and labor matters, except in each case as could not, individually or in the aggregate, have or reasonably be expected to result in a Material Adverse Effect. The Company has substantially complied with all effective requirements of the Sarbanes-Oxley Act of 2002, as amended, and the rules and regulations

thereunder, that are applicable to it, except where such noncompliance could not have or reasonably be expected to result in a Material Adverse Effect.

- (n) <u>Regulatory Permits</u>. The Company and the Subsidiaries possess all certificates, authorizations and permits issued by the appropriate federal, state, local or foreign regulatory authorities necessary to conduct their respective businesses as described in the SEC Reports, except where the failure to possess such permits could not, individually or in the aggregate, have or reasonably be expected to result in a Material Adverse Effect, and neither the Company nor any Subsidiary has received any notice of proceedings relating to the revocation or modification of any such permits.
- (o) <u>Title to Assets</u>. The Company and the Subsidiaries have good and marketable title in fee simple to all real property owned by them that is material to their respective businesses and good and valid title in all personal property owned by them that is material to their respective businesses, in each case free and clear of all Liens, except for Permitted Liens and Liens that do not materially affect the value of such property and do not materially interfere with the use made and proposed to be made of such property by the Company and the Subsidiaries. Any real property and facilities held under lease by the Company and the Subsidiaries are held by them under valid, subsisting and enforceable leases of which the Company and the Subsidiaries are in compliance, except as could not, individually or in the aggregate, have or reasonably be expected to result in a Material Adverse Effect.
- (p) <u>Patents and Trademarks</u>. The Company and the Subsidiaries have, or have rights to use, all patents, patent applications, trademarks, trademarks applications, service marks, trade names, copyrights, licenses and other similar rights that are necessary or material for use in connection with their respective businesses as described in the SEC Reports and which the failure to so have could, individually or in the aggregate, have or reasonably be expected to result in a Material Adverse Effect (collectively, the "Intellectual Property Rights"). Neither the Company nor any Subsidiary has received a written notice that the Intellectual Property Rights used by the Company or any Subsidiary violates or infringes upon the rights of any Person. Except as set forth in the SEC Reports, to the knowledge of the Company, all such Intellectual Property Rights are enforceable and there is no existing infringement by another Person of any of the Intellectual Property Rights.
- (q) <u>Insurance</u>. The Company and the Subsidiaries are insured by insurers of recognized financial responsibility against such losses and risks and in such amounts as are prudent and customary in the businesses in which the Company and the Subsidiaries are engaged. The Company has no reason to believe that it will not be able to renew its and the Subsidiaries' existing insurance coverage as and when such coverage expires or to obtain similar coverage from similar insurers as may be necessary to continue its business on terms consistent with market for the Company's and such Subsidiaries' respective lines of business.
- (r) <u>Transactions With Affiliates and Employees</u>. Except as set forth in the SEC Reports, none of the officers or directors of the Company and, to the knowledge of the

Company, none of the employees of the Company is presently a party to any transaction with the Company or any Subsidiary (other than for services as employees, officers and directors), including any contract, agreement or other arrangement providing for the furnishing of services to or by, providing for rental of real or personal property to or from, or otherwise requiring payments to or from any officer, director or such employee or, to the knowledge of the Company, any entity in which any officer, director, or any such employee has a substantial interest or is an officer, director, trustee or partner.

- (s) Internal Accounting Controls. The Company and the Subsidiaries maintain a system of internal accounting controls sufficient to provide reasonable assurance that (i) transactions are executed in accordance with management's general or specific authorizations, (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with generally accepted accounting principles and to maintain asset accountability, (iii) access to assets is permitted only in accordance with management's general or specific authorization, and (iv) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences. The Company has established disclosure controls and procedures (as defined in Exchange Act rules 13a-14 and 15d-14) for the Company and designed such disclosure controls and procedures to ensure that material information relating to the Company, including its Subsidiaries, is made known to the certifying officers by others within those entities, particularly during the period in which the Company's Form 10-K or 10-Q, as the case may be, is being prepared. The Company's certifying officers have evaluated the effectiveness of the Company's controls and procedures in accordance with Item 307 of Regulation S-K under the Exchange Act for the Company's most recently ended fiscal quarter or fiscal year-end (such date, the "Evaluation Date"). The Company presented in its most recently filed Form 10-K or Form 10-Q the conclusions of the certifying officers about the effectiveness of the disclosure controls and procedures based on their evaluations as of the Evaluation Date. Since the Evaluation Date, there have been no changes in the Company's internal controls that would be required to be disclosed pursuant to Item 308(c) of Regulation S-K under the Exchange Act or, to the Company's knowledge, in other factors that could reasonably be expected to have a Material Adverse Effect on the Company's intern
- (t) <u>Solvency</u>. Based on the financial condition of the Company as of the Closing Date (and assuming that the Closing shall have occurred and the credit commitment under the Company's Senior Credit Agreement shall have been increased as described in Section 5.1(i) hereof), (i) the Company's fair saleable value of its assets exceeds the amount that will be required to be paid on or in respect of the Company's existing debts and other liabilities (including known contingent liabilities) as they mature; (ii) the Company's assets do not constitute unreasonably small capital to carry on its business for the current fiscal year as now conducted and as proposed to be conducted including its capital needs taking into account the particular capital requirements of the business conducted by the Company, and projected capital requirements and capital availability thereof; and (iii) the current cash flow of the Company, together with the fair saleable value of its assets, after taking into account all anticipated uses of the cash, would be sufficient to pay all amounts on or in respect of its debt when such amounts

are required to be paid. The Company has no current intention to incur debts beyond its ability to pay such debts as they mature (taking into account the timing and amounts of cash to be payable on or in respect of its debt).

- (u) <u>Certain Fees</u>. Except for the fee payable to Banc of America Securities LLC for its services as exclusive placement agent with respect to the Securities, no brokerage or finder's fees or commissions are or will be payable by the Company to any broker, financial advisor or consultant, finder, placement agent, investment banker, bank or other Person with respect to the transactions contemplated by this Agreement. The Investors shall have no obligation with respect to any fees or with respect to any claims (other than such fees or commissions owed by an Investor pursuant to written agreements executed by such Investor which fees or commissions shall be the sole responsibility of such Investor) made by or on behalf of other Persons for fees of a type contemplated in this Section that may be due in connection with the transactions contemplated by this Agreement.
- (v) <u>Certain Registration Matters</u>. Assuming the accuracy of the Investors' representations and warranties set forth in Section 3.2(b)-(e), no registration under the Securities Act is required for the offer and sale of the Notes by the Company to the Investors under the Transaction Documents. The Company is eligible to register its Common Stock for resale by the Investors under Form S-3 promulgated under the Securities Act. Except as specified in <u>Schedule 3.1(v)</u>, the Company has not granted or agreed to grant to any Person any rights (including "piggy-back" registration rights) to have any securities of the Company registered with the Commission or any other governmental authority that have not been satisfied.
- (w) <u>Listing and Maintenance Requirements</u>. Except as specified in the SEC Reports, the Company has not, in the two years preceding the date hereof, received notice from any Trading Market to the effect that the Company is not in compliance with the listing or maintenance requirements thereof. The Company is, and has no reason to believe that it will not in the foreseeable future continue to be, in compliance with the listing and maintenance requirements for continued listing of the Common Stock on the Trading Market on which the Common Stock is currently listed or quoted. The issuance and sale of the Securities under the Transaction Documents does not contravene the rules and regulations of the Trading Market on which the Common Stock is currently listed or quoted, and no approval of the shareholders of the Company thereunder is required for the Company to issue and deliver to the Investors the Securities contemplated by Transaction Documents.
- (x) <u>Investment Company</u>. The Company is not, and is not an Affiliate of, and immediately following the Closing will not have become, an "investment company" within the meaning of the Investment Company Act of 1940, as amended.
- (y) <u>Application of Takeover Protections</u>. The Company has taken all necessary action, if any, in order to render inapplicable any control share acquisition, business combination, poison pill (including any distribution under a rights agreement) or other similar anti-takeover provision under the Company's Certificate of Incorporation (or similar charter

documents) or the laws of its state of incorporation that is or could become applicable to the Investors as a result of the Investors and the Company fulfilling their obligations or exercising their rights under the Transaction Documents, including without limitation the Company's issuance of the Securities and the Investors' ownership of the Securities.

- (z) No Additional Agreements. The Company does not have any agreement or understanding with any Investor with respect to the transactions contemplated by the Transaction Documents other than as specified in the Transaction Documents and the Subordination Agreement.
- (aa) <u>Compliance with ERISA</u>. (i) Each member of the ERISA Group has fulfilled its obligations under the minimum funding standards of ERISA and the Code with respect to each Plan and is in compliance in all material respects with the presently applicable provisions of ERISA and the Code with respect to each Plan. No member of the ERISA Group has (i) sought a waiver of the minimum funding standard under Section 412 of the Code in respect of any Plan, (ii) failed to make any required contribution or payment to any Plan or Multiemployer Plan or in respect of any Benefit Arrangement, or made any amendment to any Plan or Benefit Arrangement, which has resulted or could result in the imposition of a Lien or the posting of a bond or other security under ERISA or the Code or (iii) incurred any liability under Title IV of ERISA other than a liability to the PBGC for premiums under Section 4007 of ERISA.
 - (ii) The benefit plans not covered under clause (i) above (including profit sharing, deferred compensation, stock option, employee stock purchase, bonus, retirement, health or insurance plans, collectively the "Benefit Plans") relating to the employees of the Company are duly registered where required by, and are in good standing in all material respects under, all applicable laws. All required employer and employee contributions and premiums under the Benefit Plans to the date hereof have been made, the respective fund or funds established under the Benefit Plans are funded in accordance with applicable laws, and no past service funding liabilities exist thereunder.
 - (iii) No Benefit Plans have any unfunded liabilities, either on a "going concern" or "winding up" basis and determined in accordance with all applicable laws and actuarial practices and using actuarial assumptions and methods that are reasonable in the circumstances. No event has occurred and no condition exists with respect to any Benefit Plans that has resulted or could reasonably be expected to result in any pension plan having its registration revoked or wound up (in whole or in part) or refused for the purposes of any applicable laws or being placed under the administration of any relevant pension benefits regulatory authority or being required to pay any taxes or penalties (in any material amounts) under any applicable laws.
- (bb) <u>Taxes</u>. All United States federal, state, county, municipality local or foreign income tax returns and all other material tax returns (including foreign tax returns) which are required to be filed by or on behalf of the Company and each Subsidiary have been filed and

all material taxes due pursuant to such returns or pursuant to any assessment received by the Company and each Subsidiary have been paid except those being disputed in good faith and for which adequate reserves have been established. The charges, accruals and reserves on the books of the Company and each Subsidiary in respect of taxes or other governmental charges have been established in accordance with GAAP.

- (cc) <u>Absence of Any Undisclosed Liabilities or Capital Calls</u>. Except for litigation described in the SEC Reports, there are no liabilities of the Company or any Subsidiary of any kind whatsoever, whether accrued, contingent, absolute, determined, determinable or otherwise, and there is no existing condition, situation or set of circumstances which could reasonably be expected to result in such a liability, other than (i) those liabilities provided for in the Company's financial statements and (ii) other undisclosed liabilities which, individually or in the aggregate, could not have, or reasonably be expected to result in, a Material Adverse Effect.
 - (dd) Real Property Holding Company. The Company is not a United States real property holding company.
- (ee) <u>Additional Shares Listing Application</u>. The Company has filed an Additional Shares Listing Application and all necessary related documents with the Nasdaq Stock Market in accordance with Nasdaq Rule 4310(c)(17)(D).
- (ff) <u>Disclosure</u>. The Company confirms that as of the time the Company files the press release and Current Report on Form 8-K contemplated under Section 4.8, the Investor will not be in possession of material non-public information concerning the Company or its business as a result of any information delivered to such Investor or its advisor by the Company or any Person on its behalf. The Company understands and confirms that the Investors will rely on the foregoing representations and covenants in effecting transactions in securities of the Company. All disclosure provided to the Investors regarding the Company, its business and the transactions contemplated hereby, furnished by or on behalf of the Company (including the Company's representations and warranties set forth in this Agreement), together with all other information provided or available in the SEC Reports, are true and correct in all material respects and do not contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading.

Each of the Investors acknowledges and agrees that the Company has not made and is not making any representations or warranties with respect to the transactions contemplated hereby other than those specifically set forth in this Section 3.1.

- 3.2 <u>Representations and Warranties of the Investors</u>. Each Investor hereby, for itself and for no other Investor, represents and warrants to the Company as follows:
 - (a) <u>Organization; Authority</u>. Such Investor is an entity duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization with the requisite corporate or partnership power and authority to enter into and to consummate the

transactions contemplated by the applicable Transaction Documents and otherwise to carry out its obligations thereunder. The execution, delivery and performance by such Investor of the transactions contemplated by this Agreement have been duly authorized by all necessary corporate or, if such Investor is not a corporation, such partnership, limited liability company or other applicable like action, on the part of such Investor. Each of this Agreement, the Registration Rights Agreement and the Subordination Agreement has been duly executed by such Investor, and when delivered by such Investor in accordance with terms hereof, will constitute the valid and legally binding obligation of such Investor, enforceable against it in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, liquidation or similar laws relating to, or affecting generally the enforcement of, creditors' rights and remedies or by other equitable principles of general application.

- (b) <u>Investment Intent</u>. Such Investor is acquiring the Securities as principal for its own account for investment purposes only and not with a view to or for distributing or reselling such Securities or any part thereof, without prejudice, however, to such Investor's right at all times to sell or otherwise dispose of all or any part of such Securities in compliance with applicable federal and state securities laws. Subject to the immediately preceding sentence, nothing contained herein shall be deemed a representation or warranty by such Investor to hold the Securities for any period of time. Such Investor is acquiring the Securities hereunder in the ordinary course of its business. Such Investor does not have any agreement or understanding, directly or indirectly, with any Person to distribute any of the Securities.
- (c) <u>Investor Status</u>. Such Investor is an "accredited investor" as defined in Rule 501(a) under the Securities Act. Such Investor is not a registered broker-dealer under Section 15 of the Exchange Act.
- (d) <u>General Solicitation</u>. Such Investor is not purchasing the Securities as a result of any advertisement, article, notice or other communication regarding the Securities published in any newspaper, magazine or similar media or broadcast over television or radio or presented at any seminar or any other general solicitation or general advertisement.
- (e) Access to Information. Such Investor acknowledges that it has reviewed the Disclosure Materials and has been afforded (i) the opportunity to ask such questions as it has deemed necessary of, and to receive answers from, representatives of the Company concerning the terms and conditions of the offering of the Securities and the merits and risks of investing in the Securities; (ii) access to information about the Company and the Subsidiaries and their respective financial condition, results of operations, business, properties, management and prospects sufficient to enable it to evaluate its investment; and (iii) the opportunity to obtain such additional information that the Company possesses or can acquire without unreasonable effort or expense that is necessary to make an informed investment decision with respect to the investment. Neither such inquiries nor any other investigation conducted by or on behalf of such Investor or its representatives or counsel shall modify, amend or affect such Investor's right to rely on the truth, accuracy and completeness of the Disclosure Materials and the Company's

representations and warranties contained in the Transaction Documents. Such Investor understands that its investment in the Securities involves a high degree of risk. Each Investor is able to bear the risk of an investment in the Securities including, without limitation, the risk of total loss of its investment. Such Investor has sought such accounting, legal and tax advice as it has considered necessary to make an informed investment decision with respect to its acquisition of the Securities.

- (f) <u>Certain Trading Activities</u>. Such Investor has not directly or indirectly, nor has any Person acting on behalf of or pursuant to any understanding with such Investor, engaged in any purchases or sales in the securities of the Company (including, without limitations, any Short Sales involving the Company's securities) since the date on which such Investor entered into a confidentiality or similar agreement with the Company in connection with such Investor's participation in the transactions contemplated by this Agreement. Such Investor covenants that neither it nor any Person acting on its behalf or pursuant to any understanding with it will engage in any purchases or sales in the securities of the Company (including Short Sales) prior to the time that the transactions contemplated by this Agreement are publicly disclosed. Notwithstanding the foregoing, in the case of an Investor that is a multi-managed investment vehicle whereby separate portfolio managers manage separate portions of such Investor's assets and the portfolio managers have no actual knowledge of the investment decisions made by the portfolio managers managing other portions of such Investor's assets, the representation set forth above shall only apply with respect to the portion of assets managed by the portfolio manager that made the investment decision to purchase the Securities covered by this Agreement.
- (g) <u>Limited Ownership</u>. The purchase by such Investor of the Securities issuable to it at the Closing will not result in such Investor (individually or together with other Person with whom such Investor has identified, or will have identified, itself as part of a "group" in a public filing made with the Commission involving the Company's securities) acquiring, or obtaining the right to acquire, in excess of 19.999% of the outstanding shares of Common Stock or the voting power of the Company on a post transaction basis that assumes that the Closing shall have occurred and taking into affect the conversion caps contained in the Notes which would restrict such Investor's ability to exceed its conversion rights under the Notes. Such Investor does not presently intend to, alone or together with others, make a public filing with the Commission to disclose that it has (or that it together with such other Persons have) acquired, or obtained the right to acquire, as a result of the Closing (when added to any other securities of the Company that it or they then own or have the right to acquire), in excess of 19.999% of the outstanding shares of Common Stock or the voting power of the Company on a post transaction basis that assumes that the Closing shall have occurred.
- (h) <u>Independent Investment Decision</u>. Such Investor has independently evaluated the merits of its decision to purchase Securities pursuant to this Agreement, and such Investor confirms that it has not relied on the advice of any other Investor's business and/or legal counsel in making such decision. If such Investor is other than Steelhead, such Investor

represents and warrants that Bryan Cave LLP has not acted as its legal counsel in connection with the transactions contemplated by this Agreement.

The Company acknowledges and agrees that no Investor has made or makes any representations or warranties with respect to the transactions contemplated hereby other than those specifically set forth in this Section 3.2.

ARTICLE IV. OTHER AGREEMENTS OF THE PARTIES

4.1 (a) The Securities may only be disposed of in compliance with state and federal securities laws. In connection with any transfer of the Securities other than pursuant to an effective registration statement, to the Company, to an Affiliate of a Investor or in connection with a pledge as contemplated in Section 4.1(b), the Company may require the transferor thereof to provide to the Company an opinion of counsel selected by the transferor, the form and substance of which opinion shall be reasonably satisfactory to the Company, to the effect that such transfer does not require registration of such transferred Securities under the Securities Act.

(b) Certificates evidencing the Securities will contain the following legend, until such time as they are not required under Section 4.1(c): [NEITHER THESE SECURITIES NOR THE SECURITIES ISSUABLE UPON CONVERSION OR EXERCISE OF THESE SECURITIES HAVE BEEN REGISTERED] [THESE SECURITIES HAVE NOT BEEN REGISTERED] WITH THE SECURITIES AND EXCHANGE COMMISSION OR THE SECURITIES COMMISSION OF ANY STATE IN RELIANCE UPON AN EXEMPTION FROM REGISTRATION UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), AND, ACCORDINGLY, MAY NOT BE OFFERED OR SOLD EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR PURSUANT TO AN AVAILABLE EXEMPTION FROM, OR IN A TRANSACTION NOT SUBJECT TO, THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT AND IN ACCORDANCE WITH APPLICABLE STATE SECURITIES LAWS AS EVIDENCED BY A LEGAL OPINION OF COUNSEL TO THE TRANSFEROR TO SUCH EFFECT, THE SUBSTANCE OF WHICH SHALL BE REASONABLY ACCEPTABLE TO THE COMPANY. [THESE SECURITIES AND THE SECURITIES ISSUABLE UPON CONVERSION OR EXERCISE OF THESE SECURITIES] [THESE SECURITIES] MAY BE PLEDGED IN CONNECTION WITH A BONA FIDE MARGIN ACCOUNT SECURED BY SUCH SECURITIES.

The Company acknowledges and agrees that an Investor may from time to time pledge, and/or grant a security interest in some or all of the Securities pursuant to a bona fide margin agreement in connection with a bona fide margin account and, if required under the terms

of such agreement or account, such Investor may transfer pledged or secured Securities to the pledgees or secured parties. Such a pledge or transfer would not be subject to approval or consent of the Company and no legal opinion of legal counsel to the pledgee, secured party or pledgor shall be required in connection with the pledge, but such legal opinion may be required in connection with a subsequent transfer following default by the Investor transferee of the pledge. No notice shall be required of such pledge. At the appropriate Investor's expense, the Company will execute and deliver such reasonable documentation as a pledgee or secured party of Securities may reasonably request in connection with a pledge or transfer of the Securities including the preparation and filing of any required prospectus supplement under Rule 424(b)(3) of the Securities Act or other applicable provision of the Securities Act to appropriately amend the list of Selling Stockholders thereunder.

- (c) Certificates evidencing Underlying Shares shall not contain any legend (including the legend set forth in Section 4.1(b)): (i) while a registration statement (including the Registration Statement) covering such Underlying Shares is then effective, or (ii) following a sale or transfer of such Securities pursuant to Rule 144 (assuming the transferor is not an Affiliate of the Company), or (iii) while such Securities are eligible for sale under Rule 144(k). The Company may not make any notation on its records or give instructions to any transfer agent of the Company that enlarge the restrictions on transfer set forth in this Section. The Company agrees that following such time as restrictive legends would not then be required under this Section 4.1(c), it shall promptly (but in no event later than five Trading Days after surrender of legended certificates to the Company) cause certificates evidencing the Underlying Shares previously issued to be replaced with certificates which do not bear the restrictive legends specified in this Section 4.1(c), and all Underlying Shares subsequently issued shall not bear the restrictive legend specified in this Section 4.1(c).
- (d) Without the prior consent of the Company (which will not be unreasonably withheld or delayed), the Notes may not be transferred (A) in increments of less than the lesser of (x) \$2 million and (y) the remaining principal amount of the Notes held by the transferring Investor or (B) to Persons (1) identified as "competitors" of the Company or (2) engaged in, or who have threatened in writing to engage in, a material Action against the Company, in each case of (1) and (2) as disclosed in Schedule 4.1(d) or in periodic reports of the Company filed by the Company with the Commission under the Exchange Act. The foregoing prohibitions shall not apply to transfers of Underlying Shares and shall not apply upon the occurrence of an Event of Default.
- 4.2 <u>Furnishing of Information</u>. As long as any Investor owns the Securities, the Company covenants to timely file (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to the Exchange Act. As long as any Investor owns Securities, if the Company is not required to file reports pursuant to such laws, it will prepare and furnish to the Investors and make publicly available in accordance with Rule 144(c) such information as is required for the Investors to sell the Underlying Shares under Rule 144. The Company further covenants that it will take such further action as any holder of Securities may reasonably request, all to the extent

required from time to time to enable such Person to sell the Underlying Shares without registration under the Securities Act within the limitation of the exemptions provided by Rule 144.

- 4.3 <u>Acknowledgment of Dilution</u>. The Company acknowledges that the issuance of Underlying Shares upon conversion of Notes will result in dilution of the outstanding shares of Common Stock, which dilution may be substantial. The Company further acknowledges that its obligation to honor conversions under the Notes is unconditional and absolute and not subject to any right of set off, counterclaim, delay or reduction, regardless of the effect of any such dilution or any claim that the Company may have against any Investor.
- 4.4 <u>Integration</u>. The Company shall not, and shall use its best efforts to ensure that no Affiliate of the Company shall, sell, offer for sale or solicit offers to buy or otherwise negotiate in respect of any security (as defined in Section 2 of the Securities Act) that would be integrated with the offer or sale of the Securities in a manner that would require the registration under the Securities Act of the sale of the Securities to the Investors, or that would be integrated with the offer or sale of the Securities for purposes of the rules and regulations of any Trading Market in a manner that would require stockholder approval of the sale of the securities to the Investors.
- 4.5 <u>Reservation of Shares</u>. The Company shall maintain a reserve from its duly authorized shares of Common Stock to comply with its conversion obligations under the Notes.
- 4.6 <u>Conversion Procedures</u>. The form of Conversion Notice included in and as defined in the Notes sets forth the totality of the procedures required by the Investors in order to convert the Notes. The Company shall honor conversions of the Notes and shall deliver Underlying Shares in accordance with the terms, conditions and time periods set forth in the Transaction Documents.
- 4.7 <u>Subsequent Registrations</u>. Other than pursuant to the Registration Statement, prior to the Effective Date, the Company may not file any registration statement with the Commission with respect to any securities of the Company other than registration statements on Form S-8 promulgated by the Commission or any amendment to the Company's registration statement on Form S-3, File No. 333-117077 (other than to increase the amount of securities registered thereunder).
- 4.8 <u>Securities Laws Disclosure</u>; <u>Publicity</u>. By 9:00 a.m. (New York time) on the Trading Day following the execution of this Agreement, and by 9:00 a.m. (New York time) on the Trading Day following the Closing Date, the Company shall issue press releases disclosing the transactions contemplated hereby and the Closing as well as any other information disclosed previously by the Company or Persons acting on its behalf to such Investor or its agents under any confidentiality or similar agreement and not subsequently disclosed to the public. On the Trading Day following the execution of this Agreement the Company will file a Current Report on Form 8-K disclosing the material terms of the Transaction Documents as well as any other information disclosed previously by the Company or Persons acting on its behalf to such Investor or its agents under any confidentiality or similar agreement and not subsequently disclosed to the

public (and attach as exhibits thereto the Transaction Documents), and on the Closing Date the Company will file an additional Current Report on Form 8-K to disclose the Closing. In addition, the Company will make such other filings and notices in the manner and time required by the Commission and the Trading Market on which the Common Stock is listed. Notwithstanding the foregoing, the Company shall not publicly disclose the name of any Investor, or include the name of any Investor in any filing with the Commission (other than the Registration Statement and any exhibits to filings made in respect of this transaction in accordance with periodic filing requirements under the Exchange Act) or any regulatory agency or Trading Market, without the prior written consent of such Investor, except to the extent such disclosure is required by law or Trading Market regulations.

- 4.9 <u>Limitation on Issuance of Future Priced Securities</u>. During the six months following the Closing Date, the Company shall not issue any "Future Priced Securities" as such term is described by NASD IM-4350-1.
- 4.10 <u>Indemnification of Investors</u>. In addition to the indemnity provided in the Registration Rights Agreement, the Company will indemnify and hold the Investors and their directors, officers, shareholders, partners, employees and agents (each, an "Investor Party") harmless from any and all losses, liabilities, obligations, claims, contingencies, damages, costs and expenses, including all judgments, amounts paid in settlements, court costs and reasonable attorneys' fees and costs of investigation (collectively, "Losses") that any such Investor Party may suffer or incur as a result of or relating to any misrepresentation, breach or inaccuracy of any representation, warranty, covenant or agreement made by the Company in any Transaction Document. In addition to the indemnity contained herein, the Company will reimburse each Investor Party for its reasonable legal and other expenses (including the cost of any investigation, preparation and travel in connection therewith) incurred in connection therewith, as such expenses are incurred.
- 4.11 <u>Right of First Refusal</u>. If at any time while the Notes are outstanding, the Company proposes to issue any Common Stock or Common Stock Equivalents (collectively, **"New Issue Securities"**), the Company shall first offer up to all of the New Issue Securities to the Investors in accordance with the following provisions:
 - (a) The Company shall give a written notice to each Investor (the "First Notice") stating (i) its intention to issue the New Issue Securities, (ii) the number and description of the New Issue Securities proposed to be issued, (iii) the purchase price (calculated as of the proposed issuance date) and the other terms and conditions upon which the Company is offering the New Issue Securities and (iv) the transaction documents in substantially definitive form pursuant to which the Company proposes to issue New Issue Securities.
 - (b) Transmittal of the First Notice to the Investors by the Company shall constitute an offer by the Company to sell to each Investor up to its proportionate number (based upon such Investor's pro rata portion of all Investment Amounts hereunder) of the New Issue Securities for the price and upon the terms and conditions set forth in the First Notice. For a

period of five (5) Trading Days after receipt of the of the First Notice, each Investor shall have the option, exercisable by written notice to the Company, to accept the Company's offer as to all or any part of such Investor's proportionate number of the New Issue Securities. If two or more types of New Issue Securities are to be issued or New Issue Securities are to be issued together with other types of securities, including, without limitation, debt securities, in a single transaction or related transactions, the rights to purchase New Issue Securities granted to the Investors under this Section may be exercised to purchase all types of New Issue Securities and such other securities in the same proportion as such New Issue Securities and other securities are to be issued by the Company. In the event that the Investors do not timely elect to purchase all of the New Issue Securities subject to the First Notice, the Company shall only be permitted to consummate a transaction for New Issue Securities for which such elections shall not have been so made on the terms and to the Persons specified in the First Notice. Notwithstanding anything to the contrary herein and unless otherwise agreed to by the Required Investors, the Company shall either confirm in writing to the Investors that the transaction with respect to the New Issue Securities has been abandoned or shall publicly disclose its intention to issue New Securities, in either case in such a manner such that Investors will not be in possession of material non-public information, by the tenth Trading Day following delivery of the First Notice. If by the tenth Trading Day following delivery of the First Notice no public disclosure regarding a transaction with respect to the New Issue Securities has been made, and no notice regarding the abandonment of such transaction has been received by the Investor, such transaction shall be deemed to have been abandoned and the Investor shall not be deemed to be in possession of any material, non-public information with respect to the Company. Should the Company decide to pursue such transaction with respect to the New Issue Securities, the Company shall provide each Investor with another First Notice and each Investor will again have the right of first refusal set forth in this Section. If the Investors indicate in the aggregate a willingness to provide financing in excess of the amount set forth in the First Notice, then each Investor will be entitled to provide financing pursuant to such First Notice up to an amount of all such proceeds equal to such Investor's pro rata portion of all Investment Amounts hereunder. The Company shall not be permitted to deliver more than one such First Notice to the Investors in any 45 day period.

- (c) Notwithstanding the foregoing, should the transaction with respect to the New Issue Securities occur as a result of the Company's engaging a nationally recognized underwriter or nationally recognized investment bank to solicit offers to purchase such New Issue Securities, then the Company shall be obligated to offer up to only 50% of the New Issue Securities in such a transaction to the Investors in accordance with the terms of this Section 4.11.
- (d) The rights contained in this Section shall not apply to the issuance and sale by the Company of (i) shares of Common Stock or Common Stock Equivalents to employees, officers, or directors of the Company, as compensation for their services to the Company or any of its direct or indirect Subsidiaries pursuant to arrangements approved by the Board of Directors of the Company and consistent with past practice or (ii) Securities pursuant to the Transaction Documents.

- 4.12 <u>Non-Public Information</u>. The Company covenants and agrees that neither it nor any other Person acting on its behalf will provide any Investor or its agents or counsel with any information that the Company believes constitutes material non-public information, unless prior thereto such Investor shall have executed a written agreement regarding the confidentiality and use of such information. The Company understands and confirms that each Investor shall be relying on the foregoing representations in effecting transactions in securities of the Company.
- 4.13 <u>Listing of Securities</u>. The Company agrees, (i) if the Company applies to have the Common Stock traded on any other Trading Market, it will include in such application the Underlying Shares, and will take such other action as is necessary or desirable to cause the Underlying Shares to be listed on such other Trading Market as promptly as possible, and (ii) for a period of not less than six years from the date hereof, it will take all action reasonably necessary to continue the listing and trading of its Common Stock on a Trading Market and will comply in all material respects with the Company's reporting, filing and other obligations under the bylaws or rules of the Trading Market.
- 4.14 <u>Use of Proceeds</u>. The Company will use the net proceeds from the sale of the Securities hereunder to repay Term Loan B under the Senior Credit Agreement in full and use the remainder for working capital purposes. The proceeds may not be used to redeem any Common Stock or Common Stock Equivalents.
- 4.15 <u>Payment of Cash Dividend</u>. The Company agrees, so long as any of the Notes are outstanding, not to declare, pay or make any provision for any cash dividend or distribution with respect to the Common Stock of the Company.
- 4.16 Existence; Conduct of Business. The Company will, and will cause each of the Subsidiaries to, do or cause to be done all things necessary to preserve, renew and keep in full force and effect its legal existence and the rights, licenses, permits, privileges and franchises material to the conduct of its business, provided that the foregoing shall not prohibit (a) any restructuring activity consistent with the Company's previously announced program to reduce its cost structure and improve its operating results, (b) any sale, lease, transfer or other disposition permitted by this Agreement, (c) any merger of (i) any domestic Subsidiary with any other domestic Subsidiary with any other foreign Subsidiary or (d) the dissolution of any inactive Subsidiary.
- 4.17 <u>Financial Ratios</u>. Commencing fifteen (15) months from the Closing Date and so long as any of the Notes are outstanding, the Company covenants and agrees that (i) the Leverage Ratio shall not at any time exceed 4.25:1.00 and (ii) (a) until and including the second anniversary of the Closing Date, the Cash Interest Coverage Ratio shall at all times exceed 2.5:1.00 and (b) after the second anniversary of the Closing Date, the Interest Coverage Ratio shall at all times exceed 2.5:1.00.
- 4.18 <u>Subordination Agreement</u>. Certain of the obligations of the Company to the Investors hereunder are subordinated to the Senior Obligations pursuant to, and to the extent provided in, the Subordination Agreement.

ARTICLE V. CONDITIONS PRECEDENT TO CLOSING

- 5.1 <u>Conditions Precedent to the Obligations of an Investor to Purchase Securities</u>. The obligation of each Investor to acquire Securities and make loans at the Closing is subject to the satisfaction or waiver by such Investor, at or before the Closing, of each of the following conditions:
 - (a) <u>Representations and Warranties</u>. The representations and warranties of the Company contained herein shall be true and correct as of the date when made and as of the Closing as though made on and as of such date;
 - (b) <u>Performance</u>. The Company shall have performed, satisfied and complied with all covenants, agreements and conditions required by the Transaction Documents to be performed, satisfied or complied with by it at or prior to the Closing;
 - (c) <u>Officer's Certificate</u>. A certificate executed by a duly authorized officer of the Company certifying that all representations and warranties made by the Company and information furnished by the Company in any schedules to this Agreement, are true and correct in all material respects as of the Closing Date, and all covenants, agreements and obligations required by this Agreement to be performed or complied with by the Company, prior to or at the Closing, have been performed or complied with in all material respects.
 - (d) <u>No Injunction</u>. No statute, rule, regulation, executive order, decree, ruling or injunction shall have been enacted, entered, promulgated or endorsed by any court or governmental authority of competent jurisdiction that prohibits the consummation of any of the transactions contemplated by the Transaction Documents;
 - (e) <u>Adverse Changes</u>. Since the date of execution of this Agreement, no event or series of events shall have occurred that has had or would reasonably be expected to result in a Material Adverse Effect:
 - (f) No Suspensions of Trading in Common Stock; Listing. Trading in the Common Stock shall not have been suspended by the Commission or any Trading Market (except for any suspensions of trading of not more than one Trading Day solely to permit dissemination of material information regarding the Company) at any time since the date of execution of this Agreement, and the Common Stock shall have been at all times since such date listed for trading on a Trading Market;
 - (g) Company Deliverables. The Company shall have delivered the Company Deliverables in accordance with Section 2.2(a);
 - (h) <u>Nasdaq Listing</u>. The Nasdaq Stock Market shall have provided a written interpretation to the Company that the issuance of the Underlying Shares does not require approval by the shareholders of the Company and shall have orally advised the Company that it

need not wait fifteen days following the filing of its Additional Shares Listing Application filed by the Company in connection with this transaction for the Closing to occur; and

- (i) <u>Estoppel Certificate</u>. JPMorgan Chase Bank, N.A. and the various Lenders party to the Senior Documents, shall have executed an Estoppel Certificate (in a form satisfactory to the Investors), which shall include, but not be limited to, such Lenders consent to the issuance of the Notes by the Company, such Lenders acknowledgement of the absence of any default pursuant to the Senior Documents and the acknowledgement of such Lenders that the maximum available revolving credit under the Senior Credit Facility, subject to the terms of the Senior Credit Agreement, is \$45,000,000.
- 5.2 <u>Conditions Precedent to the Obligations of the Company to sell Securities</u>. The obligation of the Company to sell Securities at the Closing is subject to the satisfaction or waiver by the Company, at or before the Closing, of each of the following conditions:
 - (a) <u>Representations and Warranties</u>. The representations and warranties of each Investor contained herein shall be true and correct as of the date when made and as of the Closing Date as though made on and as of such date;
 - (b) <u>Performance</u>. Each Investor shall have performed, satisfied and complied in all material respects with all covenants, agreements and conditions required by the Transaction Documents to be performed, satisfied or complied with by such Investor at or prior to the Closing;
 - (c) No Injunction. No statute, rule, regulation, executive order, decree, ruling or injunction shall have been enacted, entered, promulgated or endorsed by any court or governmental authority of competent jurisdiction that prohibits the consummation of any of the transactions contemplated by the Transaction Documents;
 - (d) Investors Deliverables. Each Investor shall have delivered its Investors Deliverables in accordance with Section 2.2(b); and
 - (e) <u>Nasdaq Listing</u>. The Nasdaq Stock Market shall have provided a written interpretation to the Company that the issuance of the Underlying Shares does not require approval by the shareholders of the Company and shall have orally advised the Company that it need not wait fifteen days following the filing of its Additional Shares Listing Application filed by the Company in connection with this transaction for the Closing to occur.

ARTICLE VI. NEGATIVE COVENANTS

The Company hereby agrees that, from and after the date hereof and so long as any of the Notes are outstanding, the Company shall be bound according to the restrictions set forth in each of following negative covenants unless any such restriction shall have been expressly waived in writing by the Required Investors:

- 6.1 <u>Restrictions on Certain Amendments</u>. The Company will not amend the rights and privileges granted under the Notes, to adversely affect the rights or privileges granted under the Notes.
 - 6.2 Restricted Payment. The Company shall not make any Restricted Payment.
- 6.3 <u>Debt</u>. (a) Commencing on the Closing Date and continuing to and including the one year anniversary of the Closing Date, neither the Company nor any Subsidiary shall create, incur, assume, become or be liable in any manner in respect of, or suffer to exist, any Debt or obligations under any operating lease, except (i) Debt or Contingent Liabilities in existence on the date hereof, as shown on <u>Schedule 6.3</u>; (ii) Permitted Indebtedness (as defined below); (iii) Senior Obligations permitted under the Senior Credit Agreement or the refinancing or replacement thereof, including new and replacement letters of credit, all in an amount not to exceed \$90,000,000; (iv) capital and operating leases to the extent permitted under the Senior Credit Agreement, or the refinancing or replacement thereof, but, notwithstanding anything to the contrary in the Senior Credit Agreement, not to exceed \$1,000,000 outstanding at any time with respect to capital leases and not to exceed \$15,000,000 outstanding at any time with respect to operating leases; (v) the transaction described in Schedule 6.6; and (vi) Debt under the Company's performance and bonding line, not to exceed \$150,000,000.
 - (b) Commencing one day following the one year anniversary of the Closing Date, neither the Company nor any Subsidiary shall create, incur, assume, become or be liable in any manner in respect of, or suffer to exist, any Debt or obligations under any operating lease, except (i) Contingent Liabilities resulting from the endorsement of negotiable instruments for collection in the ordinary course of business; (ii) Debt incurred to finance the acquisition of fixed or capital assets (whether pursuant to a loan, capital lease obligation or otherwise) in an aggregate principal amount not to exceed one million dollars (\$1,000,000), unless the written consent of the Required Investors is obtained, at any time outstanding, provided that such Debt is incurred simultaneously with such acquisition (the "Purchase Money Financing") (collectively (i) and (ii) shall be referred to as the "Permitted Indebtedness"); (iii) Senior Obligations permitted under the Senior Credit Agreement or the refinancing or replacement thereof, including new and replacement letters of credit, all in an amount not to exceed \$90,000,000; (iv) capital and operating leases to the extent permitted under the Senior Credit Agreement, or the refinancing or replacement thereof, but, notwithstanding anything to the contrary in the Senior Credit Agreement, not to exceed \$1,000,000 outstanding at any time with respect to capital leases and not to exceed \$15,000,000 outstanding at any time with respect to operating leases; (v) the transaction described in Schedule 6.6; and (vi) Debt under the Company's performance and bonding line, not to exceed \$150,000,000.
 - (c) The Company shall not, and the Company shall not permit any of its Subsidiaries to, directly or indirectly, incur or guarantee, assume or suffer to exist any Debt which shall rank senior to, or *pari passu* with, the Notes, other than Senior Obligations and Permitted Indebtedness.

- 6.4 <u>Liens</u>. The Company shall not create or suffer to exist any Lien upon any of its properties, except (a) Liens in existence on the date hereof, as shown on <u>Schedule 6.4</u>, (b) Permitted Liens, (c) Liens securing the Senior Documents and (d) Liens securing Purchase Money Financing up to \$1,000,000 (unless the written consent of the Required Investors is obtained), <u>provided</u>, that (x) any Lien securing any Purchase Money Financing shall be created substantially simultaneously with the acquisition of such fixed or capital asset, (y) such Liens do not at any time encumber any property other than the property that is the subject of the Purchase Money Financing, and (z) the principal amount of Debt secured by any such Purchase Money Financing shall at no time exceed one hundred percent (100%) of the original purchase price of such property at the time it was acquired. Except as provided in this Section 6.4, the Company shall not hereafter agree with any Person (other than the Investors) to grant a Lien on any of its assets or to permit the pledge of any of its equity interests.
- 6.5 <u>Amendment of Organizational Documents</u>. The Company shall not permit any amendment to its articles of incorporation so as to adversely affect the rights or privileges granted under the Notes.
- 6.6 <u>Sale and Leaseback</u>. Except as set forth in <u>Schedule 6.6</u>, neither the Company nor any Subsidiary shall enter into any arrangement whereby it sells or transfers any of its assets, and thereafter rents or leases such assets.
- 6.7 <u>Business</u>. The Company shall not change the nature of the business as now conducted by the Company and the Subsidiaries; <u>provided</u>, that the foregoing shall not prohibit (a) any restructuring activity consistent with the Company's previously announced program to reduce its cost structure and improve its operating results, (b) any sale, lease, transfer or other disposition permitted by this Agreement, (c) any merger of (i) any domestic Subsidiary with any other domestic Subsidiary, (ii) any domestic Subsidiary with and into the Company, or (iii) any foreign Subsidiary with any other foreign Subsidiary or (d) the dissolution of any inactive Subsidiary.
- 6.8 <u>Transactions with Affiliates</u>. Neither the Company nor any Subsidiary shall, directly or indirectly, pay any funds to or for the account of, make any investment (whether by acquisition of stock or indebtedness, by loan, advance, transfer of property, guarantee or other agreement to pay, purchase or service, directly or indirectly, any Debt, or otherwise) in, lease, sell, transfer or otherwise dispose of any assets, tangible or intangible, to, or participate in, or effect any transaction in connection with any joint enterprise or other joint arrangement with, any Affiliate, except, on terms no less favorable than terms that could be obtained by the Company or such Subsidiary from a Person that is not an Affiliate of the Company or such Subsidiary upon negotiation at arms' length, as determined in good faith by the Board; provided that no determination of the Board of Directors shall be required with respect to any such transactions entered into in the ordinary course of business.
- 6.9 <u>Limitation on Restrictions</u>. Neither the Company nor any Subsidiary shall enter into, or suffer to exist, any agreement with any Person (other than the Subordination Agreement)

which prohibits or limits the ability of the Company or any Subsidiary to (a) pay Debt owed to the Investors, (b) make loans or advances to the Company or (c) transfer any of its properties or assets to the Company.

- 6.10 <u>Stock Buy-Backs</u>. At any time while the Notes are outstanding, without the written consent of all Investors, the Company may not engage in any purchase of Common Stock or Common Stock Equivalents, except to satisfy its obligations pursuant to any grant or award under its stock plans. If the Company wishes to obtain such consent from the Investors, it shall do so in a manner so as to not provide any Investor with material, non-public information regarding the Company.
- 6.11 <u>Restrictions on Issuance of Additional Equity</u>. At any time while the Notes are outstanding, the Company may not issue any Common Stock or Common Stock Equivalents without the consent of the Required Investors, unless such issuance is pursuant to (i)(A) an underwritten public offering (which shall not include equity lines of credit or similar transactions) with proceeds of at least \$10,000,000 and (B) conducted by a nationally recognized underwriter, (ii) a private placement transaction conducted by a nationally recognized investment bank, or (iii) the Company's stock plans.

ARTICLE VII. MISCELLANEOUS

- 7.1 Fees and Expenses. At the Closing, the Company shall pay to Bryan Cave LLP and Weil, Gotshal & Manges LLP an aggregate of \$70,000 as partial reimbursement of Steelhead for its legal fees in connection with the Transaction Documents (Steelhead may deduct such amount from the Investment Amount deliverable to the Company at Closing), it being understood that Bryan Cave LLP and Weil, Gotshal & Manges LLP has only rendered legal advice to Steelhead, and not to the Company or any other Investor in connection with the transactions contemplated hereby, and that each of the Company and the other Investors has relied for such matters on the advice of its own respective counsel. Except as specified in the immediately preceding sentence, each party shall pay the fees and expenses of its advisers, counsel, accountants and other experts, if any, and all other expenses incurred by such party incident to the negotiation, preparation, execution, delivery and performance of the Transaction Documents. The Company shall pay all stamp and other taxes and duties levied in connection with the sale of the Notes.
- 7.2 Entire Agreement. The Transaction Documents, together with the Exhibits and Schedules thereto, contain the entire understanding of the parties with respect to the subject matter hereof and supersede all prior agreements, understandings, discussions and representations, oral or written, with respect to such matters, which the parties acknowledge have been merged into such documents, exhibits and schedules.
- 7.3 Notices. Any and all notices or other communications or deliveries required or permitted to be provided hereunder shall be in writing and shall be deemed given and effective on the earliest of (a) the date of transmission, if such notice or communication is delivered via

facsimile (provided the sender receives a machine-generated confirmation of successful transmission) at the facsimile number specified in this Section prior to 6:30 p.m. (New York City time) on a Trading Day, (b) the next Trading Day after the date of transmission, if such notice or communication is delivered via facsimile at the facsimile number specified in this Section on a day that is not a Trading Day or later than 6:30 p.m. (New York City time) on any Trading Day, (c) the Trading Day following the date of mailing, if sent by U.S. nationally recognized overnight courier service, or (d) upon actual receipt by the party to whom such notice is required to be given. The address for such notices and communications shall be as follows:

If to the Company: Matrix Service Company

10701 E. Ute Street Tulsa, OK 74116-1517 Facsimile: (918) 838-8810 Attention: George L. Austin

With a copy to: Conner & Winters, LLP

3700 First Place Tower 15 East 5th Street Tulsa, OK 74103-4344 Facsimile: (918) 586-8548 Attention: Mark D. Berman, Esq.

If to an Investor: To the address set forth under such Investor's name

on the signature pages hereof;

With a copy to: Bryan Cave LLP

1290 Avenue of the Americas New York, NY 10104 Facsimile: (212) 541-1432 Attention: Eric L. Cohen, Esq.

or such other address as may be designated in writing hereafter, in the same manner, by such Person.

7.4 Amendments; Waivers; No Additional Consideration. No provision of this Agreement may be waived or amended except in a written instrument signed by the Company and the Required Investors. No waiver of any default with respect to any provision, condition or requirement of this Agreement shall be deemed to be a continuing waiver in the future or a waiver of any subsequent default or a waiver of any other provision, condition or requirement hereof, nor shall any delay or omission of either party to exercise any right hereunder in any manner impair the exercise of any such right. No consideration shall be offered or paid to any Investor to amend or consent to a waiver or modification of any provision of any Transaction

Document unless the same consideration is also offered to all Investors who then hold Notes. Without the written consent or the affirmative vote of each Holder of Securities affected thereby, an amendment or waiver under this Section 7.4 may not:

- (a) change the maturity of the principal amount of, or the interest payment date under, or the payment of liquidated damages, is due on, any Note;
- (b) make any change that impairs the conversion or exercise rights of the Notes;
- (c) reduce the Event Equity Value under the Notes or amend or modify in any manner adverse to the Holders of the Notes the Company's obligation to make such payments;
 - (d) amend the definition of Required Investors;
 - (e) change the currency of any amount owed or owing under the Securities or any interest thereon from U.S. Dollars;
 - (f) impair the right of any Investor to institute suit for the enforcement of any payment with respect to, or conversion or exercise of, any Security; or
 - (g) modify the provisions of this Section 7.4 or Section 7.5.

It shall not be necessary for the consent of the Holders under this Section 7.4 to approve the particular form of any proposed amendment, but it shall be sufficient if such consent approves the substance thereof.

- 7.5 Termination. This Agreement may be terminated prior to Closing:
 - (a) by written agreement of the Investors and the Company;
- (b) by the Company or an Investor (as to itself but no other Investor) upon written notice to the other, if the Closing shall not have taken place by 6:30 p.m. Eastern time on the Outside Date; <u>provided</u>, that the right to terminate this Agreement under this Section 7.5(b) shall not be available to any Person whose failure to comply with its obligations under this Agreement has been the cause of or resulted in the failure of the Closing to occur on or before such time. The failure of the Company to satisfy the condition precedent contained in Section 5.1(g) or 5.2(d) by the Outside Date shall not be grounds for the Company to terminate this Agreement.
- (c) by an Investor (as to itself but no other Investor) if it concludes in good faith that any of the conditions precedent contained in Section 5.1(d), (e) or (f) shall have been breached or shall not be capable of being satisfied by the Outside Date despite the assumed best efforts of the Company.

In the event of a termination pursuant to this Section, the Company shall promptly notify all non-terminating Investors and shall pay to Steelhead all of the fees and expenses incurred by it (including reasonable legal fees and expenses) in connection with this Agreement and the transactions contemplated by this Agreement through the termination date in an amount not to exceed \$100,000 in the aggregate. Other than as to the foregoing fees and expenses, upon a termination in accordance with this Section 7.5, the Company and the terminating Investor(s) shall not have any further obligation or liability (including as arising from such termination) to the other and no Investor will have any liability to any other under the Transaction Documents Investor as a result therefrom.

- 7.6 <u>Construction</u>. The headings herein are for convenience only, do not constitute a part of this Agreement and shall not be deemed to limit or affect any of the provisions hereof. The language used in this Agreement will be deemed to be the language chosen by the parties to express their mutual intent, and no rules of strict construction will be applied against any party. This Agreement shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provisions of this Agreement or any of the Transaction Documents.
- 7.7 <u>Successors and Assigns</u>. This Agreement shall be binding upon and inure to the benefit of the parties and their successors and permitted assigns. The Company may not assign this Agreement or any rights or obligations hereunder without the prior written consent of the Investors. Any Investor may assign any or all of its rights under this Agreement to any Person to whom such Investor assigns or transfers any Notes, provided such transferee agrees in writing to be bound, with respect to the transferred Securities, by the provisions hereof that apply to the "Investors."
- 7.8 No Third-Party Beneficiaries. This Agreement is intended for the benefit of the parties hereto and their respective successors and permitted assigns and is not for the benefit of, nor may any provision hereof be enforced by, any other Person, except as otherwise set forth in Section 4.10 (as to each Investor Party).
- 7.9 Governing Law. All questions concerning the construction, validity, enforcement and interpretation of this Agreement shall be governed by and construed and enforced in accordance with the laws of the State of Delaware. Each party agrees that all Proceedings concerning the interpretations, enforcement and defense of the transactions contemplated by this Agreement and any other Transaction Documents (whether brought against a party hereto or its respective Affiliates, employees or agents) shall be commenced exclusively in the Delaware Courts. Each party hereto hereby irrevocably submits to the exclusive jurisdiction of the Delaware Courts for the adjudication of any dispute hereunder or in connection herewith or with any transaction contemplated hereby or discussed herein (including with respect to the enforcement of the any of the Transaction Documents), and hereby irrevocably waives, and agrees not to assert in any Proceeding, any claim that it is not personally subject to the jurisdiction of any such Delaware Court, or that such Proceeding has been commenced in an improper or inconvenient forum. Each party hereto hereby irrevocably waives personal service

of process and consents to process being served in any such Proceeding by mailing a copy thereof via registered or certified mail or overnight delivery (with evidence of delivery) to such party at the address in effect for notices to it under this Agreement and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law. Each party hereto hereby irrevocably waives, to the fullest extent permitted by applicable law, any and all right to trial by jury in any legal proceeding arising out of or relating to this Agreement or the transactions contemplated hereby. If either party shall commence a Proceeding to enforce any provisions of a Transaction Document, then the prevailing party in such Proceeding shall be reimbursed by the other party for its reasonable attorneys' fees and other costs and expenses incurred with the investigation, preparation and prosecution of such Proceeding.

- 7.10 Survival. The representations, warranties, agreements and covenants contained herein shall survive the Closing and the delivery of the Securities.
- 7.11 Execution. This Agreement may be executed in two or more counterparts, all of which when taken together shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other party, it being understood that both parties need not sign the same counterpart. In the event that any signature is delivered by facsimile transmission, such signature shall create a valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile signature page were an original thereof.
- 7.12 <u>Severability</u>. If any provision of this Agreement is held to be invalid or unenforceable in any respect, the validity and enforceability of the remaining terms and provisions of this Agreement shall not in any way be affected or impaired thereby and the parties will attempt to agree upon a valid and enforceable provision that is a reasonable substitute therefor, and upon so agreeing, shall incorporate such substitute provision in this Agreement.
- 7.13 <u>Rescission and Withdrawal Right</u>. Notwithstanding anything to the contrary contained in (and without limiting any similar provisions of) the Transaction Documents, whenever any Investor exercises a right, election, demand or option under a Transaction Document and the Company does not timely perform its related obligations within the periods therein provided, then such Investor may rescind or withdraw, in its sole discretion from time to time upon written notice to the Company, any relevant notice, demand or election in whole or in part without prejudice to its future actions and rights.
- 7.14 <u>Replacement of Securities</u>. If any certificate or instrument evidencing any Securities is mutilated, lost, stolen or destroyed, the Company shall issue or cause to be issued in exchange and substitution for and upon cancellation thereof, or in lieu of and substitution therefor, a new certificate or instrument, but only upon receipt of evidence reasonably satisfactory to the Company of such loss, theft or destruction and customary and reasonable indemnity, if requested. The applicants for a new certificate or instrument under such

circumstances shall also pay any reasonable third-party costs associated with the issuance of such replacement Securities. If a replacement certificate or instrument evidencing any Securities is requested due to a mutilation thereof, the Company may require delivery of such mutilated certificate or instrument as a condition precedent to any issuance of a replacement.

7.15 Remedies. In addition to being entitled to exercise all rights provided herein or granted by law, including recovery of damages, each of the Investors and the Company will be entitled to specific performance under the Transaction Documents. The parties agree that monetary damages may not be adequate compensation for any loss incurred by reason of any breach of obligations described in the foregoing sentence and hereby agrees to waive in any action for specific performance of any such obligation the defense that a remedy at law would be adequate.

7.16 <u>Payment Set Aside</u>. To the extent that the Company makes a payment or payments to any Investor pursuant to any Transaction Document or an Investor enforces or exercises its rights thereunder, and such payment or payments or the proceeds of such enforcement or exercise or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside, recovered from, disgorged by or are required to be refunded, repaid or otherwise restored to the Company, a trustee, receiver or any other person under any law (including, without limitation, any bankruptcy law, state or federal law, common law or equitable cause of action), then to the extent of any such restoration the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such enforcement or setoff had not occurred.

7.17 Independent Nature of Investors' Obligations and Rights. The obligations of each Investor under any Transaction Document are several and not joint with the obligations of any other Investor, and no Investor shall be responsible in any way for the performance of the obligations of any other Investor under any Transaction Document. The decision of each Investor to purchase Securities pursuant to the Transaction Documents has been made by such Investor independently of any other Investor. Nothing contained herein or in any Transaction Document, and no action taken by any Investor pursuant thereto, shall be deemed to constitute the Investors as a partnership, an association, a joint venture or any other kind of entity, or create a presumption that the Investors are in any way acting in concert or as a group with respect to such obligations or the transactions contemplated by the Transaction Documents. Each Investor acknowledges that no other Investor has acted as agent for such Investor in connection with making its investment hereunder and that no Investor will be acting as agent of such Investor in connection with monitoring its investment in the Securities or enforcing its rights under the Transaction Documents. Each Investor shall be entitled to independently protect and enforce its rights, including without limitation the rights arising out of this Agreement or out of the other Transaction Documents, and it shall not be necessary for any other Investor to be joined as an additional party in any proceeding for such purpose. The Company acknowledges that each of the Investors has been provided with the same Transaction Documents for the purpose of closing a transaction with multiple Investors and not because it was required or requested to do so by any Investor.

7.18 <u>Limitation of Liability</u>. Notwithstanding anything herein to the contrary, the Company acknowledges and agrees that the liability of an Investor arising directly or indirectly, under any Transaction Document of any and every nature whatsoever shall be satisfied solely out of the assets of such Investor, and that no trustee, officer, other investment vehicle or any other Affiliate of such Investor or any investor, shareholder or holder of shares of beneficial interest of such a Investor shall be personally liable for any liabilities of such Investor.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the parties hereto have caused this Securities Purchase Agreement to be duly executed by their respective authorized signatories as of the date first indicated above.

MATRIX SERVICE COMPANY

By: /s/ George L. Austin

Name: George L. Austin Title: Chief Financial Officer

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK SIGNATURE PAGES FOR INVESTORS FOLLOW]

IN WITNESS WHEREOF, the parties hereto have caused this Securities Purchase Agreement to b as of the date first indicated above.	e duly executed by their respective authorized signatories
NAME	OF INVESTOR

Ву:
Name:
Title:
Investment Amount: \$
Tax ID No.:
ADDRESS FOR NOTICE
c/o:
Street:
City/State/Zip:
Attention:
Tel:
Fax:
DELIVERY INSTRUCTIONS (if different from above)
c/o:
Street:
City/State/Zip:
Attention:
Tel:

EXHIBIT A

NEITHER THESE SECURITIES NOR THE SECURITIES ISSUABLE UPON CONVERSION OF THESE SECURITIES HAVE BEEN REGISTERED WITH THE SECURITIES AND EXCHANGE COMMISSION OR THE SECURITIES COMMISSION OF ANY STATE IN RELIANCE UPON AN EXEMPTION FROM REGISTRATION UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), AND, ACCORDINGLY, MAY NOT BE OFFERED OR SOLD EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR PURSUANT TO AN AVAILABLE EXEMPTION FROM, OR IN A TRANSACTION NOT SUBJECT TO, THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT AND IN ACCORDANCE WITH APPLICABLE STATE SECURITIES LAWS AS EVIDENCED BY A LEGAL OPINION OF COUNSEL TO THE TRANSFEROR TO SUCH EFFECT, THE SUBSTANCE OF WHICH SHALL BE REASONABLY ACCEPTABLE TO THE COMPANY. THESE SECURITIES AND THE SECURITIES ISSUABLE UPON CONVERSION OF THESE SECURITIES MAY BE PLEDGED IN A MANNER CONSISTENT WITH THE SECURITIES ACT IN CONNECTION WITH A BONA FIDE MARGIN ACCOUNT SECURED BY SUCH SECURITIES.

THE INDEBTEDNESS EVIDENCED OR SECURED BY THIS CONVERTIBLE NOTE IS SUBORDINATED TO THE SENIOR OBLIGATIONS (AS DEFINED IN THE SUBORDINATION AGREEMENT DESCRIBED BELOW) PURSUANT TO, AND TO THE EXTENT PROVIDED IN, THE SUBORDINATION AGREEMENT DATED AS OF APRIL 22, 2005, (THE "SUBORDINATION AGREEMENT") MADE BY THE SUBORDINATED CREDITOR AND OBLIGORS REFERRED TO THEREIN IN FAVOR OF JPMORGAN CHASE BANK, N.A. (SUCCESSOR BY MERGER TO BANK ONE, N.A. (MAIN OFFICE CHICAGO)), AS A LENDER, LC ISSUER AND AS AGENT FOR THE LENDERS UNDER THE CREDIT AGREEMENT DATED AS OF MAY 22, 2003, AS IT HAS BEEN AMENDED, AND IN FAVOR OF THE LENDERS PARTY TO THE CREDIT AGREEMENT.

No. [] \$[] Original Issue Date: April 25, 2005

MATRIX SERVICE COMPANY SENIOR UNSECURED CONVERTIBLE NOTE DUE APRIL 25, 2010

THIS SENIOR UNSECURED CONVERTIBLE NOTE is one of a series of duly authorized and issued notes of Matrix Service Company, a Delaware corporation (the "Company"), designated as its Senior Unsecured Convertible Notes due April 25, 2010, in the

original aggregate principal amount of thirty million dollars (\$30,000,000) (collectively, the "Notes" and each Note comprising the Notes, a "Note").

FOR VALUE RECEIVED, the Company promises to pay to the order of [_____] or its registered assigns (the "Investor"), the principal sum of [_____] (\$), on April 25, 2010 or such earlier date as this Note is required to be repaid as provided hereunder (the "Maturity Date"), and to pay interest to the Investor on the principal amount of this Note outstanding from time to time in accordance with the provisions hereof. All holders of Notes are referred to collectively, as the "Investors." This Note is subject to the following additional provisions:

1. <u>Definitions</u>. In addition to the terms defined elsewhere in this Note: (a) capitalized terms that are used but not otherwise defined herein have the meanings given to such terms in the Securities Purchase Agreement, dated as of April 22, 2005, among the Company and the Investors identified therein (the **"Purchase Agreement"**), and (b) the following terms have the meanings indicated below:

"Bankruptcy Event" means any of the following events: (a) the Company or any Subsidiary commences a case or other proceeding under any bankruptcy, reorganization, arrangement, adjustment of debt, relief of debtors, dissolution, insolvency or liquidation or similar law of any jurisdiction relating to the Company or any Subsidiary thereof; (b) there is commenced against the Company or any Subsidiary any such case or proceeding that is not dismissed within 60 days after commencement; (c) the Company or any Subsidiary is adjudicated by a court of competent jurisdiction insolvent or bankrupt or any order of relief or other order approving any such case or proceeding is entered; (d) the Company or any Subsidiary suffers any appointment of any custodian or the like for it or any substantial part of its property that is not discharged or stayed within 60 days; (e) under applicable law the Company or any Subsidiary makes a general assignment for the benefit of creditors; (f) the Company or any Subsidiary fails to pay, or states that it is unable to pay or is unable to pay, its debts generally as they become due; (g) the Company or any Subsidiary calls a meeting of its creditors with a view to arranging a composition, adjustment or restructuring of its debts; or (h) the Company or any Subsidiary, by any act or failure to act, expressly indicates its consent to, approval of or acquiescence in any of the foregoing or takes any corporate or other action for the purpose of effecting any of the foregoing.

"Change of Control" shall mean (a) any sale or disposition of all or substantially all of the assets of the Company to a third party in one or a number of related transactions, (b) any merger of the Company with or into another corporation in which the holders of the Company's Common Stock immediately prior to the consummation of the merger do not control 50% of the surviving entity, or (c) the acquisition in one or a number of related transactions by any Person or "group" of persons (as such term is defined in Section 13(d) and 14(d) of the Exchange Act, and the related regulations) who have expressed intent to control the affairs of the Company of more than 50% of the total voting power of outstanding voting securities of the Company.

"Closing Price" means, for any date, the price determined by the first of the following clauses that applies: (a) if the Common Stock is then listed or quoted on a Trading Market, the closing sale price per share of the Common Stock for such date (or the nearest preceding date) on the primary Trading Market or exchange on which the Common Stock is then listed or quoted;

(b) if prices for the Common Stock are then reported in the "Pink Sheets" published by the Pink Sheets, LLC (or a similar organization or agency succeeding to its functions of reporting prices), the most recent sale price per share of the Common Stock so reported; or (c) in all other cases, the fair market value of a share of Common Stock as determined by an independent qualified appraiser selected in good faith and paid for by a majority in interest of the Investors.

"Common Stock" means the common stock of the Company, \$0.01 par value per share, and any securities into which such common stock may hereafter be reclassified.

"Conversion Date" means the date a Conversion Notice together with the Conversion Schedule is delivered to the Company in accordance with Section 5(a).

"Conversion Notice" means a written notice in the form attached hereto as Exhibit A.

"Conversion Price" means \$4.69, subject to adjustment from time to time in accordance with Section 11.

"Debt" of any Person shall mean, without duplication, (a) all obligations of such Person for borrowed money or with respect to deposits or advances of any kind, (b) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (c) all obligations of such Person upon which interest charges are customarily paid, (d) all obligations of such Person under conditional sale or other title retention agreements relating to property or assets purchased by such Person, (e) all obligations of such Person issued or assumed as the deferred purchase price of property or services (other than unsecured accounts payable incurred in the ordinary course of business), (f) all Debt of others secured by (or for which the holder of such Debt has an existing right, contingent or otherwise, to be secured by) any Lien on property owned or acquired by such Person, whether or not the obligations secured thereby have been assumed, (g) all obligations of such Person in respect of interest rate protection agreements, foreign currency exchange agreements or other interest or exchange rate hedging arrangements that exceed amounts necessary to hedge the Company's cross-currency exposure, (h) all obligations of such Person as an account party in respect of letters of credit and bankers' acceptances. The Debt of any Person shall include the Debt of any partnership in which such Person is a general partner, (i) all obligations of such Person as lessee which (y) are capitalized in accordance with GAAP or (z) arise pursuant to sale-leaseback transactions and (j) all Debt of others guaranteed by such Person.

"Default" means any event or condition which constitutes an Event of Default or that upon notice, lapse of time or both would, unless cured or waived, become an Event of Default.

"Event Equity Value" means the average of the Closing Prices for the five consecutive Trading Days preceding either: (a) the date of an Event Notice or the date the Company becomes obligated to pay the Event Price under Section 7(b), as applicable, or (b) the date on which the Event Price with respect thereto (together with any other payments, expenses and liquidated damages then due and payable under the Transaction Documents) is paid in full, whichever is greater.

"Event of Default" means any one of the following events (whatever the reason and whether it shall be voluntary or involuntary or effected by operation of law or pursuant to any judgment, decree or order of any court, or any order, rule or regulation of any administrative or governmental body):

- (i) any default in the payment when the same becomes due and payable (whether on a Prepayment Date, the Maturity Date or by acceleration or prepayment or otherwise), of (a) liquidated damages in respect of this Note which default continues unremedied for a period of five Trading Days after the date on which written notice of such default is first given to the Company by the Investor, or (b) principal or interest in respect of this Note.
- (ii) the Company or any Subsidiary (1) fails to pay when due any monetary obligation other than those obligations set forth in Schedule 1 hereto (regardless of amount) under any currently existing or hereafter arising debenture (other than a Note) or any mortgage, credit agreement or other facility (including the Company's Senior Credit Facility), indenture agreement, factoring agreement or other instrument under which there may be issued, or by which there may be secured or evidenced, any Debt or under any long term leasing or factoring arrangement, if the aggregate amount of the obligations and liabilities of the Company and the Subsidiaries thereunder exceed \$1,000,000 (each of the foregoing a "Material Debt Agreement"), or (2) fails to observe or perform any other obligation under any Material Debt Agreement, and such failure results in the obligations thereunder becoming or being declared due and payable prior to the date on which they would otherwise become due and payable.
 - (iii) the occurrence or entering into by the Company or any Subsidiary, or consummation of, any Change of Control transaction.
- (iv) the Company shall fail to observe, satisfy, or perform any covenant, condition or agreement contained in any Transaction Document (other than those specified in clause (i) above and clause (xvi) below), and such failure shall continue unremedied for a period of 15 days after the date on which written notice of such default is first given to the Company by the Investor (it being understood that no prior notice need be given in the case of a default that cannot reasonably be cured within 15 days).
 - $\left(v\right)$ the occurrence and continuance of an Event of Default under any other Note.
- (vi) any prepayment by the Company of any other Note or any other Debt issued by it (except the Senior Obligations) or any issuance of securities in exchange for any Notes issued by it (other than Underlying Shares upon conversion of such Notes in accordance with their terms as in effect on the Original Issue Date thereof), except in each case (i) if the Company offers to the Investor in writing the same prepayment of this Note and all other Notes then held by such Investor on the same economic terms on which the Company prepays or offers to prepay (whichever is more favorable to the holder of such Note) such Notes, and (ii) in accordance with the prepayment provisions of Section 13 of this Note.
- (vii) any of the Company's representations and warranties set forth in the Purchase Agreement shall be incorrect in any material respect as of the Original Issue Date.
 - (viii) the occurrence of a Bankruptcy Event.

- (ix) one or more judgments for the payment of money in an aggregate amount in excess of \$1,000,000 shall be rendered against the Company or any Subsidiary or any combination thereof (which shall not be fully covered by insurance without taking into account any applicable deductibles) and which shall remain undischarged or unbonded for a period of 30 consecutive days during which execution shall not be effectively stayed, or any action shall be legally taken by a judgment creditor to attach or levy upon any assets of the Company or any Subsidiary to enforce any such judgment.
- (x) any Transaction Document shall cease, for any reason, to be in full force and effect, or the Company shall so assert in writing or shall disayow any of its obligations thereunder.
- (xi) the Common Stock shall not be listed or quoted, or is suspended from trading, on a Trading Market for a period of five Trading Days (which need not be consecutive Trading Days).
- (xii) the Company fails to deliver a stock certificate evidencing Underlying Shares to an Investor within five Trading Days after a Conversion Date, or the conversion rights of the Investors pursuant to the terms hereof are otherwise suspended for any reason (other than as a result of the limitations set forth in Section 5(b).
- (xiii) the Company fails to have available a sufficient number of authorized but unissued and otherwise unreserved shares of Common Stock available to issue Underlying Shares upon any conversion of Notes.
- (xiv) the Company effects or publicly announces its intention to effect any exchange, recapitalization or other transaction that effectively requires or rewards physical delivery of certificates evidencing the Common Stock, unless following such transaction, the holders of the Company's securities prior to the first such transaction continue to beneficially own at least two-thirds of the voting rights and equity interests in the surviving entity or acquirer of such assets.
- (xv) a Registration Statement under the Registration Rights Agreement is not declared effective by the Commission on or prior to the date that is 30 days after the Effectiveness Date, or is not effective as to all Registrable Securities (as defined in the Registration Rights Agreement), and available for use by the holders of Registrable Securities, for in excess of an aggregate of 20 Trading Days (which need not be consecutive Trading Days) in any 18-month period during the Effectiveness Period (as defined in the Registration Rights Agreement).
- (xvi) the Company fails to make any cash payment required under the Transaction Documents (other than as set forth in paragraph (i) above) and such failure is not cured within five Trading Days after notice of such default is first given to the Company by an Investor.
- (xvii) the Company fails to refinance its Senior Credit Facility on or before March 31, 2006 on terms substantially similar to, or more favorable to the Company, than those set forth in the Senior Documents; provided, however, that if on March 31, 2006 the

Company is in good faith negotiations to refinance the Senior Credit Facility on terms no less favorable to the Company than those set forth in the Senior Documents, an Event of Default pursuant to this subsection shall not be deemed to have occurred until such time as good faith negotiations have ceased without a refinancing of the Senior Credit Facility.

"Original Issue Date" has the meaning set forth on the face of this Note.

"Proceeding" means an action, claim, suit, investigation or proceeding (including, without limitation, an investigation or partial proceeding, such as a deposition), whether commenced or threatened.

"Trading Day" means (i) a day on which the Common Stock is traded on a Trading Market, or (ii) if the Common Stock is not quoted on a Trading Market, a day on which the Common Stock is quoted in the over-the-counter market as reported by the Pink Sheets, LLC (or any similar organization or agency succeeding to its functions of reporting prices); provided, that in the event that the Common Stock is not listed or quoted as set forth in (i) and (ii) hereof, then Trading Day shall mean a Business Day.

"Trading Market" means whichever of the New York Stock Exchange, the American Stock Exchange, the NASDAQ National Market, the NASDAQ SmallCap Market or OTC Bulletin Board on which the Common Stock is listed or quoted for trading on the date in question

"Underlying Shares" means the shares of Common Stock issuable upon conversion of the Notes and payment of interest thereunder.

2. Interest.

(a) The Company shall pay interest in cash to the Investor on the aggregate unconverted and then outstanding principal amount of this Note as follows: (a) on the Closing Date, interest shall be payable in advance at the rate of 7% per annum for the period from and including the Closing Date to and including the second anniversary of the Closing Date, which amount shall equal \$_____ and shall be non-refundable, and (b) thereafter, interest shall be payable in arrears on each March 31, June 30, September 30 and December 31 (each, an "Interest Payment Date"), beginning on June 30, 2007, at the rate of 7% per annum (5% per annum if the Leverage Ratio at all times during the two fiscal quarter period of the Company ending on the Interest Payment Date is less than 3:00 to 1:00 and no Default or Event of Default shall exist on the Interest Payment Date). If an Interest Payment Date is not a Trading Day, interest shall be payable on the next succeeding Trading Day. Interest shall be calculated on the basis of a 360-day year for the actual number of days elapsed and shall accrue daily commencing on the Original Issue Date. Concurrently with the filing by the Company of any Form 10-Q or Form 10-K, beginning with the filing of the Company's Form 10-K for the period ended May 31, 2007, the Company shall provide the Investor with a notice setting forth the Company's Leverage Ratio and an explanation of the calculations leading to such ratio for the period covered by such concurrently filed Form 10-Q or Form 10-K. Any such notice shall not contain any material, non-public information with respect to the Company.

- (b) If the Company fails to refinance its Senior Credit Facility on or before September 30, 2005 on terms substantially similar to, or more favorable to the Company, than those set forth in the Senior Documents, the interest rate payable on this Note shall, beginning on October 1, 2005, be increased by 5% per annum until the date the Senior Credit Facility is refinanced in accordance herewith (such 5% per annum increase, the "Additional Interest"). The Additional Interest shall be accreted to the principal amount then outstanding under this Note on each applicable Interest Payment Date and on such date as the Senior Credit Facility is refinanced. Any Additional Interest payable shall be in addition to any interest payable pursuant to Section 2(a) above.
- 3. <u>Registration of Notes</u>. The Company shall register the Notes upon records maintained by the Company for that purpose (the "**Note Register**") in the name of each record Investor thereof from time to time. The Company may deem and treat the registered Investor of this Note as the absolute owner hereof for the purpose of any conversion hereof or any payment of interest hereon, and for all other purposes, absent actual notice to the contrary from such record Investor.
- 4. <u>Registration of Transfers and Exchanges</u>. The Company shall register the transfer of any portion of this Note in the Note Register upon surrender of this Note to the Company at its address for notice set forth herein. Upon any such registration or transfer, a new Note, in substantially the form of this Note (any such new debenture, a "New Note"), evidencing the portion of this Note so transferred shall be issued to the transferee and a New Note evidencing the remaining portion of this Note not so transferred, if any, shall be issued to the transferring Investor. The acceptance of the New Note by the transferee thereof shall be deemed the acceptance by such transferee of all of the rights and obligations of a holder of a Note, including, without limitation, the rights and obligations described in the Subordination Agreement, and such transferee shall be a Subordinated Creditor under the Subordination Agreement. This Note is exchangeable for an equal aggregate principal amount of Notes of different authorized denominations, as requested by the Investor surrendering the same. No service charge or other fee will be imposed in connection with any such registration of transfer or exchange. The Company agrees that its prior consent is not required for the transfer of any portion of this Note, except that without the prior consent of the Company (which will not be unreasonably withheld or delayed), Notes may not be transferred (A) in increments of less than the lesser of (x) \$2 million and (y) the remaining principal amount of Notes held by the transferring Investor or (B) to Persons (1) identified as "competitors" of the Company or (2) engaged in, or who have threatened in writing to engage in, a material Action against the Company, in each case of (1) and (2) as disclosed in Schedule 4.1(d) to the Purchase Agreement or in periodic reports of the Company filed by the Company with the Commission under the Exchange Act. The foregoing prohibitions shall not apply to transfers of Underlying Shares.

5. Conversion.

(a) At the Option of the Investor. All or any portion of the principal amount of this Note then outstanding together with any accrued and unpaid interest hereunder shall be convertible into shares of Common Stock at the Conversion Price (subject to limitation or reduction set forth in Section 5(b)), at the option of the Investor, at any time and from time to time from and after the Original Issue Date. The Investor may effect conversions under this

Section 5(a), by delivering to the Company a Conversion Notice together with a schedule in the form of Annex 1 attached hereto (the "Conversion Schedule"). If the Investor is converting less than all of the principal amount represented by this Note, or if a conversion hereunder may not be effected in full due to the application of Section 5(b), the Company shall honor such conversion to the extent permissible hereunder and shall promptly deliver to the Investor a Conversion Schedule indicating the principal amount which has not been converted.

(b) Certain Conversion Restrictions.

- (i) Notwithstanding anything to the contrary contained herein, the number of shares of Common Stock that may be acquired by an Investor upon each conversion of Notes (or otherwise in respect hereof) shall be limited to the extent necessary to insure that, following such conversion (or other issuance), the total number of shares of Common Stock then beneficially owned by such Investor and its Affiliates and any other Persons whose beneficial ownership of Common Stock would be aggregated with such Investor's for purposes of Section 13(d) of the Exchange Act, does not exceed 9.9% of the total number of issued and outstanding shares of Common Stock (including for such purpose the shares of Common Stock issuable upon such conversion). For such purposes, beneficial ownership shall be determined in accordance with Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder. This provision shall not restrict the number of shares of Common Stock which an Investor may receive or beneficially own in order to determine the amount of securities or other consideration that such Investor may receive in the event of a Fundamental Transaction (defined below) involving the Company as contemplated herein. This restriction may not be waived.
- (ii) <u>Reduction in Underlying Shares Converted</u>. In the event an Investor delivers to the Company a Conversion Notice prior to the second anniversary of the Closing Date with respect to all or any portion of the principal amount of this Note then outstanding, then, at the Investor's option, (y) the Investor will pay the Company in immediately available funds the total amount of prepaid interest attributable to the principal amount to be converted multiplied by a fraction, the numerator of which is the number of days remaining from the Conversion Date until the second anniversary of the Closing Date and the denominator of which is the total number of days between the Closing Date and the second anniversary thereof, or (z) the number of Underlying Shares issuable upon such conversion under Section 6(a) hereof shall be reduced by a number of shares equal to the total amount of prepaid interest which is attributable to such principal amount converted divided by the closing price of the Common Stock on the Conversion Date, and multiplied by a fraction, the numerator of which is the number of days remaining until the second anniversary of the Closing Date and the denominator of which is the total number of days between the Closing Date and the second anniversary thereof.

6. Mechanics of Conversion.

(a) The number of Underlying Shares issuable upon any conversion hereunder shall equal the outstanding principal amount of this Note to be converted, divided by the Conversion Price on the Conversion Date, plus (if indicated in the applicable Conversion Notice) the amount of any accrued but unpaid interest on this Note through the Conversion Date, divided by the Conversion Price on the Conversion Date.

- (b) The Company shall, by the third Trading Day following each Conversion Date, issue or cause to be issued and cause to be delivered to or upon the written order of the Investor and in such name or names as the Investor may designate a certificate for the Underlying Shares issuable upon such conversion, free of restrictive legends if at such time a Registration Statement is then effective and available for use by the Investor. The Investor, or any Person so designated by the Investor to receive Underlying Shares, shall be deemed to have become holder of record of such Underlying Shares as of such Conversion Date. The Company shall use its best efforts to deliver Underlying Shares hereunder electronically (via a DWAC) through the Depository Trust Corporation or another established clearing corporation performing similar functions.
- (c) The Investor shall not be required to deliver the original Note in order to effect a conversion hereunder. Execution and delivery of the Conversion Notice shall have the same effect as cancellation of the Note and issuance of a New Note representing the remaining outstanding principal amount.
- (d) The Company's obligations to issue and deliver Underlying Shares upon conversion of this Note in accordance with the terms hereof are absolute and unconditional, irrespective of any action or inaction by the Investor to enforce the same, any waiver or consent with respect to any provision hereof, the recovery of any judgment against any Person or any action to enforce the same, or any setoff, counterclaim, recoupment, limitation or termination, or any breach or alleged breach by the Investor or any other Person of any obligation to the Company or any violation or alleged violation of law by the Investor or any other Person, and irrespective of any other circumstance which might otherwise limit such obligation of the Company to the Investor in connection with the issuance of such Underlying Shares.
- (e) If by the third Trading Day after a Conversion Date the Company fails to deliver to the Investor such Underlying Shares in such amounts and in the manner required pursuant to Section 5, then the Investor will have the right to rescind the Conversion Notice pertaining thereto by giving written notice to the Company prior to such Investor's receipt of such Underlying Shares.
- (f) If by the third Trading Day after a Conversion Date the Company fails to deliver to the Investor the required number of Underlying Shares in the manner required pursuant to Section 5, and if after such third Trading Day and prior to the receipt of such Underlying Shares, the Investor purchases (in an open market transaction or otherwise) shares of Common Stock to deliver in satisfaction of a sale by the Investor of the Underlying Shares which the Investor anticipated receiving upon such conversion (a "Buy-In"), then the Company shall: (1) pay in cash to the Investor (in addition to any other remedies available to or elected by the Investor) the amount by which (x) the Investor's total purchase price (including brokerage commissions, if any) for the shares of Common Stock so purchased exceeds (y) the amount obtained by multiplying (A) the number of Underlying Shares that the Company was required to deliver to the Investor in connection with the exercise at issue by (B) the Closing Price on the Conversion Date and (2) at the option of the Investor, either void the conversion at issue and reinstate the principal amount of Notes (plus accrued interest therein) for which such conversion was not timely honored or deliver to the Investor the number of shares of Common Stock that would have been issued had the Company timely complied with its exercise and delivery

obligations hereunder. The Investor shall provide the Company reasonably detailed evidence or written notice indicating the amounts payable to the Investor in respect of the Buy-In.

7. Events of Default.

- (a) At any time or times following the occurrence and during the continuance of an Event of Default (other than under clause (viii) of such defined term with respect to the Company), the Investor may elect, by notice to the Company (an "Event Notice"), to require the Company to purchase all or any portion of the outstanding principal amount of this Note, as indicated in such Event Notice, at a purchase price in Dollars in cash equal to the greater of:
 (A) 100% of such outstanding principal amount (except that such amount shall equal 110% in the case of an Event of Default under clause (iii) of the definition of "Event of Default"), plus all accrued but unpaid interest thereon and any unpaid liquidated damages and other amounts then owing to the Investor under the Transaction Documents, through the date of purchase, or (B) the Event Equity Value of the Underlying Shares that would be issuable upon conversion of such principal amount and payment in Common Stock of all such accrued but unpaid interest thereon (without regard to any condition precedent or conversion limitation contained herein). The aggregate amount payable pursuant to the preceding sentence is referred to as the "Event Price." The Company shall pay the aggregate Event Price to the Investor (free of any claim of subordination) no later than the third Trading Day following the date of delivery of the Event Notice, and upon receipt thereof the Investor shall deliver the original Note so repurchased to the Company.
- (b) Upon the occurrence of any Bankruptcy Event with respect to the Company, all outstanding principal and accrued but unpaid interest on this Note and any unpaid liquidated damages and other amounts then owing under the Transaction Documents shall immediately become due and payable in full in Dollars in cash (free of any claim of subordination), without any action by the Investor, and the Company shall immediately be obligated to repurchase this Note held by such Investor at the Event Price pursuant to the preceding paragraph as if the Investor had delivered an Event Notice immediately prior to the occurrence of such Bankruptcy Event.
- (c) In connection with any Event of Default, the Investor need not provide and the Company hereby waives any presentment, demand, protest or other notice of any kind (other than the Event Notice), and the Investor may immediately enforce any and all of its rights and remedies hereunder and all other remedies available to it under applicable law. Any such declaration may be rescinded and annulled by the Investor at any time prior to payment hereunder. No such rescission or annulment shall affect any subsequent Event of Default or impair any right consequent thereto.
- 8. <u>Ranking</u>. This Note ranks pari passu with all other Notes now or hereafter issued pursuant to the Transaction Documents and is senior in all respects to all existing and hereafter created unsecured Debt of the Company except as set forth in <u>Schedule 8</u>. The Investor acknowledges and agrees that certain Indebtedness which ranks *pari passu* with the Notes may be secured by Permitted Liens and, to the extent so secured, may effectively rank senior to the Notes to the extent of any security for such Indebtedness. The Company will not, directly or indirectly, enter into, create, incur, assume or suffer to exist any unsecured Debt of any kind, on

or with respect to any of its property or assets now owned or hereafter acquired or any interest therein or any income or profits therefrom, that is senior in any respect to the Company's obligations under the Notes. Notwithstanding the foregoing, this Section 8 shall not apply to any unsecured Debt that is part of or included within the Senior Obligations, including, but not limited to, any unsecured Debt arising from or as a result of 11 U.S.C. §506(a) or any provision of any plan of reorganization confirmed in any bankruptcy case of the Company.

- 9. <u>Charges, Taxes and Expenses</u>. Issuance of certificates for Underlying Shares upon conversion of (or otherwise in respect of) this Note shall be made without charge to the Investor for any issue or transfer tax, withholding tax, transfer agent fee or other incidental tax or expense in respect of the issuance of such certificate, all of which taxes and expenses shall be paid by the Company; <u>provided</u>, however, that the Company shall not be required to pay any tax which may be payable in respect of any transfer involved in the registration of any certificates for Underlying Shares or Notes in a name other than that of the Investor. The Investor shall be responsible for all other tax liability that may arise as a result of holding or transferring this Note or receiving Underlying Shares in respect hereof.
- 10. Reservation of Underlying Shares. The Company covenants that it will at all times reserve and keep available out of the aggregate of its authorized but unissued and otherwise unreserved Common Stock, solely for the purpose of enabling it to issue Underlying Shares as required hereunder, the number of Underlying Shares which are then issuable and deliverable upon the conversion of (and otherwise in respect of) this entire Note (taking into account the adjustments of Section 11), free from preemptive rights or any other contingent purchase rights of persons other than the Investor. The Company covenants that all Underlying Shares so issuable and deliverable shall, upon issuance in accordance with the terms hereof, be duly and validly authorized, issued and fully paid and nonassessable.
 - 11. Certain Adjustments. The Conversion Price is subject to adjustment from time to time as set forth in this Section 11.
 - (a) Stock Dividends and Splits. If the Company, at any time while this Note is outstanding: (i) pays a stock dividend on its Common Stock or otherwise makes a distribution on any class of capital stock that is payable in shares of Common Stock, (ii) subdivides outstanding shares of Common Stock into a larger number of shares, or (iii) combines outstanding shares of Common Stock into a smaller number of shares, then in each such case the Conversion Price shall be multiplied by a fraction of which the numerator shall be the number of shares of Common Stock outstanding immediately before such event and of which the denominator shall be the number of shares of Common Stock outstanding immediately after such event. Any adjustment made pursuant to clause (i) of this paragraph shall become effective immediately after the record date for the determination of shareholders entitled to receive such dividend or distribution, and any adjustment pursuant to clause (ii) or (iii) of this paragraph shall become effective immediately after the effective date of such subdivision or combination.
 - (b) <u>Pro Rata Distributions</u>. If the Company, at any time while this Note is outstanding, distributes to all holders of Common Stock (i) evidences of its indebtedness, (ii) any security (other than a distribution of Common Stock covered by the preceding paragraph), (iii) rights or warrants to subscribe for or purchase any security, or (iv) any other

asset (in each case, "Distributed Property"), then, at the request of the Investor delivered before the 90th day after the record date fixed for determination of shareholders entitled to receive such distribution, the Company will deliver to the Investor, within five Trading Days after such request (or, if later, on the effective date of such distribution), the Distributed Property that the Investor would have been entitled to receive in respect of the Underlying Shares for which this Note could have been converted immediately prior to such record date. If such Distributed Property is not delivered to the Investor pursuant to the preceding sentence, then upon any conversion of this Note that occurs after such record date, the Investor shall be entitled to receive, in addition to the Underlying Shares otherwise issuable upon such conversion, the Distributed Property that the Investor would have been entitled to receive in respect of such number of Underlying Shares had the Investor been the record holder of such Underlying Shares immediately prior to such record date.

Notwithstanding the foregoing, this Section 11(b) shall not apply to any distribution of rights or securities in respect of adoption by the Company of a shareholder rights plan, which events shall be covered by Section 11(a).

(c) Fundamental Transactions. If, at any time while this Note is outstanding, (i) the Company effects any merger or consolidation of the Company with or into another Person, (ii) the Company effects any sale of all or substantially all of its assets in one or a series of related transactions, (iii) any tender offer or exchange offer (whether by the Company or another Person) is completed pursuant to which holders of Common Stock tender or exchange their shares for other securities, cash or property, or (iv) the Company effects any reclassification of the Common Stock or any compulsory share exchange pursuant to which the Common Stock is effectively converted into or exchanged for other securities, cash or property (other than as a result of a subdivision or combination of shares of Common Stock covered by Section 11(a) above) (in any such case, a "Fundamental Transaction"), then upon any subsequent conversion of this Note, the Investor shall have the right to: (x) declare an Event of Default pursuant to clause (iii) thereunder, (y) receive, for each Underlying Share that would have been issuable upon such conversion absent such Fundamental Transaction, the same kind and amount of securities, cash or property as it would have been entitled to receive upon the occurrence of such Fundamental Transaction if it had been, immediately prior to such Fundamental Transaction, the holder of one share of Common Stock (the "Alternate Consideration") or (z) require the surviving entity to issue to the Investor an instrument identical to this Note (with an appropriate adjustments to the conversion price). For purposes of any such conversion, the Company shall apportion the Conversion Price among the Alternate Consideration in a reasonable manner reflecting the relative value of any different components of the Alternate Consideration. If holders of Common Stock are given any choice as to the securities, cash or property to be received in a Fundamental Transaction, then the Investor shall be given the same choice as to the Alternate Consideration it receives upon any conversion of this Note following such Fundamental Transaction. To the extent necessary to effectuate the foregoing provisions, any successor to the Company or surviving entity in such Fundamental Transaction (or, if different, the ultimate parent of such successor or entity or the entity issuing the Alternate Consideration) shall issue to the Investor a new debenture consistent with the foregoing provisions and evidencing the Investor's right to convert such debenture into Alternate Consideration. The terms of any agreement pursuant to which a Fundamental Transaction is effected shall include terms requiring any such successor or surviving entity to comply with the provisions of this paragraph (c) and insuring that this Note (or any such replacement security) will be similarly adjusted upon any subsequent transaction analogous to a Fundamental Transaction.

- (d) <u>Reclassifications</u>; <u>Share Exchanges</u>. In case of any reclassification of the Common Stock, or any compulsory share exchange pursuant to which the Common Stock is converted into other securities, cash or property (other than compulsory share exchanges which constitute Change of Control transactions), the Investors of the Notes then outstanding shall have the right thereafter to convert such shares only into the shares of stock and other securities, cash and property receivable upon or deemed to be held by holders of Common Stock following such reclassification or share exchange, and the Investors shall be entitled upon such event to receive such amount of securities, cash or property as a holder of the number of shares of Common Stock of the Company into which such shares of Notes could have been converted immediately prior to such reclassification or share exchange would have been entitled. This provision shall similarly apply to successive reclassifications or share exchanges.
- (e) <u>Calculations</u>. All calculations under this Section 11 shall be made to the nearest cent or the nearest 1/100th of a share, as applicable. The number of shares of Common Stock outstanding at any given time shall not include shares owned or held by or for the account of the Company, and the disposition of any such shares shall be considered an issue or sale of Common Stock.
- (f) <u>Notice of Adjustments</u>. Upon the occurrence of each adjustment pursuant to this Section 11, the Company at its expense will promptly compute such adjustment in accordance with the terms hereof and prepare a certificate describing in reasonable detail such adjustment and the transactions giving rise thereto, including all facts upon which such adjustment is based. The Company will promptly deliver a copy of each such certificate to the Investor.
- (g) Notice of Corporate Events. If the Company (i) declares a dividend or any other distribution of cash, securities or other property in respect of its Common Stock, including without limitation any granting of rights or warrants to subscribe for or purchase any capital stock of the Company or any Subsidiary, (ii) authorizes and publicly approves, or enters into any agreement contemplating or solicits shareholder approval for any Fundamental Transaction or (iii) publicly authorizes the voluntary dissolution, liquidation or winding up of the affairs of the Company, then the Company shall deliver to the Investor a notice describing the material terms and conditions of such transaction, at least 20 calendar days prior to the applicable record or effective date on which a Person would need to hold Common Stock in order to participate in or vote with respect to such transaction, and the Company will take all steps reasonably necessary in order to insure that the Investor is given the practical opportunity to convert this Note prior to such time so as to participate in or vote with respect to such transaction; provided, however, that the failure to deliver such notice or any defect therein shall not affect the validity of the corporate action required to be described in such notice.
- 12. <u>Fractional Shares</u>. The Company shall not be required to issue or cause to be issued fractional Underlying Shares on conversion of this Note. If any fraction of an Underlying Share would, except for the provisions of this Section, be issuable upon conversion of this Note or payment of interest hereon, the number of Underlying Shares to be issued will be rounded up to the nearest whole share.

13. Prepayment at Option of Company. The Company may prepay this Note, if following the two year anniversary of the Closing Date, (i) the Closing Price of the Common Stock for any 20 Trading Days within a 30 consecutive Trading Day period (each occurring after the two year anniversary of the Closing Date) prior to the date of such Company Prepayment Notice (as defined below) is greater than 175% of the Conversion Price (subject to equitable adjustment as a result of the events set forth in Sections 11(a), (b) and (c)) and (ii) from the beginning of the time period referenced in clause (i) above through the Company Prepayment Date the Underlying Shares are registered for resale under the Registration Statement and the prospectus under such Registration Statement is available for the sale of all Registrable Securities (as defined in the Registration Rights Agreement) held by the Investor, by delivering a written notice (such notice, a "Company Prepayment Notice") to the Investor stating its irrevocable election to prepay all (but not less than all) of the outstanding principal amount of this Note. Subject to the terms and conditions of this Section 13, the Company shall (subject to the following) effect the prepayment of this Note as well as any interest due hereunder pursuant to a Company Prepayment Notice on the 30th day immediately succeeding the date of the Company Prepayment Notice (the "Company Prepayment Date"). Notwithstanding anything to the contrary set forth in this Note, the Investor shall have the right to nullify such Company Prepayment Notice if any of the conditions set forth in this Section 13 shall not have been met. The Company covenants and agrees that it will honor all Conversion Notices tendered from the time of delivery of the Company Prepayment Notice through 6:30 p.m. on the Trading Day prior to the Company Prepayment Date. Notwithstanding the foregoing, the Company and the Investor agree that, if and to the extent Section 5(b) of this Note would restrict the right of the Company to issue or the right of the Investor to receive any of the Underlying Shares otherwise issuable upon conversion in respect of the principal amount of this Note subject to a Company Prepayment Notice, then notwithstanding anything to the contrary set forth in the Company Prepayment Notice, the Company Prepayment Notice shall be deemed automatically amended to apply only to such portion of the principal amount of this Note as would permit conversion in full in compliance with Section 5(b) if such principal amount being prepaid were being converted by the Investor into Underlying Shares. The Investor will promptly (and, in any event, prior to the Company Prepayment Date) notify the Company in writing following receipt of a Company Prepayment Notice if Section 5(b) would restrict its right to receive the full number of otherwise issuable Underlying Shares in respect of the principal amount of this Note being prepaid following such Company Prepayment Notice.

14. Notices. Any and all notices or other communications or deliveries hereunder (including, without limitation, any Conversion Notice or Company Prepayment Notice) shall be in writing and shall be deemed given and effective on the earliest of (i) the date of transmission, if such notice or communication is delivered via facsimile at the facsimile number specified in this Section prior to 6:30 p.m. (New York City time) on a Trading Day, (ii) the next Trading Day after the date of transmission, if such notice or communication is delivered via facsimile at the facsimile number specified in this Section on a day that is not a Trading Day or later than 6:30 p.m. (New York City time) on any Trading Day, (iii) the Trading Day following the date of mailing, if sent by nationally recognized overnight courier service, or (iv) upon actual receipt by the party to whom such notice is required to be given. The addresses for such communications shall be: (i) if to the Company, to 10701 E. Ute Street, Tulsa, Oklahoma 74116, facsimile: (918) 838-8810, attention Chief Financial Officer, (ii) if to the Investor, to the address or facsimile number appearing on the Company's shareholder records or such other

address or facsimile number as the Investor may provide to the Company in accordance with this Section.

15. Miscellaneous.

- (a) This Note shall be binding on and inure to the benefit of the parties hereto and their respective successors and assigns.
- (b) Subject to Section 15(a), above, nothing in this Note shall be construed to give to any person or corporation other than the Company and the Investor any legal or equitable right, remedy or cause under this Note. This Note shall inure to the sole and exclusive benefit of the Company and the Investor.
- (c) All questions concerning the construction, validity, enforcement and interpretation of this Note shall be governed by and construed and enforced in accordance with the laws of the State of Delaware. Each party agrees that all Proceedings shall be commenced exclusively in the state and federal courts sitting in the State of Delaware (the "Delaware Courts"). Each party hereto hereby irrevocably submits to the exclusive jurisdiction of the Delaware Courts for any Proceeding, and hereby irrevocably waives, and agrees not to assert in any Proceeding, any claim that it is not personally subject to the jurisdiction of any Delaware Court or that a Delaware Court is an inconvenient forum for such Proceeding. Each party hereto hereby irrevocably waives personal service of process and consents to process being served in any such Proceeding by mailing a copy thereof via registered or certified mail or overnight delivery (with evidence of delivery) to such party at the address in effect for notices to it under this Note and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law. Each party hereto hereby irrevocably waives, to the fullest extent permitted by applicable law, any and all right to trial by jury in any legal Proceeding. The prevailing party in a Proceeding shall be reimbursed by the other party for its reasonable attorneys' fees and other costs and expenses incurred with the investigation, preparation and prosecution of such Proceeding.
- (d) The headings herein are for convenience only, do not constitute a part of this Note and shall not be deemed to limit or affect any of the provisions hereof.
- (e) In case any one or more of the provisions of this Note shall be invalid or unenforceable in any respect, the validity and enforceability of the remaining terms and provisions of this Note shall not in any way be affected or impaired thereby and the parties will attempt in good faith to agree upon a valid and enforceable provision which shall be a commercially reasonable substitute therefor, and upon so agreeing, shall incorporate such substitute provision in this Note.
- (f) No provision of this Note may be waived or amended except (i) in accordance with the requirements set forth in the Purchase Agreement, and (ii) in a written instrument signed, in the case of an amendment, by the Company and the Investor or, in the case of a waiver, by the party against whom enforcement of any such waiver is sought. No waiver of any default with respect to any provision, condition or requirement of this Note shall be deemed

to be a continuing waiver in the future or a waiver of any subsequent default or a waiver of any other provision, condition or requirement hereof, nor shall any delay or omission of either party to exercise any right hereunder in any manner impair the exercise of any such right.

(g) To the extent it may lawfully do so, the Company hereby agrees not to insist upon or plead or in any manner whatsoever claim, and will resist any and all efforts to be compelled to take the benefit or advantage of, usury laws wherever enacted, now or at any time hereafter in force, in connection with any claim, action or Proceeding that may be brought by any Investor in order to enforce any right or remedy under the Notes. Notwithstanding any provision to the contrary contained in the Notes, it is expressly agreed and provided that the total liability of the Company under the Notes for payments in the nature of interest shall not exceed the maximum lawful rate authorized under applicable law (the "Maximum Rate"), and, without limiting the foregoing, in no event shall any rate of interest or default interest, or both of them, when aggregated with any other sums in the nature of interest that the Company may be obligated to pay under the Notes exceed such Maximum Rate. It is agreed that if the maximum contract rate of interest allowed by law and applicable to the Notes is increased or decreased by statute or any official governmental action subsequent to the date hereof, the new maximum contract rate of interest allowed by law will be the Maximum Rate of interest applicable to the Notes from the effective date forward, unless such application is precluded by applicable law. If under any circumstances whatsoever, interest in excess of the Maximum Rate is paid by the Company to any Investor with respect to indebtedness evidenced by the Notes, such excess shall be applied by such Investor to the unpaid principal balance of any such indebtedness or be refunded to the Company, the manner of handling such excess to be at such Investor's election.

IN WITNESS WHEREOF, the Company has caused this Note to be duly executed by a duly authorized officer as of the date first above indicated.

By: Name:

MATRIX SERVICE COMPANY

Title:

EXHIBIT A

CONVERSION NOTICE

(To be Executed by the Registered Investor in order to convert Notes)

The undersigned hereby elects to convert the principal amount of Note indicated below, into shares of Common Stock of Matrix Service Company, as of the date written below. If shares are to be issued in the name of a Person other than undersigned, the undersigned will pay all transfer taxes payable with respect thereto and is delivering herewith such certificates and opinions as reasonably requested by the Company in accordance therewith. No fee will be charged to the Investor for any conversion, except for such transfer taxes, if any. All terms used in this notice shall have the meanings set forth in the Note.

Conversion calculations:	
	Date to Effect Conversion
	Principal amount of Note owned prior to conversion
	Principal amount of Note to be Converted
	Principal amount of Note remaining after Conversion
	DTC Account
	Number of shares of Common Stock to be Issued
	Applicable Conversion Price
	Name of Investor
	By:
	Name:
	Title:

By the delivery of this Conversion Notice the Investor represents and warrants to the Company that its ownership of the Common Stock does not exceed the restrictions set forth in Section 5(b) of the Note.

Annex 1

Matrix Service Company Senior Unsecured Convertible Notes due April 25, 2010

CONVERSION SCHEDULE

This Conversion Schedule reflects conversions made under the above referenced Notes.

Dated:

Date of Conversion Amount of Conversion Remaining Subsequent to Conversion Price

SUBORDINATION AGREEMENT

THIS SUBORDINATION AGREEMENT dated as of April 22, 2005, made by the subordinated creditors identified on the signature pages hereto (together with any successor thereto, each a "<u>Subordinated Creditor</u>" and collectively the "<u>Subordinated Creditors</u>") and the various obligors listed on the signature pages of this Agreement, including **MATRIX SERVICE COMPANY**, a Delaware corporation ("<u>Borrower</u>") (each an "<u>Obligor</u>" and collectively the "<u>Obligors</u>"), in favor of **JPMORGAN CHASE BANK, N.A.** (successor by merger to Bank One, N.A. (Main Office Chicago)), as a Lender, LC Issuer and as Agent (the "<u>Agent</u>") for the Lenders under the Credit Agreement defined below, and in favor of the Lenders now or hereafter party to the Credit Agreement (whether one or more, the "<u>Lenders</u>").

WITNESSETH:

WHEREAS, Borrower intends to borrow \$30,000,000.00 from the Subordinated Creditors, with the indebtedness created thereby being or to be evidenced by those certain Senior Unsecured Convertible Notes due 2010 to be issued by Borrower in the aggregate principal amount of \$30,000,000.00 (the "Subordinated Notes"); and

WHEREAS, the Lenders have extended credit to Borrower pursuant to the Credit Agreement and the other Loan Documents described therein (the "Loans"), and the remainder of the Obligors have guaranteed repayment of the Loans and provided security for the Loans, all as described in the Credit Agreement and the other Loan Documents; and

WHEREAS, under the terms of the Credit Agreement, Borrower is prohibited from executing the Subordinated Notes and incurring the debt evidenced thereby, without the consent of the Agent and Lenders; and

WHEREAS, Agent and Lenders have conditioned their consent to the extension of credit by the Subordinated Creditors to Borrower upon, among other things, the execution and delivery to the Agent of an agreement subordinating all indebtedness of the Obligors from time to time existing in favor of the Subordinated Creditors under the Subordinated Documents (as defined herein) to the prior Payment in Full (as defined herein) of all of the obligations of the Obligors from time to time existing in favor of the Lenders and the Agent under the Senior Documents (as defined herein) and to all security interests and other liens securing such obligations.

NOW, THEREFORE, in consideration of the premises and in order to induce the Agent and the Lenders to consent to the extension of credit by the Subordinated Creditors to Borrower, the Subordinated Creditors and the Obligors hereby agree with the Agent and the Lenders as follows:

SECTION 1. **<u>Definitions</u>**. As used in this Agreement, the following terms shall have the respective meanings indicated below, such meanings to be applicable equally to both the singular and plural forms of such terms:

"**Agent**" has the meaning specified for such term in the introductory paragraph herein.

- "Borrower" has the meaning specified for such term in the introductory paragraph herein.
- "Collateral" means any and all "Collateral" as defined in the Senior Documents.
- "Credit Agreement" means the Original Credit Agreement, as amended by that certain Amendment One to Credit Agreement dated as of May 22, 2003, that certain Amendment Two to Credit Agreement dated as of August 27, 2003, that certain Amendment Three to Credit Agreement dated as of December 19, 2003, that certain Amendment Four to Credit Agreement dated as of March 11, 2004, that certain Amendment Five to Credit Agreement dated as of May 6, 2004, that certain Amendment Six to Credit Agreement dated as of August 5, 2004, that certain Amendment Seven to Credit Agreement dated as of October 6, 2004, that certain Amendment Eight to Credit Agreement dated as of November 30, 2004, that certain letter agreement dated March 23, 2005 and that certain letter agreement dated April 8, 2005, and that certain Amendment Nine to Credit Agreement dated as of April 22, 2005, and as may hereafter be amended from time to time
- "Equity Interest" means capital stock and all warrants, options or other rights to acquire capital stock (but excluding any debt security that is convertible into, or exchangeable for, capital stock).
 - "Holders" means the Subordinated Creditors and/or any subsequent holders of the Subordinated Notes.
- "<u>Interest Prepayment</u>" means the amount of \$4,200,000.00 Borrower has agreed to pay to the Subordinated Creditors at the closing of the transaction involving the Subordinated Notes in payment of interest payable under the Subordinated Notes through the second anniversary of the Closing Date (as defined in the Subordinated Documents).
 - "Lenders" has the meaning specified for such term in the introductory paragraph herein.
- "<u>Liens</u>" means any security interest, mortgage, pledge, hypothecation, assignment, encumbrance, deposit arrangement, lien (statutory or otherwise), charge against or interest in property to secure payment of a debt or performance of an obligation.
 - "Loans" has the meaning specified for such term in the Recitals above.
 - "Notes" means any and all promissory notes payable by Borrower in favor of any of the Lenders issued pursuant to the Credit Agreement.
 - "Obligors" has the meaning specified for such term in the introductory paragraph herein.
- "Original Credit Agreement" means that certain Credit Agreement dated as of March 7, 2003, among Borrower, Agent and the various Lenders party thereto.
 - "Payment Blockage Notice" has the meaning specified in Section 4(b) below.

"Payment in Full" means the indefeasible, full and final payment of all the Senior Obligations, whether or not any of the Senior Obligations shall have been voided, disallowed or subordinated pursuant to any provision of the United States Bankruptcy Code or any applicable state fraudulent conveyance laws, in any case whether asserted directly or indirectly, and the termination, expiration or reduction to zero of all the "Commitments" under the Credit Agreement, and the expression "Paid in Full" shall have a correlative meaning. Notwithstanding the foregoing, for purposes of this definition, the Senior Obligations shall be deemed Paid in Full if the Senior Obligations are fully cash collateralized (i.e., the Agent or Lenders hold, or have a perfected first priority security interest in, cash and cash equivalents in amount sufficient to pay the Senior Obligations).

"Person" means an individual or corporation, partnership, trust, incorporated or unincorporated association, joint venture, limited liability company, joint stock company, government (or an agency or subdivision thereof) or other entity of any kind.

"<u>Reorganization</u>" means any voluntary or involuntary dissolution, winding-up, total or partial liquidation or reorganization, or bankruptcy, insolvency, receivership or other statutory or common law proceeding or arrangement involving the liquidation of the Borrower or the Borrower's assets, or any assignment for the benefit of creditors or any marshalling of the assets or liabilities of the Borrower.

"<u>Senior Documents</u>" means the Credit Agreement and the other Loan Documents, as defined in the Credit Agreement, each as may be amended, restated, supplemented, modified, extended, renewed, refinanced or replaced from time to time, and any replacement thereof in connection with any refinancing of the Senior Obligations.

"<u>Senior Nonpayment Default</u>" means the occurrence or existence of any event, circumstance, condition or state of facts that, by the terms of the Senior Documents, permits the Agent or the Lenders to declare the Senior Obligations due and payable prior to maturity, other than a Senior Payment Default.

"Senior Obligations" means all indebtedness, obligations and other liabilities of the Obligors, or any of them, now or hereafter existing in favor of the Lenders and the Agent under the Senior Documents, whether for principal (including, without limitation, protective advances), reimbursement obligations, interest (including interest accruing subsequent to the filing of any petition initiating any Reorganization, whether or not a claim for such interest is allowed in any such proceeding), guaranteed obligations, fees, premiums, indemnities, costs, expenses (including, without limitation, auditor, legal and other professional fees, costs and expenses), or otherwise, provided that the principal amount of extensions of credit under the Senior Documents that shall constitute Senior Obligations for purposes of this Agreement shall under no circumstances exceed \$90 million.

"Senior Payment Default" means a default under the Senior Obligations arising from the failure of Borrower to make a payment of principal, interest or any other payment required in respect of the Senior Obligations (i) that, by the terms of the Senior Documents, permits the Agent or the Lenders to declare the Senior Obligations then due and payable prior to maturity or

- (ii) that has resulted from acceleration, whether automatic or upon notice, of the Senior Obligations.
 - "Subordinated Creditors" has the meaning specified for such term in the introductory paragraph herein.
- "<u>Subordinated Documents</u>" means the Subordinated Notes and any other instrument or document described in <u>Schedule I</u> hereto, and all other documents executed concurrently or in conjunction therewith, as the same may be amended or modified from time to time (in accordance with this Agreement).
 - "Subordinated Notes" has the meaning assigned to such term in the Recitals above.
- "Subordinated Obligations" means all indebtedness, obligations and other liabilities of the Obligors, or any of them, incurred pursuant to the Subordinated Documents, now or hereafter existing in favor of the Subordinated Creditors, including without limitation, all interest thereon and all fees, premiums, costs, expenses, indemnities and other amounts payable in respect thereof, and all dividends, redemptions, put and call rights and other contractual obligations of Borrower to Subordinated Creditors, whether now existing or hereafter arising.
 - "Uniform Commercial Code" shall mean the Uniform Commercial Code adopted by the State of New York, as amended from time to time.
- SECTION 2. Agreement to Subordinate. The Subordinated Creditors and the Obligors agree that (i) the Subordinated Obligations are and shall be subordinated, to the extent and in the manner hereinafter set forth, in right of payment to the prior Payment in Full of the Senior Obligations and (ii) any security interests or other Liens on all or any part of any and all collateral securing payment of any Subordinated Obligation or guaranty thereof, whether now existing or hereafter granted or arising, are and shall be subordinated, to the fullest extent permitted by law and as hereinafter set forth, to any and all Liens in favor of the Agent securing the Senior Obligations, notwithstanding the perfection, order of perfection or failure to perfect any such Lien in favor of the Agent, or the filing or recording, order of filing or recording or failure to file or record this Agreement or any instrument or other document in any filing or recording office in any jurisdiction.

SECTION 3. Reorganization.

- (a) In the event of any Reorganization relating to the Borrower prior to the Payment in Full of the Senior Obligations:
- (i) the holders of the Senior Obligations shall be Paid in Full all amounts due or to become due on or in respect of all Senior Obligations before any payment or distribution in cash, securities or other property that otherwise would be payable or deliverable upon or with respect to any of the Subordinated Obligations, and to that end the holders of the Senior Obligations shall be entitled to receive, for application to the payment thereof, any payment or distribution of any kind or character, whether in cash, securities or other

property which may be payable or deliverable in respect of the Subordinated Obligations in any Reorganization; and

- (ii) the Subordinated Creditors will duly and promptly take such action as the Agent may reasonably request (A) to file appropriate claims or proofs of claim with respect thereto, (B) to enforce any and all claims with respect to, and any security interests and other liens securing payment of, the Subordinated Obligations, and (C) to collect and receive any and all payments or distributions which may be payable or deliverable upon or with respect to the Subordinated Obligations.
- (b) For purposes of this Agreement, "cash, securities or other property" shall not be deemed to include securities of the Borrower as reorganized, or securities of the Borrower or any other Person provided for by a Reorganization plan authorized by an order or decree of a court of competent jurisdiction that are subordinated in right of payment to all Senior Obligations and any debt securities issued in exchange for Senior Obligations in connection with the Reorganization to substantially the same extent as, or to a greater extent than, the Subordinated Obligations are subordinated to the Senior Obligations hereunder.

SECTION 4. Restrictions on Payments and Actions.

- (a) In the event that any Senior Payment Default (including, without limitation, any Senior Payment Default arising from acceleration, whether automatic or upon notice, of the Senior Obligations following the occurrence of a Senior Nonpayment Default) shall have occurred, then no payment in respect of the Subordinated Obligations shall be made by the Borrower or any other Obligor, and no such payment shall be accepted by the Holders, until such time as the Senior Payment Default is cured or waived.
- (b) In the event that any Senior Nonpayment Default shall have occurred, then upon the receipt by the Borrower and the Holders of written notice of such Senior Nonpayment Default (a "Payment Blockage Notice") from the Agent or the Lenders, no payment in respect of the Subordinated Obligations shall be made by the Borrower or any other Obligor, and no such payment shall be accepted by the Holders, prior to the earlier of (i) the date on which such default is cured or waived and (ii) 179 days after the date on which the applicable Payment Blockage Notice was received by the Holders, unless the maturity of the Senior Obligations has been accelerated (in which case such acceleration shall automatically result in a Senior Payment Default). If the Holders receive any such Payment Blockage Notice, no subsequent Payment Blockage Notice will be effective for purposes of this Section 4(b) unless and until (A) at least 360 days have elapsed since the commencement of the immediately prior Payment Blockage Notice and (B) all scheduled payments of principal, premium and special interest, if any, on the Subordinated Notes that have come due have been Paid in Full. No Senior Nonpayment Default that existed or was continuing on the date of delivery of any Payment Blockage Notice to the Holders may be, or may be made, the basis for a subsequent Payment Blockage Notice unless such default has been waived for a period of not less than 180 days.
- (c) Notwithstanding the terms of Section 4(a) and Section 4(b) above, Holders of Subordinated Notes may receive and retain Equity Interests as part of a conversion of the

Subordinated Notes to Equity Interests as contemplated by the explicit terms of the Subordinated Notes.

- (d) The provisions of Section 4(a) and Section 4(b) above and of Section 5(a) below shall have no applicability to the Interest Prepayment, which is nonrefundable except in the limited circumstances set forth in Section 5(b)(ii) of the Subordinated Notes. In the event a Holder elects, pursuant to Section 5(b)(ii) of the Subordinated Notes, to refund the applicable portion of the Interest Prepayment to the Borrower in connection with a conversion of Subordinated Notes into Equity Interests, such refund shall be paid over by the Holder to the Agent, or failing that, by the Company to the Agent, for application to the repayment of the Senior Obligations.
- (e) If a default shall occur and be continuing under the Subordinated Documents and a Senior Payment Default or Senior Nonpayment Default has occurred and is continuing, the Holders shall not ask, demand or sue for any payment or distribution or seek any other remedy (other than as provided in Section 4(f) below) in respect of the Subordinated Obligations or commence or join with any other creditor (other than the Agent or the Lenders) in commencing any Reorganization proceeding in respect of the Borrower prior to the earliest of (i) such time as the Senior Payment Default or Senior Nonpayment Default is cured or waived, (ii) such time as the maturity of the Senior Obligations is accelerated, or (iii) the 179th day after occurrence of such Senior Payment Default or the date of receipt of the Payment Blockage Notice in respect of such Senior Nonpayment Default.
- (f) Nothing in this Section 4 shall limit the rights of Holders to (i) file proofs of claim and all necessary responsive or defensive pleadings in opposition to any motion, claim, adversary proceeding or other pleading made by any Person objecting to or otherwise seeking the disallowance of the claims of the Holders, (ii) vote on any Reorganization plan and appear and be heard on any matter in connection therewith, and (iii) file all pleadings, objections, motions or agreements which assert rights or interests available to unsecured creditors of the Borrower arising under applicable bankruptcy or non-bankruptcy laws, provided that any of the foregoing actions may not be otherwise inconsistent with the terms of this Agreement.
- (g) The failure to make a payment in respect of the Subordinated Obligations by reason of the provisions of this Section 4 or any other provision of this Agreement shall not be construed as preventing the occurrence of a default or event of default under the Subordinated Obligations. Except as expressly set forth in Section 4(e) above, nothing in this Agreement shall have any effect on the right of the Holders to accelerate the maturity of the Subordinated Obligations in accordance with their terms or to make a claim for payment thereunder.

SECTION 5. Additional Provisions Concerning Subordination.

(a) All payments or distributions upon or with respect to the Subordinated Obligations which are received by any Subordinated Creditor contrary to the provisions of this Agreement shall be received in trust for the benefit of the Agent, shall be segregated from other funds and property held by such Subordinated Creditor and shall be forthwith paid over to the Agent in the same form as so received (with any necessary endorsement) to be applied (in the case of cash) to or held as collateral (in the case of securities or other non-cash property) to the

extent necessary for the payment or prepayment of the Senior Obligations until the Senior Obligations shall have been Paid in Full.

- (b) The Agent is hereby authorized to demand specific performance of this Agreement at any time when any Subordinated Creditor or any Obligor shall have failed to comply with any of the provisions of this Agreement applicable to such Subordinated Creditor or such Obligor, as the case may be, whether or not the particular Subordinated Creditor or the particular Obligor against whom such relief is sought shall have complied with any of the provisions hereof, and the Subordinated Creditors and the Obligors hereby irrevocably waive any defense based on the adequacy of a remedy at law which might be asserted as a bar to such remedy of specific performance.
- (c) Borrower shall promptly notify Agent of (i) any facts known to Borrower that would cause a payment with respect to the Subordinated Obligations to violate this Agreement or (ii) any default under any Subordinated Document, but failure to give such notice shall not affect the terms of this Agreement. The Subordinated Creditors undertake to notify the Agent of the occurrence of any event of default under the Subordinated Obligations, but the Subordinated Creditors shall not incur any liability hereunder for failure to provide such notification.
- (d) Whenever a distribution is to be made or a notice given to any holders of Senior Obligations with respect to the Senior Obligations, the distribution may be made and the notice given to the Agent.
- (e) Agent and Lenders may do any of the following from time to time without the consent of or notice to the Subordinated Creditors, without incurring responsibility to Subordinated Creditor and without impairing or releasing any of Agent's or any Lender's rights or any of the obligations of the Subordinated Creditors hereunder: (i) change any of the Senior Obligations, including but not limited to the amount, manner, place or terms of payment, the time of payment, or any renewal or other alteration of the Senior Obligations in any respect; (ii) sell, exchange, release or otherwise deal with all or any part of the Collateral; (iii) exercise or refrain from exercising any rights against Borrower or others (including without limitation, the Subordinated Creditors); and (iv) apply any sums received, by whomsoever paid or howsoever realized, to the Senior Obligations. Neither Agent nor any of the Lenders shall at any time be required to institute suit or exercise or exhaust remedies against any Person obligated to pay any of the Senior Obligations prior to exercising its rights under or receiving the benefits of this Agreement.
- SECTION 6. <u>Negative Covenants of the Subordinated Creditors</u>. So long as Payment in Full has not occurred, the Subordinated Creditors will not, without the prior written consent of the Agent:
 - (a) (i) obtain or otherwise receive the benefit of any Lien on any property to secure any of the Subordinated Obligations; (ii) obtain or otherwise receive the benefit of any guaranty with respect to any of the Subordinated Obligations; or (iii) subordinate any security interest or other Lien securing the Subordinated Obligations to any security interest or other Lien in favor of any Person other than the Agent; provided that the foregoing provisions of this

Section 6(a) shall not apply to the issuance of Equity Interests pursuant to the Subordinated Documents;

- (b) sell, assign, pledge, encumber or otherwise dispose of any Subordinated Obligation or any collateral securing any Subordinated Obligation, without first obtaining a subordination agreement from the proposed transferee of such Subordinated Obligation in form and substance reasonably satisfactory to the Agent (provided that Agent shall be deemed to be acting reasonably in requesting any term of any such subordination agreement if such term, or any substantially similar term, is included in this Agreement); or
- (c) permit any Subordinated Document or the terms of any Subordinated Obligation to be amended or otherwise modified in any respect, except in such a manner as shall not (i) grant collateral to the Subordinated Creditors or (ii) materially increase Borrower's obligations under the Subordinated Documents, and in the case of clause (i) or (ii), which shall not adversely affect the rights of the Agent or any of the Lenders.

SECTION 7. Collateral.

- (a) The parties acknowledge that the Subordinated Obligations are unsecured obligations of the Borrower. None of the provisions of this <u>Section 7</u> or otherwise contained in this Agreement that contemplate or address Liens or security interests in favor of the Subordinated Creditors in order to secure payment or performance of the Subordinated Obligations are intended to grant or imply the grant of any consent to any such Lien or security interest.
- (b) The Subordinated Creditors hereby confirm that regardless of the relative times of attachment or perfection thereof, and regardless of anything in any Subordinated Document or any Senior Document to the contrary, any security interests or Liens granted from time to time to the Agent in the Collateral as security for the Senior Obligations shall in all respects be first and senior security interests and Liens, superior to any security interests or Liens at any time granted to the Subordinated Creditors in the Collateral as security for the Subordinated Obligations. The priorities specified herein are applicable irrespective of the time or order of attachment or perfection of security interests or the time or order of filing of financing statements. The Subordinated Creditors agree not to seek to challenge, to avoid, to subordinate or to contest or directly or indirectly to support any other Person in challenging, avoiding, subordinating or contesting in any judicial or other proceeding, including, without limitation, any Reorganization involving the Obligors or other Person, the priority, validity, extent, perfection or enforceability of any Lien held by the Agent or the Lender in all or any part of the Collateral.
- (c) In foreclosing or realizing on the Liens in favor of the Agent, for the benefit of the Lenders, on any of the Collateral, the Agent may proceed in any manner consistent with the provisions of the Uniform Commercial Code or other applicable law (either individually or collectively with any other remedies available to it) which the Agent, in its sole and absolute discretion, shall choose, even though a higher price might have been realized if the Agent had proceeded to foreclose or realize on its security interests and/or Liens in another manner. The Agent hereby reserves the right, in its sole and absolute discretion (and without in any way

diminishing or altering its rights hereunder or the subordination provisions contained herein), to modify, amend, waive or release any of the terms of the Credit Agreement and other Senior Documents and any other agreement, instrument and other document relating to any of the foregoing or the Senior Obligations and to exercise or refrain from exercising any powers or rights which it may have thereunder. Nothing herein shall be deemed to obligate the Agent to foreclose or realize or take any action whatsoever with respect to any of the Collateral, and the Subordinated Creditors hereby irrevocably waive any right they may have, now or in the future, to require Agent or any of the Lenders to foreclose or realize or take any action whatsoever with respect to any of the Collateral.

(d) The Subordinated Creditors agrees for the benefit of the Agent and the Lenders that until the Senior Obligations are Paid in Full, the Subordinated Creditors will not take, and will not consent to the Obligors taking, any action to sell, transfer, foreclose or realize on, levy against or execute upon, or exercise any rights or remedies whatsoever in respect of, any Lien on the Collateral granted to secure the Subordinated Obligations without the prior written consent of the Agent in each instance. Until the Payment in Full of the Senior Obligations, the Agent shall retain the sole right to exercise the rights of a "secured party" in respect of the Collateral, whether under law, including, without limitation, the Uniform Commercial Code, the Credit Agreement, any other Senior Document or any other agreement, instrument or other document relating to the Collateral or any of the foregoing or otherwise. The Subordinated Creditors hereby agrees that any Collateral and all payments or distributions upon or with respect to any Collateral (including proceeds thereof) that are received by the Subordinated Creditors by reason of any Lien on all or any part of the Collateral shall be received in trust for the benefit of the Agent, shall be segregated from other funds and property held by the Subordinated Creditors and shall be forthwith turned over to the Agent in the same form received (with any necessary endorsement) to be applied (in the case of cash) to or held as Collateral (in the case of securities or other non-cash property) for the payment or prepayment of the Senior Obligations until the Senior Obligations shall have been Paid in Full.

(e) In the event of any Reorganization of any of the Obligors, unless the Agent shall otherwise consent in writing, and until the Payment in Full of the Senior Obligations, (i) if the Agent shall consent to any use, sale or lease of any Collateral (pursuant to Section 363(f) of the Bankruptcy Code or otherwise), the Subordinated Creditors (A) shall not object to such use, sale or lease of such Collateral, (B) shall be deemed to have consented to such use, sale or lease of such Collateral, and (C) shall (promptly after the written request of the Agent) consent in writing to such use, sale or lease of such Collateral, (ii) if the Subordinated Creditors have a Lien on any of the Collateral, the Subordinated Creditors shall not seek relief from the automatic stay imposed pursuant to Section 362 of the Bankruptcy Code in order to exercise any rights or remedies in respect of any of the Collateral, (iii) the Subordinated Creditors shall not object to or otherwise file any motion seeking to prevent the Agent from obtaining relief from the automatic stay imposed pursuant to Section 362 of the Bankruptcy Code in order to exercise any rights or remedies in respect of any of the Collateral, and (iv) if the Subordinated Creditors have a Lien on any of the Collateral, the Subordinated Creditors shall not assert any right to require adequate protection of its interest in any Collateral or any other adequate protection to the extent that (A) any Obligor, as a debtor under the Bankruptcy Code, seeks to subject any of

the Collateral to a security interest to secure any indebtedness or (B) Agent or any of the Lenders propose to provide debtor in possession financing to Borrower.

- (f) In the event the Agent consents to any sale, transfer or other disposition of any of the Collateral, or sells, transfers or otherwise disposes of any of the Collateral in connection with the enforcement of its rights as a secured party and/or as mortgagee (whether prior to, during or after a Reorganization), such sale, transfer or other disposition shall be effective to transfer such Collateral free of any Lien of the Subordinated Creditors in such Collateral and the Subordinated Creditor having a Lien on all or any part of the Collateral shall take such action and execute such documents as shall be necessary or as the Agent shall reasonably request to effect the foregoing. The sole right of the Subordinated Creditors, if they have a Lien on all or any part of the Collateral so sold, transferred or otherwise disposed of by the Agent, shall be to receive the proceeds thereof after the Payment in Full of all Senior Obligations.
- (g) The Subordinated Creditors acknowledge and agrees with the Agent that the Subordinated Creditors shall have no right whatsoever to require the Agent to consult with, or solicit the approval of, the Subordinated Creditors in connection with any sale, transfer or other disposition of any of the Collateral, including, without limitation, any such transaction during a Reorganization; and the Subordinated Creditors agrees to take no action to challenge, restrain or prevent the consummation of any such transaction. However, nothing herein shall absolve the Obligors of any obligation to notify the Subordinated Creditors of any such disposition of Collateral.

SECTION 8. Legend; Further Assurances.

- (a) The Subordinated Creditors and the Obligors will cause the Subordinated Notes to be indorsed with the following legend:
- "The indebtedness evidenced or secured by this instrument is subordinated to the prior Payment in Full of the Senior Obligations (as defined in the Subordination Agreement described below) pursuant to, and to the extent provided in, the Subordination Agreement dated as of April 22, 2005, (the "Subordination Agreement") made by the Subordinated Creditors and Obligors referred to therein in favor of JPMORGAN CHASE BANK, N.A. (successor by merger to Bank One, N.A. (Main Office Chicago)), as a Lender, LC Issuer and as Agent for the Lenders under the Credit Agreement dated as of March 7, 2003, as it has been amended, and in favor of the Lenders party to the Credit Agreement.
- (b) The Subordinated Creditors and the Obligors will (i) mark their books of account in such a manner as shall be effective to give proper notice of the effect of this Agreement and (ii) in the case of any Subordinated Obligation which is not evidenced by any instrument, upon the Agent's request, cause such Subordinated Obligation to be evidenced by an

appropriate instrument or instruments indorsed with the above legend. The Subordinated Creditors and the Obligors will, at their expense and at any time and from time to time, promptly execute and deliver all further instruments and other documents, and take all further action, that may be necessary or, in the reasonable opinion of the Agent, desirable, or that the Agent may reasonably request, in order to protect any right or interest granted or purported to be granted hereby or to enable the Agent to exercise and enforce its rights and remedies hereunder.

SECTION 9. Obligations Unconditional.

- (a) All rights and interests of the Agent hereunder, and all agreements and obligations of the Subordinated Creditors and the Obligors hereunder, shall remain in full force and effect irrespective of: (i) any lack of validity or enforceability of any Senior Document or any other agreement or instrument relating thereto; (ii) any change in the time, manner or place of payment of, or in any other term in respect of, all or any of the Senior Obligations, or any other amendment or waiver of or any consent to departure from any Senior Document; (iii) any exchange or release of, or non-perfection of any Lien on or security interest in, any Collateral, or any release or amendment or waiver of or consent to departure from any guaranty, for all or any of the Senior Obligations; (iv) all other notices, demands and protests, and all other formalities of every kind in connection with the enforcement of the Senior Obligations or of the obligations of the Subordinated Creditors and the Obligors hereunder, the omission of or delay in which, but for the provisions of this Section 9, might constitute grounds for relieving the Subordinated Creditors or the Obligors of its obligations hereunder; (v) any requirement that the Agent protect, secure, perfect or insure any security interest or other Lien or any property subject thereto or exhaust any right to take any action against any Obligor or any other Person or any Collateral; or (vi) any other circumstance which might otherwise constitute a defense available to, or a discharge of, any of the Obligors in respect of any of the Senior Obligations or a Subordinated Creditor or any of the Obligors in respect of this Agreement.
- (b) This Agreement shall automatically terminate without further action by any party upon Payment in Full of all of the Senior Obligations; provided that it shall continue to be effective or shall be reinstated, as the case may be, if at any time any payment of any of the Senior Obligations is rescinded or must otherwise be returned by the Lender or the Agent upon the insolvency, bankruptcy or reorganization of Obligors or otherwise, all as though such payment had not been made.
- (c) No right of the Agent or any of the Lenders may be impaired by any act or failure to act by Borrower or any Holder of any Subordinated Obligations or by any failure of Borrower or any Subordinated Creditor to comply with any provision of this Agreement.

SECTION 10. <u>Subrogation</u>. After all Senior Obligations have been Paid In Full and until the Subordinated Obligations are paid in full, Holders will be subrogated (equally and ratably with all other indebtedness pari passu with the Subordinated Notes) to the rights of holders of the Senior Obligations to receive distributions applicable to the Senior Obligations to the extent that distributions otherwise payable to the Holders have been applied to the payment of Senior Obligations. A distribution made under this Agreement to holders of Senior Obligations that otherwise would have been made to Holders is not, as between the Borrower

and Holders, a payment by the Borrower on the Subordinated Notes. No payment or distribution to the Agent pursuant to the provisions of this Agreement shall entitle any Subordinated Creditor to exercise any rights of subrogation in respect thereof until 91 days after the Senior Obligations shall have been Paid in Full; provided, however, that if any payment to the Agent or any Lender is avoided or otherwise rescinded in or as a result of a Reorganization or otherwise, the claim of the Agent and the Lenders resulting therefrom shall be considered to be Senior Obligations for all purposes of this Agreement and the rights of subrogation of the Subordinated Creditors as provided herein shall likewise be rescinded until Payment in Full of the Senior Obligations.

SECTION 11. Representations and Warranties.

- (a) The Obligors hereby represents and warrants as follows:
- (i) Each Subordinated Document constitutes the legal, valid and binding obligation of the Borrower, enforceable against the Borrower in accordance with its terms, except as such enforceability may be limited by general principles of equity and subject to applicable bankruptcy, insolvency, reorganization, moratorium, liquidation or similar laws relating to, or affecting generally, the enforcement of applicable creditors rights and remedies.
- (ii) The execution, delivery and performance of this Agreement by each Obligor have been duly authorized by all necessary corporate action, and do not and will not contravene any of its organizational documents or any contractual restriction binding on or affecting such Obligor.
- (iii) This Agreement constitutes the legal, valid and binding obligation of each Obligor Creditor, enforceable against such Obligor in accordance with its terms.
 - (iv) None of the Subordinated Documents has been amended or otherwise modified, and there exists no default in respect of any thereof.
- (v) <u>Schedule I</u> hereto is a complete and correct description of all instruments, agreements and other writings evidencing, governing, guaranteeing or securing any Subordinated Obligations, and true copies thereof have been delivered to the Agent.
- (b) Each Subordinated Creditor hereby represents and warrants as follows:
 - (i) The aggregate principal amount of the Subordinated Obligations outstanding as of the date hereof is \$30,000,000.00.
- (ii) Such Subordinated Creditor is an organization validly existing and in good standing under the laws of its jurisdiction of formation, and has all requisite power and authority to execute, deliver and perform this Agreement.
- (iii) The execution, delivery and performance of this Agreement by such Subordinated Creditor have been duly authorized by all necessary corporate action, and

do not and will not contravene any of its organizational documents or any contractual restriction binding on or affecting such Subordinated Creditor.

- (iv) No authorization or approval or other action by, and no notice to or filing with, any governmental authority or other regulatory body is required for the due execution, delivery and performance by such Subordinated Creditor of this Agreement.
- (v) <u>Schedule I</u> hereto is a complete and correct description of all instruments, agreements and other writings evidencing, governing, guaranteeing or securing any Subordinated Obligations, and true copies thereof have been delivered to the Agent.
- (vi) This Agreement constitutes the legal, valid and binding obligation of such Subordinated Creditor, enforceable against such Subordinated Creditor in accordance with its terms.
- SECTION 12. **Relative Rights**. Nothing in this Agreement will (i) impair, as between Borrower and the Holders of Subordinated Notes, the obligation of Borrower, which is absolute and unconditional, to pay principal of, premium and interest and special interest, if any, on the Subordinated Note, or (ii) affect the relative rights of Holders of Subordinated Notes and creditors of the Borrower other than their rights in relation to holders of Senior Obligations. If Borrower fails, because of this Agreement, to pay principal of, premium or interest or special interest, if any, on a Subordinated Notes on the due date, the failure is still a Default or Event of Default under the Subordinated Notes.
- SECTION 13. **Expenses**. The Obligors agree, jointly and severally, to pay to the Agent upon demand the amount of any and all expenses, including the fees and expenses of any counsel for the Agent or for any of the Lenders, which the Agent or any of the Lenders may incur in connection with the exercise or enforcement of any of the rights or interests of the Agent against the Subordinated Creditors or the Obligors hereunder.
- SECTION 14. Notices. All notices and other communications provided for hereunder shall be in writing and shall be mailed, telecopied or delivered, if to the addresses specified below, or as to any such Person at such other address as shall be designated by such Person in a written notice to each such other Person complying as to delivery with the terms of this Section 14. All such demands, notices, and other communications shall be effective (i) if mailed, when received or three Business Days after mailing, whichever occurs first, (ii) if telecopied, when transmitted and appropriate confirmation is received or (iii) if delivered, upon delivery.

If to a Subordinated Creditor:

If to Obligors:

To the address set forth under such Subordinated Creditor's name on the signature pages hereof

With a copy to:

Bryan Cave LLP

1290 Avenue of the Americas New York, NY 10104 FAX: (212) 541-1432 Attention: Eric L. Cohen, Esq.

Matrix Service Company 10701 East Ute Street Tulsa, OK 74116

Attention: Michael J. Hall, CEO Telephone: (918) 838-8822 FAX: (918) 838-8810

With a copy to:

Conner & Winters, LLP 3700 First Place Tower 15 East 5th Street Tulsa, OK 74103-4344 FAX: (918) 586-8548

Attention: Mark D. Berman, Esq.

If to Agent:

JPMorgan Chase Bank, N.A. P.O. Box 655415 Dallas, TX 75265-5414 Address for Deliveries: 1717 Main Street 4th Floor

Telephone: (214) 290-2799 FAX: (214) 290-2740

With a copy to:

Dallas, TX 75201

Gable & Gotwals 1100 ONEOK Plaza 100 West Fifth Street Tulsa, OK 74103 FAX: (918) 595-4990

Attention: Jeffrey D. Hassell, Esq.

SECTION 15. Miscellaneous.

- (a) No amendment of any provision of this Agreement shall be effective unless it is in writing and signed by the Subordinated Creditors, the Obligors and the Agent, and no waiver of any provision of this Agreement, and no consent to any departure therefrom, shall be effective unless it is in writing and signed by the Agent, and then such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given.
- (b) No failure on the part of the Agent to exercise, and no delay in exercising, any right hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right preclude any other or further exercise thereof or the exercise of any other right.
- (c) Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining portions hereof or thereof or affecting the validity or enforceability of such provision in any other jurisdiction.
- (d) This Agreement shall (i) be binding on the Subordinated Creditors and the Obligors and their respective successors and assigns and (ii) inure, together with all rights and remedies of the Agent, to the benefit of the Agent and its successors, transferees and assigns. Without limiting the generality of clause (ii) of the immediately preceding sentence, any of the Lenders may assign or otherwise transfer a Note, and its rights under any other Senior Document, to any other Person, and such other Person shall thereupon become vested with all of

the benefits in respect thereof in favor of the Lender herein or otherwise. None of the rights or obligations of the Subordinated Creditors or the Obligors hereunder may be assigned or otherwise transferred without the prior written consent of the Agent except as expressly otherwise provided herein.

- (e) This Agreement shall be governed by and construed in accordance with the laws of the State of New York.
- (f) THE PARTIES HEREBY IRREVOCABLY SUBMIT TO THE EXCLUSIVE JURISDICTION OF ANY UNITED STATES FEDERAL COURT OR STATE COURT SITTING IN THE STATE OF NEW YORK IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT AND THE PARTIES HEREBY IRREVOCABLY AGREE THAT ALL CLAIMS IN RESPECT OF SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN ANY SUCH COURT AND IRREVOCABLY WAIVE ANY OBJECTION ANY OF THEM MAY NOW OR HEREAFTER HAVE AS TO THE VENUE OF ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN SUCH A COURT OR THAT SUCH COURT IS AN INCONVENIENT FORUM. NOTHING HEREIN SHALL LIMIT THE RIGHT OF THE AGENT OR ANY LENDER TO BRING PROCEEDINGS AGAINST ANY OBLIGOR IN THE COURTS OF ANY OTHER JURISDICTION PROVIDED THAT ANY JUDICIAL PROCEEDING INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH THIS AGREEMENT SHALL BE BROUGHT ONLY IN A UNITED STATES FEDERAL OR STATE COURT SITTING IN NEW YORK.
- (g) THE SUBORDINATED CREDITORS, THE OBLIGORS AND THE AGENT (BY ITS ACCEPTANCE OF THIS AGREEMENT) EACH HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM CONCERNING THIS AGREEMENT, ANY SENIOR DOCUMENT OR ANY AMENDMENT, MODIFICATION OR OTHER DOCUMENT NOW OR HEREAFTER DELIVERED IN CONNECTION WITH ANY OF THE FOREGOING, AND AGREE THAT ANY SUCH ACTION, PROCEEDING OR COUNTERCLAIM SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.
- (h) In the event that any suit or action is instituted by a party hereto to enforce the provisions of this Agreement, the prevailing party in such dispute shall be entitled to recover its reasonable attorneys fees and expenses incurred in connection therewith from the non-prevailing party. Nothing in this Section 15(h) shall impair or limit the rights of the Agent and Lenders under Section 13 above.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK.]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered by their respective officers thereunto duly authorized, as of the date first above written.

BORROWER:

 ${\bf MATRIX\ SERVICE\ COMPANY}, \ a\ Delaware\ corporation$

By: /s/ George L. Austin
George L. Austin, Vice President

OTHER OBLIGORS:

MATRIX SERVICE INC., an Oklahoma corporation; MATRIX SERVICE INDUSTRIAL CONTRACTORS, INC. (formerly known as MATRIX SERVICE MID-CONTINENT, INC.), an Oklahoma corporation; MATRIX SERVICE, INC. CANADA, an Ontario, Canada corporation; HAKE GROUP, INC., a Delaware corporation; BOGAN, INC. (including Fiberspec, a division), a Pennsylvania corporation; MATRIX SERVICE SPECIALIZED TRANSPORT, INC. (formerly known as FRANK W. HAKE, INC.), a Pennsylvania corporation; HOVER SYSTEMS, INC., a Pennsylvania corporation; I & S, INC., a Pennsylvania corporation; MCBISH MANAGEMENT, INC., a Pennsylvania corporation; MECHANICAL CONSTRUCTION, INC., a Delaware corporation; MID-ATLANTIC CONSTRUCTORS, INC., a Pennsylvania corporation; TALBOT REALTY, INC., a Pennsylvania corporation; BISH INVESTMENTS, INC., a Delaware corporation; I & S JOINT VENTURE, L.L.C., a Pennsylvania limited liability company

By: /s/ George L. Austin

George L. Austin, Vice President

AGENT:

JPMORGAN CHASE BANK, N.A.

(successor by merger to Bank One, N.A. (Main Office Chicago)) Lender, LC Issuer and as Agent

By: /s/ Hal E. Fudge
Hal E. Fudge, First Vice President

By:		
Name:		
Title:		
ADDRES	S FOR NOTICE	
c/o:	S FOR NOTICE	
	S FOR NOTICE	
c/o: Street:		
c/o:	/Zip:	
c/o: Street: City/State	/Zip:	

$\underline{Schedule\ I}$

- 1. Securities Purchase Agreement dated April 22, 2005 between Borrower and the Subordinated Creditors.
- 2. Form of Subordinated Note, attached as Exhibit A to the Securities Purchase Agreement.
- 3. Registration Rights Agreement dated April 22, 2005 between Borrower and the Subordinated Creditors.

REGISTRATION RIGHTS AGREEMENT

This Registration Rights Agreement (this "Agreement") is made and entered into as of April 22, 2005, by and among Matrix Service Company, a Delaware corporation (the "Company"), and the investors signatory hereto (each a "Investor" and collectively, the "Investors").

This Agreement is made pursuant to the Securities Purchase Agreement, dated as of April 22, 2005, among the Company and the Investors (the "Purchase Agreement").

The Company and the Investors hereby agree as follows:

1. <u>Definitions</u>. Capitalized terms used and not otherwise defined herein that are defined in the Purchase Agreement shall have the meanings given such terms in the Purchase Agreement. As used in this Agreement, the following terms shall have the respective meanings set forth in this Section 1:

"Effective Date" means the date that a Registration Statement filed pursuant to Sections 2(a) or 2(b) is first declared effective by the Commission.

"Effectiveness Date" means (a) with respect to the initial Registration Statement required to be filed under Section 2(a), the earlier of: (i) the 120th day following the Closing Date, and (ii) the fifth Trading Day following the date on which the Company is notified by the Commission that the initial Registration Statement will not be reviewed or is no longer subject to further review and comments, and (b) with respect to any additional Registration Statements that may be required pursuant to Section 2(b), the earlier of (i) the 120th day following (x) if such Registration Statement is required because the Commission shall have notified the Company in writing that certain Registrable Securities were not eligible for inclusion on a previously filed Registration Statement, the date or time on which the Commission shall indicate as being the first date or time that such Registrable Securities may then be included in a Registration Statement, or (y) if such Registration Statement is required for a reason other than as described in (x) above, the date on which the Company first knows, or reasonably should have known, that such additional Registration Statement(s) is required, and (ii) the fifth Trading Day following the date on which the Company is notified by the Commission that such additional Registration Statement will not be reviewed or is no longer subject to further review and comment.

"Effectiveness Period" shall have the meaning set forth in Section 2(a).

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Filing Date" means (a) with respect to the initial Registration Statement required to be filed under Section 2(a), the 30th day following the Closing Date, and (b) with respect to any additional Registration Statements that may be required pursuant to Section 2(b), the 30th day following (x) if such Registration Statement is required because the Commission shall have notified the Company in writing that certain Registrable Securities were not eligible for inclusion on a previously filed Registration Statement, the date or time on which the Commission shall indicate as being the first date or time that such Registrable Securities may then be included in a Registration Statement, or (y) if such Registration Statement is required for a reason other than as described in (x) above, the date on which the Company first knows, or reasonably should have known, that such additional Registration Statement(s) is required.

- "Holder" or "Holders" means the holder or holders, as the case may be, from time to time of Registrable Securities.
- "Indemnified Party" shall have the meaning set forth in Section 5(c).
- "Indemnifying Party" shall have the meaning set forth in Section 5(c).
- "Losses" shall have the meaning set forth in Section 5(a).
- "Notes" means, collectively, each of the promissory notes issued or issuable under the Purchase Agreement.
- "Proceeding" means an action, claim, suit, investigation or proceeding (including, without limitation, an investigation or partial proceeding, such as a deposition), whether commenced or threatened in writing.

"Prospectus" means the prospectus included in a Registration Statement (including, without limitation, a prospectus that includes any information previously omitted from a prospectus filed as part of an effective registration statement in reliance upon Rule 430A promulgated under the Securities Act), as amended or supplemented by any prospectus supplement, with respect to the terms of the offering of any portion of the Registrable Securities covered by a Registration Statement, and all other amendments and supplements to the Prospectus, including post-effective amendments, and all material incorporated by reference or deemed to be incorporated by reference in such Prospectus.

"Registrable Securities" means the Underlying Shares issuable upon conversion of the Notes and any accrued and unpaid interest thereon, together with any securities issued or issuable upon any stock split, dividend or other distribution, recapitalization or similar event, or any conversion price adjustment with respect thereto.

"Registration Statement" means each of the following: (i) an initial registration statement under Section 2(a) which is required to register the resale of the Registrable Securities, (ii) each additional registration statement, if any, contemplated by Section 2(b), and including, in each case, the Prospectus, amendments and supplements to each such registration statement or Prospectus, including pre- and post-effective

amendments, all exhibits thereto, and all material incorporated by reference or deemed to be incorporated by reference in such registration statement.

"Rule 144" means Rule 144 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"Rule 415" means Rule 415 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"Rule 424" means Rule 424 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect as such Rule.

"Securities Act" means the Securities Act of 1933, as amended.

2. Registration.

(a) On or prior to each Filing Date, the Company shall prepare and file with the Commission a Registration Statement covering the resale of all Registrable Securities not already covered by an existing and effective Registration Statement for an offering to be made on a continuous basis pursuant to Rule 415. The Registration Statement shall be on Form S-3 (except if the Company is not then eligible to register for resale the Registrable Securities on Form S-3, in which case such registration shall be on another appropriate form for such purpose) and shall contain (except if otherwise required pursuant to written comments received from the Commission upon a review of such Registration Statement) the "Plan of Distribution" in the form attached hereto as Annex A. The Company shall cause each Registration Statement to be declared effective under the Securities Act as soon as possible but, in any event, no later than its Effectiveness Date, and shall use its best efforts to keep each Registration Statement continuously effective under the Securities Act until the fifth year after the date that the Registration Statement is declared effective by the Commission or such earlier date when all Registrable Securities covered by the Registration Statement have been sold or may be sold without volume restrictions pursuant to Rule 144(k) as determined by the counsel to the Company pursuant to a written opinion letter to such effect, addressed and acceptable to the Company's transfer agent and the affected Holders (the "Effectiveness Period").

(b) If for any reason the Commission does not permit all of the Registrable Securities to be included in the Registration Statement filed pursuant to Section 2(a) or for any other reason all Registrable Securities then outstanding are not then included in an effective Registration Statement, then the Company shall prepare and file as soon as reasonably possible after the date on which the Commission shall indicate as being the first date or time that such filing may be made, but in any event by the Filing

Date therefore, an additional Registration Statement covering the resale of all Registrable Securities not already covered by an existing and effective Registration Statement for an offering to be made on a continuous basis pursuant to Rule 415, on Form S-3 (except if the Company is not then eligible to register for resale the Registrable Securities on Form S-3, in which case such registration shall be on another appropriate form for such purpose). Each such Registration Statement shall contain (except if otherwise required pursuant to written comments received from the Commission upon a review of such Registration Statement) the "Plan of Distribution" attached hereto as <u>Annex A</u>. The Company shall use its best efforts to cause each such Registration Statement to be declared effective under the Securities Act as soon as possible but, in any event, no later than its Effectiveness Date, and shall use its best efforts to keep such Registration Statement continuously effective under the Securities Act during the entire Effectiveness Period.

(c) If: (i) a Registration Statement is not filed on or prior to its Filing Date (if the Company files a Registration Statement without affording the Holders the opportunity to review and comment on the same as required by Section 3(a) hereof, the Company shall not be deemed to have satisfied this clause (i)), or (ii) a Registration Statement is not declared effective by the Commission on or prior to its required Effectiveness Date, or (iii) after its Effective Date, without regard for the reason thereunder or efforts therefore, such Registration Statement ceases for any reason to be effective and available to the Holders as to all Registrable Securities to which it is required to cover at any time prior to the expiration of its Effectiveness Period for more than an aggregate of 20 Trading Days (which need not be consecutive) during any 18-month period (the parties understand that any unused days in a particular period may not be carried forward to any subsequent period) (any such failure or breach being referred to as an "Event," and for purposes of clauses (i) or (ii) the date on which such Event occurs, or for purposes of clause (iii) the date which such 20 Trading Day-period is exceeded, being referred to as "Event Date"), then in addition to any other rights the Holders may have hereunder or under applicable law: (x) on each such Event Date the Company shall pay to each Holder an amount in cash, as partial liquidated damages and not as a penalty, equal to 1.0% of the aggregate Investment Amount paid by such Holder for Notes pursuant to the Purchase Agreement; and (y) on each monthly anniversary of each such Event Date (if the applicable Event shall not have been cured by such date) until the applicable Event is cured, the Company shall pay to each Holder an amount in cash, as partial liquidated damages and not as a penalty, equal to 1.0% of the aggregate Investment Amount paid by such Holder for Notes pursuant to the Purchase Agreement. If the Company fails to pay any partial liquidated damages pursuant to this Section in full within seven days after the date payable, the Company will pay interest thereon at a rate of 10% per annum (or such lesser maximum amount that is permitted to be paid by applicable law) to the Holder, accruing daily from the date such partial liquidated damages are due until such amounts, plus all such interest thereon, are paid in full. The partial liquidated damages pursuant to the terms hereof shall apply on a daily pro-rata basis for any portion of a month prior to the cure of an Event, except in the case of the first Event Date.

(d) Each Holder agrees to furnish to the Company a completed Questionnaire in the form attached to this Agreement as Annex B (a "Selling Holder Questionnaire"). The Company shall not be required to include the Registrable Securities of a Holder in a Registration Statement and shall not be required to pay any liquidated or other damages under Section 2(c) to any Holder who fails to furnish to the Company a fully completed Selling Holder Questionnaire at least seven Trading Days prior to the Filing Date (subject to the requirements set forth in Section 3(a)).

3. Registration Procedures.

In connection with the Company's registration obligations hereunder, the Company shall:

- (a) Not less than four Trading Days prior to the filing of a Registration Statement or any related Prospectus or any amendment or supplement thereto, the Company shall furnish to each Holder copies of the "Selling Stockholders" section of such document, the "Plan of Distribution" and any risk factor contained in such document that addresses specifically this transaction or the Selling Stockholders, as proposed to be filed which documents will be subject to the review of such Holder. The Company shall not file a Registration Statement, any Prospectus or any amendments or supplements thereto in which the "Selling Stockholder" section thereof differs from the disclosure received from a Holder in its Selling Holder Questionnaire (as amended or supplemented).
- (b) (i) Prepare and file with the Commission such amendments, including post-effective amendments, to each Registration Statement and the Prospectus used in connection therewith as may be necessary to keep such Registration Statement continuously effective as to the applicable Registrable Securities for its Effectiveness Period and prepare and file with the Commission such additional Registration Statements in order to register for resale under the Securities Act all of the Registrable Securities; (ii) cause the related Prospectus to be amended or supplemented by any required Prospectus supplement, and as so supplemented or amended to be filed pursuant to Rule 424; (iii) respond as promptly as reasonably possible to any comments received from the Commission with respect to each Registration Statement or any amendment thereto and, as promptly as reasonably possible provide the Holders true and complete copies of all correspondence from and to the Commission relating to such Registration Statement that would not result in the disclosure to the Holders of material and non-public information concerning the Company; and (iv) comply in all material respects with the provisions of the Securities Act and the Exchange Act with respect to the Registration Statements and the disposition of all Registrable Securities covered by each Registration Statement.
- (c) Notify the Holders as promptly as reasonably possible (and, in the case of (i)(A) below for amendments and supplements that include modified or additional disclosure concerning a Holder in the Selling Stockholders section or the Plan of Distribution, not less than three Trading Days prior to such filing) and (if requested by any such Person) confirm such notice in writing no later than one Trading Day following the day (i)(A) when a Prospectus or any Prospectus supplement or post-effective

amendment to a Registration Statement is proposed to be filed; (B) when the Commission notifies the Company whether there will be a "review" of such Registration Statement and whenever the Commission comments in writing on such Registration Statement (the Company shall provide true and complete copies of any such comments and all written responses thereto to each Holder that pertains to such Holder as a Selling Shareholder or to the Plan of Distribution, but not information which the Company believes would constitute material and non-public information); and (C) with respect to each Registration Statement or any post-effective amendment, when the same has become effective; (ii) of any request by the Commission or any other Federal or state governmental authority for amendments or supplements to a Registration Statement or Prospectus or for additional information; (iii) of the issuance by the Commission of any stop order suspending the effectiveness of a Registration Statement covering any or all of the Registrable Securities or the initiation of any Proceedings for that purpose; (iv) of the receipt by the Company of any notification with respect to the suspension of the qualification or exemption from qualification of any of the Registrable Securities for sale in any jurisdiction, or the initiation or threatening of any Proceeding for such purpose; and (v) of the occurrence of any event or passage of time that makes the financial statements included in a Registration Statement ineligible for inclusion therein or any statement made in such Registration Statement or Prospectus or any document incorporated or deemed to be incorporated therein by reference untrue in any material respect or that requires any revisions to such Registration Statement, Prospectus or other documents so that, in the case of such Registration Statement or the Prospectus, as the case may be, it will not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the

- (d) Use its reasonable best efforts to avoid the issuance of, or, if issued, obtain the withdrawal of (i) any order suspending the effectiveness of a Registration Statement, or (ii) any suspension of the qualification (or exemption from qualification) of any of the Registrable Securities for sale in any jurisdiction, at the earliest practicable moment.
- (e) Furnish to each Holder, without charge, at least one conformed copy of each Registration Statement and each amendment thereto and all exhibits to the extent requested by such Person (excluding those previously furnished or incorporated by reference) promptly after the filing of such documents with the Commission.
- (f) Promptly deliver to each Holder, without charge, as many copies of each Prospectus or Prospectuses (including each form of prospectus) and each amendment or supplement thereto as such Persons may reasonably request. The Company hereby consents to the use of such Prospectus and each amendment or supplement thereto by each of the selling Holders in connection with the offering and sale of the Registrable Securities covered by such Prospectus and any amendment or supplement thereto during the Effectiveness Period until such time, if any, as the Company provides the selling Holders with written notice that such Prospectus, as amended or supplemented, is outdated or defective.

- (g) Prior to any public offering of Registrable Securities, use its reasonable best efforts to register or qualify or cooperate with the selling Holders in connection with the registration or qualification (or exemption from such registration or qualification) of such Registrable Securities for offer and sale under the securities or Blue Sky laws of all jurisdictions within the United States, to keep each such registration or qualification (or exemption therefrom) effective during the Effectiveness Period and to do any and all other acts or things necessary or advisable to enable the disposition in such jurisdictions of the Registrable Securities covered by the Registration Statements; provided, that the Company shall not be required to qualify generally to do business in any jurisdiction where it is not then so qualified or subject the Company to any material tax in any such jurisdiction where it is not then so subject.
- (h) Cooperate with the Holders to facilitate the timely preparation and delivery of certificates representing Registrable Securities to be delivered to a transferee pursuant to the Registration Statements, which certificates shall be free, to the extent permitted by the Purchase Agreement, of all restrictive legends, and to enable such Registrable Securities to be in such denominations and registered in such names as any such Holders may request.
- (i) Upon the occurrence of any event contemplated by Section 3(c)(v), as promptly as reasonably possible, prepare a supplement or amendment, including a post-effective amendment, to the affected Registration Statements or a supplement to the related Prospectus or any document incorporated or deemed to be incorporated therein by reference, and file any other required document so that, as thereafter delivered, no Registration Statement nor any Prospectus will contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.
- 4. Registration Expenses. All fees and expenses incident to the performance of or compliance with this Agreement by the Company shall be borne by the Company whether or not any Registrable Securities are sold pursuant to a Registration Statement. The fees and expenses referred to in the foregoing sentence shall include, without limitation, (i) all registration and filing fees (including, without limitation, fees and expenses (A) with respect to filings required to be made with any Trading Market on which the Common Stock is then listed for trading, and (B) in compliance with applicable state securities or Blue Sky laws), (ii) printing expenses (including, without limitation, expenses of printing certificates for Registrable Securities and of printing prospectuses if the printing of prospectuses is reasonably requested by the Holders of a majority of the Registrable Securities included in the Registration Statement), (iii) messenger, telephone and delivery expenses, (iv) fees and disbursements of counsel for the Company, but not counsel for any Holder other than pursuant to Article 5 herein, (v) Securities Act liability insurance, if the Company so desires such insurance, and (vi) fees and expenses of all other Persons retained by the Company in connection with the consummation of the transactions contemplated by this Agreement. In addition, the Company shall be responsible for all of its internal expenses incurred in connection with the consummation of the transactions contemplated by this Agreement (including,

without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties), the expense of any annual audit and the fees and expenses incurred in connection with the listing of the Registrable Securities on any securities exchange as required hereunder. Notwithstanding the foregoing or anything in this Agreement to the contrary, each Holder shall pay all commissions, placement agent fees and underwriting discounts and commissions with respect to any Registrable Securities sold by such Holder.

5. Indemnification.

- (a) Indemnification by the Company. The Company shall, notwithstanding any termination of this Agreement, indemnify and hold harmless each Holder, the officers, directors, agents, investment advisors, partners, members, shareholders and employees of each of them, each Person who controls any such Holder (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) and the officers, directors, agents and employees of each such controlling Person, to the fullest extent permitted by applicable law, from and against any and all losses, claims, damages, liabilities, costs (including, without limitation, reasonable costs of preparation and reasonable attorneys' fees) and expenses (collectively, "Losses"), as incurred, arising out of or relating to any untrue or alleged untrue statement of a material fact contained in any Registration Statement, any Prospectus or any form of prospectus or in any amendment or supplement thereto or in any preliminary prospectus, or arising out of or relating to any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein (in the case of any Prospectus or form of prospectus or supplement thereto, in light of the circumstances under which they were made) not misleading, except to the extent, but only to the extent, that (1) such untrue statements or omissions are based solely upon information regarding such Holder furnished in writing to the Company by such Holder expressly for use therein, or to the extent that such information relates to such Holder or such Holder's proposed method of distribution of Registrable Securities and was reviewed and expressly approved in writing by such Holder expressly for use in the Registration Statement, such Prospectus or such form of Prospectus or in any amendment or supplement thereto (it being understood that the Holder has approved Annex A hereto for this purpose) or (2) in the case of an occurrence of an event of the type specified in Section 3(c)(ii)-(v), the use by such Holder of an outdated or defective Prospectus after the Company has notified such Holder in writing that the Prospectus is outdated or defective and prior to the receipt by such Holder of an Advice or an amended or supplemented Prospectus, but only if and to the extent that following the receipt of the Advice or the amended or supplemented Prospectus the misstatement or omission giving rise to such Loss would have been corrected. The Company shall notify the Holders promptly of the institution, threat or assertion of any Proceeding of which the Company is aware in connection with the transactions contemplated by this Agreement.
- (b) <u>Indemnification by Holders</u>. Each Holder shall, severally and not jointly, indemnify and hold harmless the Company, its directors, officers, agents and employees, each Person who controls the Company (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act), and the directors, officers, agents

or employees of such controlling Persons, to the fullest extent permitted by applicable law, from and against all Losses, as incurred, arising solely out of or based solely upon: (x) such Holder's failure to comply with the prospectus delivery requirements of the Securities Act or (y) any untrue statement of a material fact contained in any Registration Statement, any Prospectus, or any form of prospectus, or in any amendment or supplement thereto, or arising solely out of or based solely upon any omission of a material fact required to be stated therein or necessary to make the statements therein not misleading to the extent, but only to the extent that, (1) such untrue statements or omissions are based solely upon information regarding such Holder furnished in writing to the Company by such Holder expressly for use therein, or to the extent that such information relates to such Holder or such Holder's proposed method of distribution of Registrable Securities and was reviewed and expressly approved in writing by such Holder expressly for use in the Registration Statement (it being understood that the Holder has approved Annex A hereto for this purpose), such Prospectus or such form of Prospectus or in any amendment or supplement thereto or (2) in the case of an occurrence of an event of the type specified in Section 3(c)(ii)-(v), the use by such Holder of an outdated or defective Prospectus after the Company has notified such Holder in writing that the Prospectus is outdated or defective and prior to the receipt by such Holder of an Advice or an amended or supplemented Prospectus, but only if and to the extent that following the receipt of the Advice or the amended or supplemented Prospectus the misstatement or omission giving rise to such Loss would have been corrected. In no event shall the liability of any selling Holder hereunder be greater in amount than the dollar amount of the net proceeds received by such Holder upon the sale of the Registrable Securities giving rise to such indemnification obligation.

(c) <u>Conduct of Indemnification Proceedings</u>. If any Proceeding shall be brought or asserted against any Person entitled to indemnity hereunder (an "**Indemnified Party**"), such Indemnified Party shall promptly notify the Person from whom indemnity is sought (the "**Indemnifying Party**") in writing, and the Indemnifying Party shall assume the defense thereof, including the employment of counsel reasonably satisfactory to the Indemnified Party and the payment of all fees and expenses incurred in connection with defense thereof; provided, that the failure of any Indemnified Party to give such notice shall not relieve the Indemnifying Party of its obligations or liabilities pursuant to this Agreement, except (and only) to the extent that it shall be finally determined by a court of competent jurisdiction (which determination is not subject to appeal or further review) that such failure shall have proximately and materially adversely prejudiced the Indemnifying Party.

An Indemnified Party shall have the right to employ separate counsel in any such Proceeding and to participate in the defense thereof, but the fees and expenses of such counsel shall be at the expense of such Indemnified Party or Parties unless: (1) the Indemnifying Party has agreed in writing to pay such fees and expenses; (2) the Indemnifying Party shall have failed promptly to assume the defense of such Proceeding and to employ counsel reasonably satisfactory to such Indemnified Party in any such Proceeding; or (3) the named parties to any such Proceeding (including any impleaded parties) include both such Indemnified Party and the Indemnifying Party, and such Indemnified Party shall have been advised by counsel that a conflict of interest is likely

to exist if the same counsel were to represent such Indemnified Party and the Indemnifying Party (in which case, if such Indemnified Party notifies the Indemnifying Party in writing that it elects to employ separate counsel at the expense of the Indemnifying Party, the Indemnifying Party shall not have the right to assume the defense thereof and such counsel shall be at the expense of the Indemnifying Party). The Indemnifying Party shall not be liable for any settlement of any such Proceeding effected without its written consent, which consent shall not be unreasonably withheld. No Indemnifying Party shall, without the prior written consent of the Indemnified Party, effect any settlement of any pending Proceeding in respect of which any Indemnified Party is a party, unless such settlement includes an unconditional release of such Indemnified Party from all liability on claims that are the subject matter of such Proceeding.

All reasonable fees and expenses of the Indemnified Party (including reasonable fees and expenses to the extent incurred in connection with investigating or preparing to defend such Proceeding in a manner not inconsistent with this Section) shall be paid to the Indemnified Party, as incurred, within ten Trading Days of written notice thereof to the Indemnifying Party (regardless of whether it is ultimately determined that an Indemnified Party is not entitled to indemnification hereunder; provided, that the Indemnifying Party may require such Indemnified Party to undertake to reimburse all such fees and expenses to the extent it is finally judicially determined that such Indemnified Party is not entitled to indemnification hereunder).

(d) <u>Contribution</u>. If a claim for indemnification under Section 5(a) or 5(b) is unavailable to an Indemnified Party (by reason of public policy or otherwise), then each Indemnifying Party, in lieu of indemnifying such Indemnified Party, shall contribute to the amount paid or payable by such Indemnified Party as a result of such Losses, in such proportion as is appropriate to reflect the relative fault of the Indemnifying Party and Indemnified Party in connection with the actions, statements or omissions that resulted in such Losses as well as any other relevant equitable considerations. The relative fault of such Indemnifying Party and Indemnified Party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission of a material fact, has been taken or made by, or relates to information supplied by, such Indemnifying Party or Indemnified Party, and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such action, statement or omission. The amount paid or payable by a party as a result of any Losses shall be deemed to include, subject to the limitations set forth in Section 5(c), any reasonable attorneys' or other reasonable fees or expenses incurred by such party in connection with any Proceeding to the extent such party would have been indemnified for such fees or expenses if the indemnification provided for in this Section was available to such party in accordance with its terms.

The parties hereto agree that it would not be just and equitable if contribution pursuant to this Section 5(d) were determined by pro rata allocation or by any other method of allocation that does not take into account the equitable considerations referred to in the immediately preceding paragraph. Notwithstanding the

provisions of this Section 5(d), no Holder shall be required to contribute, in the aggregate, any amount in excess of the amount by which the proceeds actually received by such Holder from the sale of the Registrable Securities subject to the Proceeding exceeds the amount of any damages that such Holder has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission.

The indemnity and contribution agreements contained in this Section are in addition to any liability that the Indemnifying Parties may have to the Indemnified Parties.

6. Miscellaneous.

- (a) <u>Remedies</u>. In the event of a breach by the Company or by a Holder, of any of their obligations under this Agreement, each Holder or the Company, as the case may be, in addition to being entitled to exercise all rights granted by law and under this Agreement, including recovery of damages, will be entitled to specific performance of its rights under this Agreement. The Company and each Holder agree that monetary damages would not provide adequate compensation for any losses incurred by reason of a breach by it of any of the provisions of this Agreement and hereby further agrees that, in the event of any action for specific performance in respect of such breach, it shall waive the defense that a remedy at law would be adequate.
- (b) <u>No Piggyback on Registrations</u>. Except as and to the extent specified in <u>Schedule 3.1(v)</u> to the Purchase Agreement, neither the Company nor any of its security holders (other than the Holders in such capacity pursuant hereto) may include securities of the Company in a Registration Statement other than the Registrable Securities, and the Company shall not during the Registration Period enter into any agreement providing any such right to any of its security holders.
- (c) <u>Compliance</u>. Each Holder covenants and agrees that it will comply with the prospectus delivery requirements of the Securities Act as applicable to it in connection with sales of Registrable Securities pursuant to the Registration Statement.
- (d) <u>Discontinued Disposition</u>. Each Holder agrees by its acquisition of such Registrable Securities that, upon receipt of a notice from the Company of the occurrence of any event of the kind described in Section 3(c), such Holder will forthwith discontinue disposition of such Registrable Securities under the Registration Statement until such Holder's receipt of the copies of the supplemented Prospectus and/or amended Registration Statement or until it is advised in writing (the "Advice") by the Company that the use of the applicable Prospectus may be resumed, and, in either case, has received copies of any additional or supplemental filings that are incorporated or deemed to be incorporated by reference in such Prospectus or Registration Statement. The Company may provide appropriate stop orders to enforce the provisions of this paragraph.
- (e) <u>Piggy-Back Registrations</u>. If at any time during the Effectiveness Period there is not an effective Registration Statement covering all of the Registrable

Securities and the Company shall determine to prepare and file with the Commission a registration statement relating to an offering for its own account or the account of others under the Securities Act of any of its equity securities, other than on Form S-4 or Form S-8 (each as promulgated under the Securities Act) or their then equivalents relating to equity securities to be issued solely in connection with any acquisition of any entity or business or equity securities issuable in connection with stock option or other employee benefit plans, then the Company shall send to each Holder written notice of such determination and, if within fifteen days after receipt of such notice, any such Holder shall so request in writing, the Company shall include in such registration statement all or any part of such Registrable Securities such holder requests to be registered, subject to customary underwriter cutbacks applicable to all holders of registration rights.

- (f) <u>Amendments and Waivers</u>. The provisions of this Agreement, including the provisions of this sentence, may not be amended, modified or supplemented, and waivers or consents to departures from the provisions hereof may not be given, unless the same shall be in writing and signed by the Company and the Holders of a majority of the then outstanding Registrable Securities.
- (g) Notices. Any and all notices or other communications or deliveries required or permitted to be provided hereunder shall be in writing and shall be deemed given and effective on the earliest of (i) the date of transmission, if such notice or communication is delivered via facsimile at the facsimile telephone number specified in this Section prior to 6:30 p.m. (New York City time) on a Trading Day, (ii) the Trading Day after the date of transmission, if such notice or communication is delivered via facsimile at the facsimile telephone number specified in this Agreement later than 6:30 p.m. (New York City time) on any date and earlier than 11:59 p.m. (New York City time) on such date, (iii) the Trading Day following the date of mailing, if sent by nationally recognized overnight courier service, or (iv) upon actual receipt by the party to whom such notice is required to be given. The address for such notices and communications shall be as follows:

If to the Company: Matrix Service Company

10701 E. Ute Street Tulsa, OK 74116-1517 Facsimile No.: (918) 838-8810 Attn: George L. Austin

With a copy to: Conner & Winters, LLP

3700 First Place Tower 15 East 5th Street Tulsa, OK 74103-4344

Facsimile No.: (918) 586-8548 Attn: Mark D. Berman, Esq.

If to a Investor: To the address set forth under such Investor's name on the signature pages hereto.

If to any other Person who is then the registered Holder: To the address of such Holder as it appears in the stock transfer books of the Company or such other address as may be designated in writing hereafter, in the same manner, by such Person.

- (h) <u>Successors and Assigns</u>. This Agreement shall inure to the benefit of and be binding upon the successors and permitted assigns of each of the parties and shall inure to the benefit of each Holder. The Company may not assign its rights or obligations hereunder without the prior written consent of each Holder. Each Holder may assign its respective rights hereunder in the manner and to the Persons as permitted under the Purchase Agreement so long as such assignment complies with the Purchase Agreement.
- (i) <u>Execution and Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which when so executed shall be deemed to be an original and, all of which taken together shall constitute one and the same Agreement. In the event that any signature is delivered by facsimile transmission, such signature shall create a valid binding obligation of the party executing (or on whose behalf such signature is executed) the same with the same force and effect as if such facsimile signature were the original thereof.
- (j) Governing Law. All questions concerning the construction, validity, enforcement and interpretation of this Agreement shall be governed by and construed and enforced in accordance with the internal laws of the State of Delaware, without regard to the principles of conflicts of law thereof. Each party agrees that all Proceedings concerning the interpretations, enforcement and defense of the transactions contemplated by this Agreement (whether brought against a party hereto or its respective Affiliates, employees or agents) shall be commenced exclusively in the state and federal courts sitting in the State of Delaware (the "Delaware Courts"). Each party hereto hereby irrevocably submits to the exclusive jurisdiction of the Delaware Courts for the adjudication of any dispute hereunder or in connection herewith or with any transaction contemplated hereby or discussed herein, and hereby irrevocably waives, and agrees not to assert in any Proceeding, any claim that it is not personally subject to the jurisdiction of any Delaware Court, or that such Proceeding has been commenced in an improper or inconvenient forum. Each party hereto hereby irrevocably waives personal service of process and consents to process being served in any such Proceeding by mailing a copy thereof via registered or certified mail or overnight delivery (with evidence of delivery) to such party at the address in effect for notices to it under this Agreement and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law. Each party hereto hereby irrevocably waives, to the fullest extent permitted by applicable law, any and all right to trial by jury in any Proceeding arising out of or relating to this Agreement or the transactions contemplated hereby. If either party shall commence a Proceeding to enforce any provisions of this

Agreement, then the prevailing party in such Proceeding shall be reimbursed by the other party for its attorney's fees and other costs and expenses incurred with the investigation, preparation and prosecution of such Proceeding.

- (k) Cumulative Remedies. The remedies provided herein are cumulative and not exclusive of any remedies provided by law.
- (1) <u>Severability</u>. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction to be invalid, illegal, void or unenforceable, the remainder of the terms, provisions, covenants and restrictions set forth herein shall remain in full force and effect and shall in no way be affected, impaired or invalidated, and the parties hereto shall use their reasonable efforts to find and employ an alternative means to achieve the same or substantially the same result as that contemplated by such term, provision, covenant or restriction. It is hereby stipulated and declared to be the intention of the parties that they would have executed the remaining terms, provisions, covenants and restrictions without including any of such that may be hereafter declared invalid, illegal, void or unenforceable.
 - (m) Headings. The headings in this Agreement are for convenience of reference only and shall not limit or otherwise affect the meaning hereof.
- (n) <u>Independent Nature of Investors' Obligations and Rights</u>. The obligations of each Investor under any Transaction Document are several and not joint with the obligations of any other Investor, and no Investor shall be responsible in any way for the performance of the obligations of any other Investor under any Transaction Document. The decision of each Investor to purchase Securities pursuant to the Transaction Documents has been made by such Investor independently of any other Investor. Nothing contained herein or in any Transaction Document, and no action taken by any Investor pursuant thereto, shall be deemed to constitute the Investors as a partnership, an association, a joint venture or any other kind of entity, or create a presumption that the Investors are in any way acting in concert or as a group with respect to such obligations or the transactions contemplated by the Transaction Documents. Each Investor acknowledges that no other Investor has acted as agent for such Investor in connection with making its investment hereunder and that no Investor will be acting as agent of such Investor in connection with monitoring its investment in the Securities or enforcing its rights under the Transaction Documents. Each Investor shall be entitled to independently protect and enforce its rights, including without limitation the rights arising out of this Agreement or out of the other Transaction Documents, and it shall not be necessary for any other Investor to be joined as an additional party in any proceeding for such purpose. The Company acknowledges that each of the Investors has been provided with the same Transaction Documents for the purpose of closing a transaction with multiple Investors and not because it was required or requested to do so by any Investor.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK SIGNATURE PAGES TO FOLLOW]

IN WITNESS WHEREOF, the parties have executed this Registration Rights Agreement as of the date first written above.

MATRIX SERVICE COMPANY

By: /s/ George L. Austin

Name: George L. Austin Title: Chief Financial Officer

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK SIGNATURE PAGES OF INVESTORS TO FOLLOW]

IN WITNESS WHEREOF, the parties have executed this Registration Rights Agreement as of the date first written above.

NAME OF INVESTING ENTITY

AUTHORIZED SIGNATORY
By: Name: Title:
ADDRESS FOR NOTICE
c/o:
Street:
City/State/Zip:
Attention:
Tel:
Fax:
Email:

Plan of Distribution

The Selling Stockholders and any of their pledgees, donees, transferees, assignees and successors-in-interest may, from time to time, sell any or all of their shares of Common Stock on any stock exchange, market or trading facility on which the shares are traded or in private transactions. These sales may be at fixed or negotiated prices. The Selling Stockholders may use any one or more of the following methods when selling shares:

- ordinary brokerage transactions and transactions in which the broker-dealer solicits Investors;
- block trades in which the broker-dealer will attempt to sell the shares as agent but may position and resell a portion of the block as principal to facilitate the transaction;
- purchases by a broker-dealer as principal and resale by the broker-dealer for its account;
- an exchange distribution in accordance with the rules of the applicable exchange;
- privately negotiated transactions;
- to cover short sales made after the date that this Registration Statement is declared effective by the Commission;
- broker-dealers may agree with the Selling Stockholders to sell a specified number of such shares at a stipulated price per share;
- through the writing or settlement of options or other hedging transactions, whether through an options exchange or otherwise;
- a combination of any such methods of sale; and
 - any other method permitted pursuant to applicable law.

The Selling Stockholders may also sell shares under Rule 144 under the Securities Act, if available, rather than under this prospectus.

Broker-dealers engaged by the Selling Stockholders may arrange for other brokers-dealers to participate in sales. Broker-dealers may receive commissions or discounts from the Selling Stockholders (or, if any broker-dealer acts as agent for the purchaser of shares, from the purchaser) in amounts to be negotiated. The Selling Stockholders do not expect these commissions and discounts to exceed what is customary in the types of transactions involved.

In connection with the sale of our common stock or interests therein, the Selling Stockholders may enter into hedging transactions with broker-dealers or other financial institutions, which may in turn engage in short sales of the common stock in the course of hedging the positions they assume. The Selling Stockholders may also sell shares of our common stock short and deliver these securities to close out their short positions, or loan or pledge the common stock to broker-dealers that in turn may sell these securities. The Selling Stockholders may also enter into option or other transactions with broker-dealers or other financial institutions or the creation of one or more derivative securities which require the delivery to such broker-dealer or other financial institution of shares offered by this prospectus, which shares such broker-dealer or other financial institution may resell pursuant to this prospectus (as supplemented or amended to reflect such transaction).

Upon the Company being notified in writing by a Selling Stockholder that any material arrangement has been entered into with a broker-dealer for the sale of Common Stock through a block trade, special offering, exchange distribution or secondary distribution or a purchase by a broker or dealer, a supplement to this prospectus will be filed, if required, pursuant to Rule 424(b) under the Securities Act, disclosing (i) the name of each such Selling Stockholder and of the participating broker-dealer(s), (ii) the number of shares involved, (iii) the price at which such the shares of Common Stock were sold, (iv)the commissions paid or discounts or concessions allowed to such broker-dealer(s), where applicable, (v) that such broker-dealer(s) did not conduct any investigation to verify the information set out or incorporated by reference in this prospectus, and (vi) other facts material to the transaction. In addition, upon the Company being notified in writing by a Selling Stockholder that a donee or pledge intends to sell shares of Common Stock, a supplement to this prospectus will be filed if then required in accordance with applicable securities law.

The Selling Stockholders also may transfer the shares of Common Stock in other circumstances, in which case the transferees, pledgees or other successors in interest will be the selling beneficial owners for purposes of this prospectus.

The Selling Stockholders and any broker-dealers or agents that are involved in selling the shares may be deemed to be "underwriters" within the meaning of the Securities Act in connection with such sales. In such event, any commissions received by such broker-dealers or agents and any profit on the resale of the shares purchased by them may be deemed to be underwriting commissions or discounts under the Securities Act. Discounts, concessions, commissions and similar selling expenses, if any, that can be attributed to the sale of Securities will be paid by the Selling Stockholder and/or the purchasers. Each Selling Stockholder has represented and warranted to the Company that it acquired the securities subject to this registration statement in the ordinary course of such Selling Stockholder's business and, at the time of its purchase of such securities such Selling Stockholder had no agreements or understandings, directly or indirectly, with any person to distribute any such securities.

There can be no assurance that any Selling Stockholder will sell any or all of the shares of Common Stock registered pursuant to this Registration Statement, of which this prospectus forms a part.

The Company has advised each Selling Stockholder that it may not use shares registered on this Registration Statement to cover short sales of Common Stock made prior to the date on which this Registration Statement shall have been declared effective by the Commission. If a Selling Stockholder uses this prospectus for any sale of the Common Stock, it will be subject to the prospectus delivery requirements of the Securities Act. The Selling Stockholders will be responsible to comply with the applicable provisions of the Securities Act and Exchange Act, and the rules and regulations thereunder promulgated, including, without limitation, Regulation M, as applicable to such Selling Stockholders in connection with resales of their respective shares under this Registration Statement

The Company is required to pay all fees and expenses incident to the registration of the shares, but the Company will not receive any proceeds from the sale of the Common Stock. The Company and the Selling Stockholders have agreed to indemnify each other against certain losses, claims, damages and liabilities, including liabilities under the Securities Act.

Once sold under this Registration Statement, of which this prospectus forms a part, the shares of Common Stock will be freely tradable in the hands of persons other than the Company's affiliates.

MATRIX SERVICE COMPANY

Selling Securityholder Notice and Questionnaire

The undersigned beneficial owner of common stock (the "Common Stock"), of Matrix Service Company (the "Company") understands that the Company has filed or intends to file with the Securities and Exchange Commission (the "Commission") a Registration Statement for the registration and resale of the Registrable Securities, in accordance with the terms of the Registration Rights Agreement, dated as of April 22, 2005 (the "Registration Rights Agreement"), among the Company and the Investors named therein. A copy of the Registration Rights Agreement is available from the Company upon request at the address set forth below. All capitalized terms used and not otherwise defined herein shall have the meanings ascribed thereto in the Registration Rights Agreement.

The undersigned hereby provides the following information to the Company and represents and warrants that such information is accurate:

QUESTIONNAIRE

Full Legal Name of Selling Securityholder
Full Legal Name of Registered Holder (if not the same as (a) above) through which Registrable Securities Listed in Item 3 below are held:
Full Legal Name of Natural Control Person (which means a natural person who directly or indirectly alone or with others has power to vote or dispose of the securities covered by the questionnaire):
]

2. Address	s for Notices to Selling Securityholder:
	_
Telephone	_ :
Fax:	
E-mail:	
—— Contact Pe	erson:
3. Benefici	ial Ownership of Registrable Securities:
	Type and Principal Amount of Registrable Securities beneficially owned:
4. Broker-	Dealer Status:
(a)	Are you a broker-dealer?
	Yes □ No □
Note	e: If yes, the Commission's staff has indicated that you should be identified as an underwriter in the Registration Statement.
(b)	Are you an affiliate of a broker-dealer?
	Yes □ No □
(c)	If you are an affiliate of a broker-dealer, do you certify that you bought the Registrable Securities in the ordinary course of business, and at the time of the purchase of the Registrable Securities to be resold, you had no agreements or understandings, directly or indirectly, with any person to distribute the Registrable Securities?
	Yes □ No □
Note	e: If no, the Commission's staff has indicated that you should be identified as an underwriter in the Registration Statement.
	ial Ownership of Other Securities of the Company Owned by the Selling Securityholder.
	ept as set forth below in this Item 5, the undersigned is not the beneficial or registered owner of any securities of the Company other than the

Except as set forth below in this Item 5, the undersigned is not the beneficial or registered owner of any securities of the Company other than the Registrable Securities listed above in Item 3.

Type and Amount of Other Securities beneficially owned by the Selling Securityholder:

Except as set forth below, neither the undersigned nor any of its affiliates, officers, direct equity securities of the undersigned) has held any position or office or has had any other affiliates) during the past three years. State any exceptions here: ——————————————————————————————————	
equity securities of the undersigned) has held any position or office or has had any other affiliates) during the past three years. State any exceptions here:	
The undersigned agrees to promptly notify the Company of any inaccuracies or changes in the in	
	nformation provided herein that may occur subsequent to the date
by signing below, the undersigned consents to the disclosure of the information contained herei information in the Registration Statement and the related prospectus. The undersigned understated connection with the preparation or amendment of the Registration Statement and the related pro	ds that such information will be relied upon by the Company in
N WITNESS WHEREOF the undersigned, by authority duly given, has caused this Notice and s duly authorized agent.	Questionnaire to be executed and delivered either in person or by
Dated: Beneficia	l Owner:
Ву:	
Nar Titl	

PLEASE FAX A COPY OF THE COMPLETED AND EXECUTED NOTICE AND QUESTIONNAIRE, AND RETURN THE ORIGINAL BY **OVERNIGHT MAIL, TO:**

> Conner & Winters, LLP 3700 First Place Tower 15 East 5th Street Tulsa, OK 74103-4344 Facsimile No.: (918) 586-8548 Attn: J. Ryan Sacra, Esq.

e-mail: RSacra@cwlaw.com

AMENDMENT NO. ONE TO THE RIGHTS AGREEMENT

This Amendment No. One (this "Amendment") to the Rights Agreement, dated as of November 2, 1999 (the "Agreement"), and effective this 21st day of April 2005, is made and entered into by and between Matrix Service Company, a Delaware Corporation (the "Company") and UMB Bank, N.A., as Rights Agent (the "Rights Agent").

RECITALS

WHEREAS, on April 14, 2005, the Board of Directors of the Company (the "Board") authorized the Securities Purchase Agreement (the "SPA") among the Company and the investors named on the signature pages thereto (each individually, an "Investor"), and

WHEREAS, the Board has determined it to be in the best interests of the Company and its shareholders to amend the Agreement as herein provided in order to permit the transactions contemplated by the SPA.

NOW, THEREFORE, considering the premises and in consideration of the mutual agreements herein contained, the parties hereto hereby agree to amend the Agreement as follows:

- 1. The definition of "Acquiring Person" contained in Section 1 (a) of the Agreement is hereby amended as follows:
 - (a) The following shall be added following the second sentence of such definition:

"Notwithstanding the first sentence of this definition, no Investor shall be deemed to be or to have become an Acquiring Person solely as a result of the purchase of the Investor Securities on the Closing Date unless and until such time thereafter as such Investor shall increase its Beneficial Ownership of Common Shares to an amount in excess of such Investor's then applicable Investor's Percentage (as defined below); provided, however, that such Investor will not be deemed to be an Acquiring Person solely as a result of a reduction in the number of Common Shares outstanding unless and until such time after such reduction as (i) such Investor becomes the Beneficial Owner of additional Common Shares (other than as a result of a stock dividend, stock split or similar transaction effected by the Company in which all holders of Common Shares are treated equally) and after giving effect to such additional Common Shares such Investor is the Beneficial Owner of an amount in excess of such Investor's then applicable Investor's Percentage, or (ii) any other Person who is a Beneficial Owner of

Common Shares becomes an Affiliate or Associate of such Investor and after giving effect thereto such Investor is the Beneficial Owner of an amount in excess of such Investor's then applicable Investor's Percentage."

- (b) The fourth word in the last sentence of such definition is hereby changed from "three" to "four."
- 2. The following definitions are hereby added to Section 1 of the Agreement in their proper alphabetical order and any cross-references to subsections of Section 1 are modified as appropriate:
 - "Closing Date" means the date of closing of the purchase and sale of the Investor Securities pursuant to the Securities Purchase Agreement.
 - "Investor" means each named as an "Investor" in the Securities Purchase Agreement that purchased Investor Securities on the Closing Date.
 - "Investor's Percentage" means, as to each Investor (a) as of the Closing Date, the ratio (expressed as a percentage) that the number of Common Shares issuable upon conversion of the Investor Securities purchased by such Investor on the Closing Date bears to the total number of Common Shares outstanding on the Closing Date (including as outstanding the total number of Common Shares issuable upon conversion of all Investor Securities purchased by all Investors on the Closing Date); and (b) at any time after such Investor becomes the Beneficial Owner of less than 15% of the outstanding Common Shares, 15% or more of the then outstanding Common Shares, then the immediately preceding sentence shall be deemed to be deleted from this Agreement and the Investors shall be subject to the first sentence of this Section 1(a).
 - "Investor Securities" means the securities purchased by Investors on the Closing Date pursuant to the Securities Purchase Agreement.
 - "Securities Purchase Agreement" means the agreement so captioned and dated April 21, 2005, by and between the Company and each Investor, pursuant to which the Company issued and sold the Investor Securities to the Investors.
- 3. Except as expressly modified herein, the Agreement shall remain in full force and effect.

4. This Amendment may be executed in one or more counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The delivery of this Amendment by facsimile shall be effective for all purposes.

IN WITNESS WHEREOF, each party hereto has caused this Amendment to be duly executed on its behalf as of the date first set forth above.

MATRIX SERVICE COMPANY

By: /s/ George L. Austin

Name: George L. Austin Title: Chief Financial Officer

UMB BANK, N.A.

By: /s/ Richard J. Lehrter

Name: Richard J. Lehrter

Title: Community Bank President

AMENDMENT NINE TO CREDIT AGREEMENT

This Amendment Nine to Credit Agreement ("Amendment") is dated as of April 22, 2005 ("Closing Date"), among **MATRIX SERVICE COMPANY**, a Delaware corporation ("Borrower"), the Lenders described below, and **JPMORGAN CHASE BANK**, **N.A.** (successor by merger to Bank One, N.A. (Main Office Chicago)), as a Lender, LC Issuer and as Agent for the Lenders.

RECITALS

- A. Lenders have provided credit facilities to Borrower pursuant to that certain Credit Agreement dated as of March 7, 2003, among Borrower, Agent and the various Lenders party thereto (the "Original Credit Agreement"), as amended by that certain Amendment One to Credit Agreement dated as of May 22, 2003, that certain Amendment Two to Credit Agreement dated as of August 27, 2003, that certain Amendment Three to Credit Agreement dated as of December 19, 2003, that certain Amendment Four to Credit Agreement dated as of March 11, 2004, that certain Amendment Five to Credit Agreement dated as of May 6, 2004, that certain Amendment Six to Credit Agreement dated as of August 5, 2004, that certain Amendment Seven to Credit Agreement dated as of October 6, 2004, that certain Amendment Eight to the Credit Agreement dated as of November 30, 2004, that certain letter agreement dated March 23rd, 2005, and that certain letter agreement dated April 8, 2005 (as amended, the "Credit Agreement").
- B. Pursuant to Amendment Six to Credit Agreement referenced above, \$20,000,000 of the Revolving Commitment was converted to Term Loan B. Pursuant to this Amendment, the Term Loan B Commitment of each of the Lenders other than the Revolver B Lenders will be reduced to zero and the Term Loan B Commitments of the Revolver B Lenders will be restructured as Revolver B Loan Commitments. As a condition of the Agent and the Lenders' agreement to this Amendment, Borrower shall incur the "XYZ Subordinated Obligations" (as defined below) by borrowing from various investors on a subordinated, unsecured basis, with the proceeds of such borrowing to be applied as provided in this Amendment.
- C. Due to and in consideration of the foregoing and other circumstances, Borrower and the Lenders have agreed to make certain modifications to the Credit Agreement as provided in this Amendment.

AGREEMENT

1. <u>Definitions</u>. Capitalized terms used but not defined in this Amendment (including the Recitals) shall have the meanings given to them in the Credit Agreement. All terms defined in the foregoing Recitals are incorporated herein by reference. The term "Loan Documents" is hereby amended to include the Credit Agreement, as amended by this Amendment, all as they may be further amended from time to time with the consent of the Agent and, to the extent required by the Credit Agreement, the Lenders. The term "Agreement", as used in the Credit Agreement, is hereby amended to mean the Credit Agreement, as amended by this Amendment and as it may be further amended from time to time with the consent of the Agent and, to the extent required by the Credit Agreement, the Lenders. The term "Credit Agreement" in all other Loan Documents is hereby amended to mean the Credit Agreement, as amended by this Amendment, as it may be further

amended from time to time with the consent of the Agent and, to the extent required by the Credit Agreement, the Lenders.

The following terms shall hereafter mean the following, for purposes of this Amendment and the Credit Agreement as amended by this Agreement:

"Additional Accrued Margin" means the percentage rate per annum designated as such which is applicable with respect to Loans as set forth in the definition of Applicable Margin.

"Cash Collateralization" means the deposit of unencumbered immediately available cash into an LC Cash Collateral Account.

"LC Cash Collateral Account" means a blocked deposit account or accounts to be established and maintained at the office of the Agent (or an affiliate thereof) in the name of the Agent as collateral security for outstanding LC Obligations. Any such account and deposits shall be and are hereby designated as Pledged Deposits, as such term is defined in the Security Agreement.

"Revolver B Borrowing Base" means \$10,000,000.00 (which represents a portion of the total amount the Revolver B Lenders estimate is likely to be collected by Borrower in connection with the Large Disputed Accounts), provided that such amount shall be reduced immediately upon receipt by Borrower of written notice from the Agent at any time and from time to time if the Revolver B Lenders determine, in their sole judgment, that the amount Borrower is likely to ultimately collect from the Large Disputed Accounts has declined from the amount last estimated by the Revolver B Lenders. Notwithstanding anything to the contrary in Section 8.2, the Revolver B Lenders may, by unanimous action of the Revolver B Lenders, reduce the Revolver B Borrowing Base as provided above or otherwise unanimously agree to increase or decrease the Revolver B Borrowing Base in their sole discretion without the consent or agreement of any other Lender.

"Revolver B Lenders" means JPMorgan Chase Bank, N.A. and Wells Fargo Bank, NA, and their respective successors and assigns in regard to the Revolving B Loan.

"Revolving B Loan" means, with respect to a Revolver B Lender, such Lender's loan made pursuant to its commitment to lend set forth in <u>Section 2.1.5.</u> (or any conversion or continuation thereof).

"Revolving B Loan Commitment" means, for each Revolver B Lender, the obligation of such Lender to make Revolving B Loans to Borrower in an aggregate amount not exceeding the amount specified in <u>Section 2.1.5.</u> with respect to such Revolver B Lender, as such amount may be modified from time to time pursuant to the terms hereof.

"Revolver B Termination Date" means October 31, 2005 or any earlier date on which the Revolving B Loan Commitment is reduced to zero or otherwise terminated pursuant to the terms hereof.

"Securities Purchase Agreement" means that certain Securities Purchase Agreement dated as of the Closing Date among the Borrower and the investors identified on the signature pages thereto.

"Subordination Agreement" means that certain Subordination Agreement dated as of the Closing Date, between the Subordinated Creditors party thereto, the Agent, the Borrower and the

other various obligors listed on the signature pages of such agreement, as further amended, supplemented, restated or otherwise modified from time to time in accordance with its terms and the terms of the Credit Agreement.

"XYZ Subordinated Loan Documents" means the Securities Purchase Agreement, the Notes (as defined in the Securities Purchase Agreement), the Registration Rights Agreement (as defined in the Securities Purchase Agreement), and any other documents or agreements executed in connection with the transactions contemplated under the Securities Purchase Agreement.

"XYZ Subordinated Loans" means the loans made to the Borrower pursuant to the XYZ Subordinated Loan Documents, which loans are subordinated to the Obligations in accordance with the terms of the Subordination Agreement.

"XYZ Subordinated Obligations" has the meaning assigned to such term in the Subordination Agreement.

2. Amendments to Credit Agreement.

- 2.1 The following defined terms are hereby amended by deleting the current definition and replacing it with the following:
- "Aggregate Commitment" means the aggregate of the Revolving Credit Commitments, the Revolving B Loan Commitments and the Term Loan Commitments of all the Lenders, as reduced from time to time pursuant to the terms hereof.

"Applicable Margin" means the percentage rate per annum which is applicable at such time with respect to the listed Loans and Advances as set forth below:

	Applicable	Applicable Margin		Additional Accrued Margin	
Period	Revolver & Term Loans	Revolver B Loans	Revolver & Term Loans	Revolver B Loans	
April 22, 2005 - April 30, 2005	1.00%	0%	1.00%	0%	
May 1, 2005 - May 31, 2005	1.00%	0%	1.50%	0%	
June 1, 2005 - June 30, 2005	1.00%	0%	2.00%	0%	
July 1, 2005 - July 31, 2005	1.00%	0%	2.50%	0%	
August 1, 2005 - August 31, 2005	1.00%	0%	3.00%	0%	
September 1, 2005 - September 30, 2005	1.00%	0%	3.50%	0%	
October 1, 2005 - October 31, 2005	1.00%	0%	4.00%	0%	
November 1, 2005 - November 30, 2005	1.00%	0%	4.50%	0%	
December, 2005 and after	1.00%	0%	5.00%	0%	

[&]quot;Borrowing Base" means eighty percent (80%) of Consolidated Eligible Accounts Receivable.

"Borrowing Base Determination Date" means (i) the last day of each calendar month, and (ii) Friday of each calendar week.

"Consolidated EBITDA" means Consolidated Net Income plus, to the extent deducted in determining Consolidated Net Income, (i) Consolidated Interest Expense, (ii) expense for taxes paid or accrued, (iii) depreciation, amortization, other non-cash charges, (iv) losses on sale of fixed assets, and (v) extraordinary losses incurred other than in the ordinary course of business, minus, to the extent included in Consolidated Net Income, (i) gains on sales of fixed assets, (ii) extraordinary gains realized other than in the ordinary course of business, and (iii) income tax benefits, all calculated for the Borrower and its Subsidiaries on a consolidated basis for any period.

"Loan" means a Term Loan, a Revolving Loan, a Revolving B Loan or a Swing Line Loan.

"Non-Earmarked Cash" shall mean (i) the aggregate of all Cash Equivalent Investments of Borrower and all Subsidiaries excluding (ii) (A) funds required by Sections 2.7.2. (i), (ii), (iii), (v), (vi), or (viii) to be paid as mandatory prepayments under such Sections, provided that such amounts are paid as required by such Sections immediately upon being received by Borrower or the applicable Subsidiary in collected funds, and (B) funds kept on hand in the ordinary course of business of Borrower and its Subsidiaries for payroll funding.

"Term Loan B Commitment" means, as of the Closing Date of Amendment Nine, as to each Lender, zero.

2.2 Section 2.1.1. is hereby amended by deleting the last sentence thereof and adding the following language at the end of the section:

From and including April 22, 2005, all Revolving Loans, including those outstanding on such date and those made thereafter, will accrue interest at the Alternate Base Rate plus the Applicable Margin plus the Additional Accrued Margin. The portion of the accrued interest on all Revolving Loans attributable to the Alternate Base Rate plus the Applicable Margin shall be due and payable on each Payment Date. The portion of the accrued interest on all Revolving Loans attributable to the Additional Accrued Margin shall not be payable on each Payment Date, but shall be payable on the earlier of (i) voluntary prepayment of all Obligations associated with the Revolving Loans, (ii) as required by Section 2.7.2. or (iii) the Facility Termination Date.

2.3 Section 2.1.2. is hereby amended by adding the following language at the end of the section:

From and including April 22, 2005, the Term Loan will accrue interest at the Alternate Base Rate plus the Applicable Margin plus the Additional Accrued Margin. The portion of the accrued interest on the Term Loan attributable to the Alternate Base Rate plus the Applicable Margin shall be due and payable on each Payment Date. The portion of the interest on the Term Loan attributable to the Additional Accrued Margin shall not be payable on each Payment Date, but shall be payable on the earlier of (i) the voluntary prepayment in full of all Obligations associated with the Term Loan, (ii) as required by Section 2.7.2., or (iii) the Facility Termination Date and the Term Loan remains outstanding after such date, the portion of interest

attributable to the Additional Accrued Margin for subsequent periods shall be payable on the earlier of (i) the voluntary prepayment in full of all Obligations associated with the Term Loan, (ii) as required by Section 2.7.2., or (iii) the Term Loan Maturity Date.

2.4 A new Section 2.1.5. is hereby added to the Credit Agreement and shall read as follows:

2.1.5. Revolving Loan B Commitment. The sum of \$10,000,000 of the Term Loan B Commitment is and hereby shall be restructured and become the Revolving B Loan Commitments, as further described in this Section 2.1.5., and the remainder of the Term Loan B Commitment is hereby permanently reduced to \$0. From and including the date hereof but prior to the Revolver B Termination Date, the Revolver B Lenders agree to make Revolving B Loans to the Borrower from time to time, in amounts not to exceed in the aggregate at any one time outstanding the lesser of (i) \$10,000,000 less an amount equal to the value of all outstanding checks as reported on Borrower's most recent Borrowing Base Certificate or (ii) the Revolver B Borrowing Base less an amount equal to the value of all outstanding checks as reported on Borrower's most recent Borrowing Base Certificate. Upon any collection of funds with respect to a Large Disputed Account, receipt of Net Cash Proceeds of any sale of common stock, preferred stock, warrants or other equity by Borrower or any of its Subsidiaries, or receipt of proceeds of any asset sale by Borrower or any of its Subsidiaries, in each case, the \$10,000,000.00 amount referred to in (i) above will be automatically and permanently reduced by an amount equal to the amount of cash received by Borrower or any of its Subsidiaries, without any further action by the Agent, the Lenders or the Borrower.

The Borrower shall not be entitled to and no Revolver B Lender shall be obligated to make any Revolver B Loans until and unless the Revolver Commitments have been fully utilized through the issuance of the maximum aggregate amount of Revolving Loans and Facility LC's permitted by <u>Section 2.1.1.</u> No Revolver B Lender shall be required to make any Revolving B Loans in excess of its pro rata share of all such Loans, as determined by dividing such Revolving B Lender's Revolving B Loan Commitment by the aggregate amount of all Revolving B Loan Commitments.

Not later than 11:00 a.m. Tulsa, Oklahoma time on the date specified for each borrowing of Revolver B Loans, each Revolver B Lender shall make available the amount of the Revolving B Loan to be made by it on such date to the Agent, to an account which the Agent shall specify, in immediately available funds, for the account of the Borrower. The amounts so received by the Agent shall, subject to the terms and conditions of this Agreement, be made available to the Borrower by depositing the same, in immediately available funds, in an account of the Borrower, designated by the Borrower from time to time, written notice of the location of which shall be given to the Agent together with the notice made pursuant to Section 2.8. As of April 22, 2005, the Revolving B Loan Commitments of each of the Lenders are as follows:

JPMorgan Chase Bank	\$5,000,000.00
Wells Fargo Bank, NA	\$5,000,000.00
International Bank of Commerce	\$ 0.00
UMB Bank, N.A.	\$ 0.00
Wachovia Bank, National Association	\$ 0.00

Interest on Revolving B Loans shall accrue at a rate equal to the Alternate Base Rate plus the Applicable Margin and the Additional Accrued Margin.

- 2.5 <u>Section 2.2.</u> is hereby deleted and replaced with the following:
 - 2.2. Required Payments; Termination.
- (a) The Aggregate Outstanding Credit Exposure and all other unpaid Obligations, other than the Term Loan and the Revolver B Loans, shall be paid in full by the Borrower on the Facility Termination Date.
- (b) The Term Loan shall be payable as follows: (i) interest shall be due and payable as provided in <u>Section 2.1.2.</u>, and (ii) principal shall be payable in nineteen (19) consecutive quarterly installments of \$1,160,714.29, on the last day of each fiscal quarter ending May, August, November and February, commencing with the fiscal quarter ending August 31, 2003, and the last installment due on the Term Loan Maturity Date equal to the remaining balance.
- (c) The Revolver B Loans shall be payable as follows: (i) interest at the Alternate Base Rate plus the Applicable Margin shall be due and payable on each Payment Date, and (ii) all unpaid Obligations attributable to the Revolver B Loans shall be paid in full by the Borrower on the Revolver B Termination Date.
- 2.6 Section 2.7. is hereby deleted and replaced with the following:
 - 2.7. Prepayments.
- 2.7.1. <u>Optional Prepayments</u>. Loans bearing interest based on the Alternate Base Rate (other than Swing Line Loans) may be prepaid at any time without penalty or premium on the same Business Day prior written notice is delivered by noon in a minimum amount of \$500,000.00.
 - 2.7.2. Mandatory Prepayments. In addition to any scheduled installments due on the Loans, the following mandatory prepayments shall be made:
 - (i) <u>Sale of Assets</u>: Upon the sale, transfer or other disposition of any asset of the Borrower or any of its Subsidiaries (other than the sale of inventory in the ordinary course of business and the sale of up to \$250,000.00 of other assets per calendar year) which is permitted by the terms of the Loan Documents, the Borrower shall immediately make a mandatory prepayment of the Obligations in an amount equal to one hundred percent (100%) of the net proceeds realized from such sale, transfer or other disposition, and such payment shall be applied in the following order:
 - (a) to the Revolver B Loans (first all Obligations other than interest or principal, then interest at the Alternate Base Rate plus the Applicable Margin, then interest at the Additional Accrued Margin, then principal), and after payment thereof in full,

- (b) to the Term Loan (first all Obligations other than interest or principal, then interest at the Alternate Base Rate plus the Applicable Margin, then interest at the Additional Accrued Margin, then principal in inverse order of maturity), and after payment thereof in full,
- (c) to the Revolving Loans (first all Obligations other than interest or principal, then interest at the Alternate Base Rate plus the Applicable Margin, then interest at the Additional Accrued Margin, and then principal), and after payment thereof in full,
 - (d) to the Cash Collateralization of all LC Obligations.
- (ii) <u>Sale of Stock</u>: Upon the sale of any common stock, preferred stock, warrant or other equity (other than the exercise of stock options by employees, officers and directors) Borrower shall immediately make a mandatory prepayment of the Obligations in an amount equal to one hundred percent (100%) of the Net Cash Proceeds from such sale or issuance, and such payment shall be applied in the following order:
 - (a) to the Revolver B Loans (first all Obligations other than interest or principal, then interest at the Alternate Base Rate plus the Applicable Margin, then interest at the Additional Accrued Margin, then principal), and after payment thereof in full,
 - (b) to the Obligations under the Term Loan or the Revolving Loans, or in part of both, as the Borrower may choose, provided that all prepayments on the Term Loan shall be applied first to principal installments thereunder in the inverse maturity thereof, then to any other Obligations under the Term Loan, and after payment in full of all Obligations under the Term Loan and the Revolving Loans,
 - (c) to the Cash Collateralization of all LC Obligations.
- (iii) <u>Issuance of Subordinated Indebtedness</u>: Upon the receipt of proceeds from the issuance of any permitted Subordinated Indebtedness, other than the XYZ Subordinated Indebtedness, Borrower shall immediately make a mandatory prepayment of the Obligations in an amount equal to one hundred percent (100%) of the Net Cash Proceeds from such sale or issuance, and such payment shall be applied in the following order:
 - (a) to the Revolver B Loans (first all Obligations other than interest or principal, then interest at the Alternate Base Rate plus the Applicable Margin, then interest at the Additional Accrued Margin, then principal), and after payment thereof in full,

- (b) to the Obligations under the Term Loan or the Revolving Loans, or in part of both, as the Borrower may choose, provided that all prepayments on the Term Loan shall be applied first to principal installments thereunder in the inverse maturity thereof, then to any other Obligations under the Term Loan, and after payment in full of all Obligations under the Term Loan and the Revolving Loans,
 - (c) to the Cash Collateralization of all LC Obligations.
- (iv) Excess Cash Flow: On or before each date on which the Borrower's annual audited financial statements are required to be delivered pursuant to this Agreement, commencing with the fiscal year ending May 31, 2006, the Borrower shall make a mandatory prepayment in an amount equal to fifty percent (50%) of the Excess Cash Flow, if positive, for the most recently ended fiscal year. For purposes hereof, the term Excess Cash Flow, as to the applicable period, means Consolidated EBITDA, less (i) Consolidated Interest Expense, (ii) taxes paid, (iii) principal payments on the Term Loan, (iv) Capital Expenditures, and (v) dividends and distributions permitted in this Agreement. Such payment shall be applied in the following order:
 - (a) to the Revolver B Loans (first all Obligations other than interest or principal, then interest at the Alternate Base Rate plus the Applicable Margin, then interest at the Additional Accrued Margin, then principal), and after payment thereof in full,
 - (b) to the Term Loan (first all Obligations other than interest or principal, then interest at the Alternate Base Rate plus the Applicable Margin, then interest at the Additional Accrued Margin, then principal in inverse order of maturity), and after payment thereof in full,
 - (c) to the Revolving Loans (first all Obligations other than interest or principal, then interest at the Alternate Base Rate plus the Applicable Margin, then interest at the Additional Accrued Margin, and then principal), and after payment thereof in full,
 - (d) to the Cash Collateralization of all LC Obligations.
- (v) <u>Borrowing Base Deficiency</u>. If the aggregate principal amount of the outstanding Revolving Loans and LC Obligations as of any date exceeds the Borrowing Base, the Borrower shall immediately make a mandatory principal payment on the Revolving Loans necessary to establish compliance. Likewise, if the aggregate principal amount of outstanding Revolver B Loans as of any date exceeds the Revolver B Borrowing Base, the Borrower shall immediately make a mandatory principal payment on the Revolving B Loans necessary to establish compliance.
- (vi) <u>Collections of Large Disputed Accounts</u>. Upon the receipt of any proceeds from the collection of any Large Disputed Accounts, Borrower shall immediately make a mandatory prepayment of the Obligations in an amount equal to

one hundred percent (100%) of such proceeds. Such payment shall be applied in the following order:

- (a) to the Revolver B Loans (first all Obligations other than interest or principal, then interest at the Alternate Base Rate plus the Applicable Margin, then interest at the Additional Accrued Margin, then principal), and after payment thereof in full,
- (b) to the Obligations under the Term Loan or the Revolving Loans, or in part of both, as the Borrower may choose, provided that all prepayments on the Term Loan shall be applied first to principal installments thereunder in the inverse maturity thereof, then to any other Obligations under the Term Loan, and after payment in full of all Obligations under the Term Loan and the Revolving Loans,
 - (c) to the Cash Collateralization of all LC Obligations.
- (vii) Excess Cash. If at any time Borrower and the Subsidiaries own or hold Non-Earmarked Cash such that there is any Excess Cash Amount, Borrower shall immediately make a mandatory prepayment of the Obligations in an amount equal to the Excess Cash Amount, and such payment shall be in the following order:
 - (a) to the Revolver B Loans (first all Obligations other than interest or principal, then interest at the Alternate Base Rate plus the Applicable Margin, then interest at the Additional Accrued Margin,, then principal), and after payment thereof in full,
 - (b) to the Obligations under the Term Loan or the Revolving Loans, or in part of both, as the Borrower may choose, provided that all prepayments on the Term Loan shall be applied first to principal installments thereunder in the inverse maturity thereof, then to any other Obligations under the Term Loan, and after payment in full of all Obligations under the Term Loan and the Revolving Loans,
 - (c) to the Cash Collateralization of all LC Obligations.
- (viii) Accounts Receivable. If no Default or Unmatured Default has occurred and is continuing, immediately upon collection of any Account (other than Large Disputed Accounts) Borrower shall immediately make a mandatory principal payment in an amount equal to the full amount collected to be applied to the Revolving Loans, if such payment is required in accordance with the first sentence of Section 2.7.2.(v) to establish compliance with the Borrowing Base, and otherwise to the Revolver B Loans. If a Default or Unmatured Default has occurred and is continuing or if the Borrower is in compliance with the Borrowing Base, immediately upon collection of any Account (other than Large Disputed Accounts) Borrower shall immediately make a mandatory payment in an amount equal to the full amount collected to be applied to the Revolver B Loans, first to all related Obligations other than interest or principal, then interest, then principal.

- 2.7.3 <u>Payment Priorities</u>. The Revolver B Loans and all Obligations related thereto shall be entitled to repayment prior to all other Indebtedness of the Borrower and its Subsidiaries; provided that, as long as no Default or Unmatured Default has occurred and is continuing, Borrower and its Subsidiaries shall be permitted to (i) make payments on each Payment Date in amounts sufficient to pay accrued interest then due and payable with respect to the Revolving Loans and the Term Loan contemporaneously with payment in full of accrued interest then due and payable with respect to the Revolving B Loans, and (ii) make quarterly installment payments with respect to the Term Loan as required pursuant to Section <u>2.2.</u>, in each case, without first repaying all Obligations related to the Revolver B Loans.
- 2.7 Section 2.8. is hereby amended by deleting the word "and" at the end of subsection (iii), and adding the following at the end of subsection (iv):
 - (v) a representation of the current Consolidated Eligible Accounts Receivable, and
- (vi) a calculation showing such Consolidated Eligible Accounts Receivable minus all accounts payable of Borrower and each of its Subsidiaries that are more than sixty (60) days past due, and multiplying the difference times eighty percent (80%).
- 2.8 <u>Section 6.1(iii)</u> is hereby deleted and replaced with the following:
- (iii) Together with the financial statements required under Sections 6.1(i) and (ii), and within 30 days of the end of each calendar month, a compliance certificate in substantially the form of Exhibit B signed by its chief financial officer showing the calculations necessary to determine compliance with this Agreement, including without limitation a detailed Consolidated EBITDA calculation and a detailed listing of all non-recurring extraordinary professional fees and restructuring expenses applicable to the covenant contained in Section 6.27.1., and stating that no Default or Unmatured Default exists, or if any Default or Unmatured Default exists, stating the nature and status thereof.
- 2.9 Section 6.1.(xi) is hereby amended so that the Borrower shall be obligated to deliver an appropriately completed and certified Borrowing Base Certificate, with a Borrowing Base Determination Date as of the last day of the previous week, no later than the end of the Business Day on Thursday of each week, and, in addition to the information currently required to be included in each such certificate, Borrower will also disclose and represent the accuracy of the number and dollar amount of all checks issued by Borrower or any of its Subsidiaries that remain outstanding. Borrower may deliver more frequent Borrowing Base Certificates at its discretion.
 - 2.10 Section 6.1.(xvi) is hereby deleted and replaced with the following:
 - (xvi) On a weekly basis on or before the end of the Business Day on Wednesday of each week, (a) accounts payable reports, including a discussion of any liens filed and the resolution thereof, in form and content satisfactory to Agent, and (b) a discussion of any arrangements for direct payment of vendors by customers of the Borrower or any of its Subsidiaries and the effect of such arrangements on Accounts.

2.11 Section 6.1. is hereby amended by adding the following subsections immediately after subsection (xvi):

(xvii) No later than May 15, 2005 an appraisal of the interests of Borrower and all of its Subsidiaries in all real estate and equipment owned, leased or otherwise under the control of Borrower or any such Subsidiary, in form and substance satisfactory to the Agent and from an appraiser or appraisers satisfactory to the Agent.

(xviii) No later than five days after the end of each calendar month, a report on Borrower's efforts to obtain financing necessary to retire all of the Obligations.

- (xix) On a weekly basis on or before the end of the Business Day on Wednesday of each week, a written thirteen-week cash flow projection covering the immediately subsequent thirteen (13) weeks, in a form acceptable to Agent.
- (xx) (i) Promptly after delivery thereof, copies of all notices sent to Borrower or any holder of the XYZ Subordinated Indebtedness regarding any default under the terms of the XYZ Subordinated Loan Documents, (ii) promptly after receipt thereof, copies of all notices sent by any party to the XYZ Subordinated Loan Documents regarding any default under the terms of the XYZ Subordinated Loan Documents, and (iii) promptly after execution thereof, copies of all amendments, modifications, supplements, written waivers or other instruments modifying the XYZ Subordinated Loan Documents, provided that all such amendments, modifications, supplements, written waivers or other instruments shall remain subject to the consent of the Lenders to the extent provided in the Credit Agreement.

2.12 Section 6.15 (ii) is hereby deleted and replaced with the following:

(ii) Liens imposed by law, such as carriers', warehousemen's and mechanics' liens and other similar liens arising in the ordinary course of business which secure payment of obligations not more than 60 days past due or which are being contested in good faith by appropriate proceedings and for which adequate reserves shall have been set aside on its books; provided that the obligations incurred after April 22, 2005 secured by such liens shall not exceed \$5,000,000.00.

2.13 Sections 6.27.1., 6.27.2, 6.27.3. and 6.27.4 are hereby deleted and replaced with the following:

6.27.1. <u>Minimum EBITDA</u>. The Borrower will not permit Consolidated EBITDA for each test period set forth below, as determined on the designated test date for each such period, to be less than the minimum amount set forth opposite such period:

TEST PERIODS			M	INIMUM EBITDA	TEST DATE	
March 1, 2005	through	April 30, 2005	\$	1,200,000.00	May 31, 2005	
March 1, 2005	through	May 31, 2005	\$	3,152,930.00	June 30, 2005	
June 1, 2005	through	June 30, 2005	\$	911,193.00	July 31, 2005	
June 1, 2005	through	July 31, 2005	\$	1,822,386.00	August 31, 2005	
June 1, 2005	through	August 31, 2005	\$	3,644,771.00	September 30, 2005	
June 1, 2005	through	September 30, 2005	\$	4,639,881.00	October 31, 2005	
June 1, 2005	through	October 31, 2005	\$	5,634,991.00	November 30, 2005	
June 1, 2005	through	November 30, 2005	\$	7,625,210.00	December 31, 2005	
June 1, 2005	through	December 31, 2005	\$	8,688,507.00	January 31, 2006	
June 1, 2005	through	January 31, 2006	\$	9,751,804.00	February 28, 2006	
June 1, 2005	through	February 28, 2006	\$	11.878.399.00	March 31, 2006	

For purposes of determining compliance with this Section <u>6.27.1.</u>, Consolidated EBITDA shall be deemed increased by an amount equal to all non-recurring extraordinary professional fees and restructuring charges incurred by Borrower during the applicable test period; provided that no such fees and charges incurred after November 30, 2005 may be included and the maximum aggregate amount of such fees and charges paid in cash as to all test periods that may be so included shall not exceed \$5,000,000.00; provided further that such fees and charges shall only be deemed to increase Consolidated EBITDA to the extent they were deducted from Consolidated Net Income.

In addition, but without duplication, for purposes of determining compliance with this Section <u>6.27.1.</u> solely for the test period beginning March 1, 2005 and ending May 31, 2005, Consolidated EBITDA shall be deemed increased by the lesser of (i) \$1,200,000.00, or (ii)(a) twenty-eight percent (28%) times (b) the gross proceeds of the XYZ Subordinated Loans, minus all prepaid interest thereon, fees paid by Borrower in connection with the closing of the XYZ Subordinated Loans, and \$20,000,000.00.

6.27.2. Fixed Charge Coverage Ratio. The Borrower will not permit the ratio, determined as of the end of each of its fiscal quarters, of (i) Consolidated EBITDA for the then most recently ended fiscal quarter, <u>minus</u> cash dividends and cash distributions made or paid during the same period, <u>minus</u> cash taxes paid during the same period, to (ii) scheduled current maturities of long-term debt (determined according to generally accepted accounting practices) for the most recently ended fiscal quarter, <u>plus</u> Consolidated Interest Expense (excluding non-cash interest, interest pre-paid upon the closing of the XYZ Subordinated Loans, and fees paid in connection with Amendment Nine to the Credit Agreement) for the most recently ended fiscal quarter, <u>plus</u> current maturities on Capitalized Leases for the most recently ended fiscal quarter, <u>plus</u> Capital Expenditures paid in the most recently ended fiscal quarter to be less than:

1.00 for the fiscal quarter ending May 31, 2005

1.15 for the fiscal quarter ending August 31, 2005

1.25 for the fiscal quarter ending November 30, 2005

1.33 for the fiscal quarter ending February 28, 2006

For purposes of determining compliance with this Section <u>6.27.2.</u>, Consolidated EBITDA shall be deemed increased by an amount equal to all non-recurring extraordinary professional fees and restructuring charges incurred by Borrower during the applicable fiscal quarter; provided that no such fees and charges incurred after November 30, 2005 may be included and the maximum aggregate amount of such fees and charges paid in cash as to all fiscal quarters that may be so included shall not exceed \$5,000,000.00; provided further that such fees and charges shall only be deemed to increase Consolidated EBITDA to the extent they were deducted from Consolidated Net Income.

In addition, but without duplication, for purposes of determining compliance with this Section <u>6.27.2.</u> solely for the fiscal quarter beginning March 1, 2005 and ending May 31, 2005, Consolidated EBITDA shall be deemed increased by the lesser of (i) \$1,200,000.00, or (ii) twenty-eight percent (28%) of an amount equal to the gross proceeds of the XYZ Subordinated Loans, minus (a) all prepaid interest thereon, (b) fees paid by Borrower in connection with the closing of the XYZ Subordinated Loans, and (c) \$20,000,000.00.

- 2.14 A new Section 6.28, shall be added to the Credit Agreement immediately after Section 6.27.5, as follows:
- 6.28 <u>Monthly Meetings</u>. No later than the fifteenth (15th) day of each calendar month, Borrower will arrange for an in-person meeting with representatives of each Lender. Borrower's chief executive officer, chief financial officer, the Borrower's restructuring advisors, and each other officer and employee with responsibility for the restructuring and refinancing activities of the Borrower shall attend each such meeting.
- 2.15 New Sections 7.23., 7.24., 7.25. and 7.26. shall be added to the Credit Agreement immediately after Section 7.22. and shall read as follows:
- 7.23. Any default shall occur under the XYZ Subordinated Loan Documents (including but not limited to any "Buy-In" as described in the Subordinated Notes), or any event specified in any note, agreement, indenture or other document evidencing or relating to any such Indebtedness shall occur if the effect of such event is to cause, or (with the giving of any notice or the lapse of time or both) to permit the holder or holders of such Indebtedness (or a trustee or agent on behalf of such holder or holders) to cause, such Indebtedness to become due, or to be prepaid in full (whether by redemption, purchase, offer to purchase or otherwise) prior to its stated maturity.
- 7.24 The Subordination Agreement shall terminate or become unenforceable in whole or in part other than in accordance with its terms in the ordinary course.
 - 7.25. Any default by any of the "Obligors" or by the "Subordinated Creditor" under the Subordination Agreement.
- 7.26 Failure by the Borrower to deliver a fully completed and certified perfection certificate, in form and substance satisfactory to the Agent, on or before the end of the Business Day on April 29, 2005.
- 2.16 "Exhibit H" is hereby deleted and replaced with the new "Exhibit H" attached hereto.

3. Waivers.

3.1 The "Specified Default" as defined in and contained in that certain letter agreement between the Borrower, the other Loan Parties and the Lenders dated April 8, 2005 (the "Waiver Letter") which was waived through June 15, 2005 by such letter agreement, is hereby permanently waived.

- 3.2 <u>Section 6.11.</u> of the Credit Agreement is hereby waived to the limited extent necessary to permit the incurrence of the XYZ Subordinated Indebtedness; provided that such waiver shall be expressly conditioned upon the satisfaction of the following:
 - 3.2.1 The Agent shall have received copies of the XYZ Subordinated Loan Documents in form and substance satisfactory to the Agent, and such XYZ Subordinated Loan Documents shall be certified by the Borrower as true, correct and complete.
 - 3.2.2 The conditions to effectiveness under the XYZ Subordinated Loan Agreement shall have been fulfilled or waived in writing by lenders thereunder.
 - 3.2.3 The Borrower shall have received XYZ Subordinated Loans pursuant to the XYZ Subordinated Loan Documents with net cash proceeds to the Borrower in an aggregate amount that is not less than an amount sufficient to repay all Obligations arising from or related to Term Loan B, including without limitation the principal amount thereof and all unpaid accrued interest thereon, plus \$4,000,000.00, and the Agent shall have received (i) payment in unencumbered immediately available funds sufficient to repay all of the Obligations arising from or related to Term Loan B and (ii) payment of the additional \$4,000,000.00, which shall be applied to the outstanding Revolving Loans (first to all Obligations other than interest or principal, then interest, then principal).
 - 3.2.4 The Agent shall have received originals of the Subordination Agreement, duly executed and delivered by all the parties thereto, including the Agent.
- 3.3 Nothing herein shall be deemed or construed to waive (i) any other breach of the Credit Agreement or of the applicable Security Agreements, (ii) any obligations arising under any of the Loan Documents of Borrower and the Subsidiaries to take, and Borrower's and such Subsidiaries' authorization of Agent to take, such actions as may be requested by Agent in order to obtain or maintain a perfected first priority security interest in, and if applicable Control of, the Collateral, or (iii) any other provision of the Credit Agreement or the other Loan Documents, including without limitation the fees required to be paid pursuant to the Waiver Letter, and nothing herein shall be deemed or construed as any requirement that the Agent or Lenders grant any similar or dissimilar waiver hereafter.
- 4. <u>Conditions of this Amendment and Waiver</u>. The amendments contained herein, the obligations of Agent and Lenders herein and the waivers contained herein shall only become effective upon the satisfaction of the following conditions precedent:
- 4.1 Agent's receipt of the following, each of which shall be originals or facsimiles (followed promptly by originals) unless otherwise specified, each properly executed, each dated the same date as this Amendment (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance satisfactory to Agent and its legal counsel:
 - 4.1.1 executed counterparts of this Amendment and all other documents and instruments requested by Agent, sufficient in number for distribution to each Lender and Borrower;
 - 4.1.2 such certificates of resolutions or other action, incumbency certificates and/or other certificates of Responsible Officers of each Loan Party as Lender may require

evidencing the identity, authority and capacity of each Responsible Officer thereof authorized to act as a Responsible Officer in connection with this Amendment and the other Loan Documents to which such Loan Party is a party;

- 4.1.3 executed amended and restated promissory notes from Borrower in favor of each of the Lenders in a form acceptable to Agent and the Lenders;
- 4.1.4 such documents and certificates as Agent may reasonably require to evidence that each Loan Party is duly organized or formed and that Borrower is, validly existing, in good standing and qualified to engage in business in each jurisdiction where its ownership, lease or operation of properties or the conduct of its business requires such qualification, except to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect;
- 4.1.5 executed originals of the Amendment and Supplement to Pledge and Security Agreement and Ratification of Guaranty Agreement, in the forms set forth on Schedules "A" and "B", respectively, attached hereto, for each party thereto; and
 - 4.1.6 a legal opinion of Borrower's counsel in form and substance acceptable to the Agent.
- 4.2 Satisfaction of all conditions set forth in Section 3.2 of this Amendment.
- 4.3 Borrower shall pay a commitment fee of \$100,000.00, in unencumbered immediately available funds, to the Agent for distribution to the Revolver B Lenders on a pro rata basis.
- 4.4 Borrower shall pay a restructure fee of \$200,000.00, in unencumbered immediately available funds, to Agent for distribution to the Lenders in accordance with their Pro Rata Share.
- 4.5 Agent's receipt of such other assurances, certificates, documents, consents, evidence of perfection of all Liens securing the Obligations and opinions as Agent reasonably may require.
- 4.6 Unless waived by Agent, Borrower shall have paid all fees, expenses and disbursements of any law firm or other external counsel for Agent and of Capstone Corporate Recovery, LLC, to the extent invoiced prior to the date hereof, plus such additional amounts of such fees, expenses and disbursements as shall constitute its reasonable estimate thereof incurred or to be incurred by it through the closing proceedings as to this Amendment (provided that such estimate shall not thereafter preclude a final settling of accounts between Borrower and Agent).
- 5. <u>Representations and Warranties</u>. Borrower certifies, covenants, represents and warrants to and with Agent and Lenders that, after giving affect to the amendments to the Credit Agreement contemplated by this Amendment: (i) no Default or Unmatured Default exists; and (iv) the representations and warranties contained in <u>Article V</u> of the Credit Agreement are true and correct as of the date hereof, except to the extent such representation or warranty is stated to relate solely to an earlier date, in which case such representation or warranty shall have been true and correct on and as of such earlier date.
- 6. <u>Defaults Unaffected</u>. Except as may be expressly set forth herein, nothing contained in this Amendment shall prejudice, act as, or be deemed to be a waiver of any Default or Unmatured Default

or any right or remedy available to Agent or any Lender by reason of the occurrence or existence of any fact, circumstance or event constituting a Default or Unmatured Default.

7. Releases. Borrower, for itself and on behalf of all its predecessors, successors, assigns, agents, employees, representatives, officers, directors, general partners, limited partners, joint shareholders, beneficiaries, trustees, administrators, subsidiaries, affiliates, employees, servants and attorneys (collectively the "Releasing Parties"), hereby releases and forever discharges Agent and each Lender and their respective successors, assigns, partners, directors, officers, agents, attorneys, and employees from any and all claims, demands, cross-actions, controversies, causes of action, damages, rights, liabilities and obligations, at law or in equity whatsoever, known or unknown, whether past, present or future, now held, owned or possessed by the Releasing Parties, or any of them, or which the Releasing Parties or any of them may, as a result of any actions or inactions occurring on or prior to the Closing Date, hereafter hold or claim to hold under common law or statutory right, arising, directly or indirectly out of the Loan or any of the Loan Documents or any of the documents, instruments or any other transactions relating thereto or the transactions contemplated thereby.

Borrower understands and agrees that this is a full, final and complete release and agrees that this release may be pleaded as an absolute and final bar to any or all suit or suits pending or which may hereafter be filed or prosecuted by any of the Releasing Parties, or anyone claiming by, through or under any of the Releasing Parties, in respect of any of the matters released hereby, and that no recovery on account of the matters described herein may hereafter be had from anyone whomsoever, and that the consideration given for this release is no admission of liability.

- 8. <u>Lender Credit Decision</u>. Each Lender acknowledges that it has, independently and without reliance upon the Agent or any other Lender and based on the financial statements prepared by the Borrower and such other documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Amendment.
- 9. <u>USA PATRIOT Act Notice</u>. Agent hereby notifies Borrower that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001) (the "Patriot Act"), it and the Lenders are or may be required to obtain, verify and record information that identifies the Borrower and the other Loan Parties, which information includes the name and address of the Borrower and the other Loan Parties and other information that will allow Agent and Lenders to identify the Borrower and the other Loan Parties in accordance with the Patriot Act.
- 10. <u>Reimbursement</u>. Borrower agrees to reimburse Agent for any costs, expenses, and fees (including reasonable attorney fees) incurred in connection with the preparation of this Amendment.
- 11. <u>Governing Law</u>. This Amendment shall be construed and enforced in accordance with, and the rights of the parties shall be governed by, the laws of the State of Oklahoma.

12. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts, all of which taken together shall constitute one agreement, and any of the parties hereto may execute this Amendment by signing any such counterpart. This Amendment shall be effective when it has been executed by the Borrower, the Agent, the LC Issuer and the Lenders and each party has notified the Agent by facsimile transmission or telephone that it has taken such action.

IN WITNESS WHEREOF, the Borrower, the Lenders, the LC Issuer and the Agent have executed this Amendment as of the date first above written.

MATRIX SERVICE COMPANY

By: /s/ George L. Austin

George L. Austin

Notice Address: 10701 East Ute Street

Tulsa, OK 74116

Attention: George L. Austin

Chief Financial Officer

Telephone: (918) 838-8822 FAX: (918) 838-8810

JPMORGAN CHASE BANK, N.A.

(successor by merger to Bank One, N.A. (Main Office Chicago)) Lender, LC Issuer and as Agent

By: /s/ Hal E. Fudge

Hal E. Fudge, First Vice President

Notice Address: Mail Code TX1-2454

P.O. Box 655415

Dallas, Texas 75265-5415

Attention: Hal E. Fudge,

First Vice President

Telephone: (214) 290-2799 FAX: (214) 290-2740

INTERNATIONAL BANK OF COMMERCE,

successor in interest to LOCAL OKLAHOMA BANK, an Oklahoma Banking Corporation, formerly known as LOCAL OKLAHOMA BANK, NA

By: /s/ David G. Moore

David G. Moore, Senior Vice President

Notice Address: 3601 NW 63rd

Oklahoma City, OK 73116

Attention: David G. Moore,

Senior Vice President

Telephone: (405) 841-2966 FAX: (405) 841-2375

WACHOVIA BANK, NATIONAL ASSOCIATION

By: /s/ Steven Markunas

Steven Markunas, Asst. Vice President

Notice Address: 123 South Broad Street

14th Floor – PA1246 Philadelphia, PA 19109

Attention: Steven Markunas,

Asst. Vice President

Telephone: (215) 670-6637 FAX: (215) 670-6645

UMB BANK, N.A.

By: /s/ Richard J. Lehrter

Richard J. Lehrter, Community Bank President

Notice Address: 1437 South Boulder Avenue

Suite 150

Tulsa, OK 74119

Richard J. Lehrter, President (918) 295-2000 Attention:

Telephone: FAX: (918) 295-2020

WELLS FARGO BANK, NA (formerly known as Wells Fargo Bank Texas, NA)

By: /s/ Roger Fruendt

Roger Fruendt, Senior Vice President

Notice Address: 1000 Louisiana Street

MS T5001-047 Fourth Floor Houston, TX 77002

Attention: Roger Fruendt,

Senior Vice President

Telephone: (713) 319-1403 FAX: (713) 739-1076

INDEX OF SCHEDULES

Schedule "B" - Ratification of and Amendment to Subsidiary Guaranty

Exhibit "H" - Borrowing Base Certificate

EXHIBIT "H"

BORROWING BASE CERTIFICATE

MATRIX SERVICE COMPANY ("Borrower"), pursuant to that certain Credit Agreement dated as of March 7, 2003, among Borrower, Lenders and Agent, as amended by Amendment One to Credit Agreement dated as of May 22, 2003, Amendment Two to Credit Agreement dated as of August 27, 2003, Amendment Three to Credit Agreement dated as of December 19, 2003, Amendment Four to Credit Agreement dated as of March 11, 2004, Amendment Five to Credit Agreement dated as of May 6, 2004, Amendment Six to Credit Agreement dated as of August 5, 2004, Amendment Seven to Credit Agreement dated as of October 6, 2004, Amendment Eight to Credit Agreement dated as of November 30, 2004, that certain letter agreement dated March 23rd, 2005, that certain letter agreement dated April 8, 2005, and that certain Amendment Nine to the Credit Agreement dated as of April 22, 2005 (as amended, the "Credit Agreement"), among Borrower, Agent and the Lenders party thereto, hereby certifies to Agent and Lenders that the following Borrowing Base calculation is true and correct as of the close of business on 2005.

2.

ne cros	e of dusiness on, 20 :	
Tota	Accounts as of	\$
Less		
(a)	Invoices over 90 days old	\$
(b)	Contra Accounts	\$
(c)	Accounts owed by Account Debtors more than 10% of whose Accounts are more than 90 days old	\$
(d)	Accounts subject to a third party lien	\$
(e)	Accounts in Bankruptcy	\$
(f)	Affiliate Accounts	\$
(g)	Foreign Accounts other than Canadian Accounts	\$
(h)	Accounts owed to Foreign Grantors other than allowed Canadian Grantors	\$
(i)	U.S. Government Accounts not subject to appropriate assignment under Assignment of Claims Act	\$
(i)	Retainage Accounts	\$

	(k) Concentrations in excess of 25%	\$	
	(1) Other Ineligibles as defined by the Credit Agreement	\$	
3.	Subtotal of Eligible Accounts (Line 1 less 2 a-l)	\$	
	(a) Less an amount equal to, for all uncompleted contracts approved and accepted:		
	(1) All billings in excess of costs and estimated earnings, \$		
	(2) MINUS All costs in excess of billings under uncompleted contracts \$)		
	(1) above minus (2) above	(\$)
	(b) Amount in the aggregate in excess of \$10,000,000 from Accounts under bonded contracts	(\$)
4.	Restated Eligible Accounts (Line 3 less Lines 3(a) and 3(b))	\$	
5.	Borrowing Base (Line 4 x 80%)	\$	
6.	Lesser of (i) Line 5, (ii) \$35,000,000 or (iii) Aggregate Revolving Loan Commitment	\$	
7.	Revolving Loan Balance	\$	
8.	LC Obligations (Maximum of \$15,000,000)	\$	
9.	Total Revolver Outstanding (Line 7 + Line 8)	\$	
10.	Excess/Deficit Collateral Margin (Line 6 less Line 9) (Use Zero if Negative)	\$	
11.	Available Revolver Credit (Lesser of Line 10 or \$35,000,000)	\$	
12.	Number and dollar amount of outstanding checks		
	a. Number		
	b. \$		
13.	Revolver B Borrowing Base (Lesser of (i) \$10,000,000, (ii) amount notified by Revolver B Lenders, or (iii) as otherwise reduced)	\$	

14.	Revolver B Commitment (Line 13 less Line 12b.)	\$	
15.	Accounts Payable of Borrower and each Subsidiary more than 60 days past due.	\$	
16.	Line 4 less Line 15	\$	
17.	Line 16 x 80%	\$	
	MATRIX SERVICE COMPANY	-	

By: /s/ George L. Austin

George L. Austin, Chief Financial Officer



FOR IMMEDIATE RELEASE

MATRIX SERVICE COMPANY ANNOUNCES \$13.0 MILLION INCREASE IN AVAILABILITY UNDER SENIOR CREDIT FACILITIES

TULSA, OK – April 25, 2005 – Matrix Service Co. (Nasdaq: MTRX), a leading industrial services company, announced that under a Letter Agreement to its credit facility dated April 8, 2005, the revolving commitment is increasing from \$32.0 million to \$35.0 million. The credit facility has been amended and the revolving commitment now bears cash pay interest at a rate of prime plus 1% and accrued interest at 1% escalating fifty basis points monthly until capped at 5.0% after November 30, 2005. The term loan now bears cash pay and accrued interest at these same rates.

Matrix also announced that the amendment has established a \$10.0 million Revolving Loan B commitment. The Revolving Loan B commitment bears cash pay interest at prime and expires on October 31, 2005.

About Matrix Service Company

Matrix Service Company provides general industrial construction and repair and maintenance services principally to the petroleum, petrochemical, power, bulk storage terminal, pipeline and industrial gas industries.

The Company is headquartered in Tulsa, Oklahoma, with regional operating facilities located in Oklahoma, Texas, California, Michigan, Pennsylvania, Illinois, Washington and Delaware in the U.S. and Canada.

This release contains forward-looking statements that are made in reliance upon the safe harbor provisions of the Private Securities Litigation Reform Act of 1995. These statements are generally accompanied by words such as "anticipate", "continues", "expect", "forecast", "outlook", "believe", "estimate", "should" and "will" and words of similar effect that convey future meaning, concerning the Company's operations, economic performance and management's best judgment as to what may occur in the future. Future events involve risks and uncertainties that may cause actual results to differ materially from those we currently anticipate. The actual results for the current and future periods and other corporate developments will depend upon a number of economic, competitive and other influences, including those identified in the "Risk Factors" and "Forward Looking Statements" sections and elsewhere in the Company's reports and filings made from time to time with the Securities and Exchange Commission. Many of these risks and uncertainties are beyond the control of the Company, and any one of which, or a combination of which, could materially and adversely affect the results of the Company's operations and its financial condition. We undertake no obligation to update information contained in this release.

For More Information:

Les Austin Vice President Finance and CFO Matrix Service Company 918/838-8822 laustin@matrixservice.com

Investors:

Trúc N. Nguyen VP, Investor Relations Stern & Co. 212/888-0044 tnguyen@sternco.com



FOR IMMEDIATE RELEASE

MATRIX SERVICE COMPANY CLOSES \$30 MILLION PRIVATE PLACEMENT OF SENIOR UNSECURED CONVERTIBLE NOTES

TULSA, OK – April 25, 2005 – Matrix Service Co. (Nasdaq: MTRX), a leading industrial services company, announced that it has completed a private placement of \$30.0 million of 7.0% Senior Unsecured Convertible Notes due 2010. The notes are convertible into shares of common stock at a conversion price of \$4.69 per share. Interest will be payable in advance for the period to and including April 25, 2007 and is subject to adjustment upward or downward in certain circumstance. The net proceeds to Matrix will be approximately \$24.6 million. The Company intends to use \$20.0 million of the net proceeds to repay its Term Note B, which currently bears 18% interest and to use the remaining net proceeds to provide additional liquidity for working capital needs. The Company has agreed to file a registration statement with the SEC covering the resale of the shares underlying the notes within 30 days.

The notes and shares of common stock issuable upon conversion of the notes have not been registered under the Securities Act of 1933, as amended, and may not be offered or sold in the United States absent registration or an applicable exemption from registration requirements.

This press release does not constitute an offer to sell or the solicitation of an offer to buy any of the notes or the shares of common stock issuable upon conversion of the notes and shall not constitute an offer, solicitation or sale of these securities in any state in which such offer, solicitation or sale would be unlawful prior to the registration or qualification under the securities laws of any such state. This press release is being issued pursuant to Rule 135c under the Securities Act.

About Matrix Service Company

Matrix Service Company provides general industrial construction and repair and maintenance services principally to the petroleum, petrochemical, power, bulk storage terminal, pipeline and industrial gas industries.

The Company is headquartered in Tulsa, Oklahoma, with regional operating facilities located in Oklahoma, Texas, California, Michigan, Pennsylvania, Illinois, Washington and Delaware in the U.S. and Canada.

This release contains forward-looking statements that are made in reliance upon the safe harbor provisions of the Private Securities Litigation Reform Act of 1995. These statements are generally accompanied by words such as "anticipate", "continues", "expect", "forecast", "outlook", "believe", "estimate", "should" and "will" and words of similar effect that convey future meaning, concerning the Company's operations, economic performance and management's best judgment as to what may occur in the future. Future events involve risks and uncertainties that may cause actual results to differ materially from those we currently anticipate. The actual results for the current and future periods and other corporate developments will depend upon a number of economic, competitive and other influences, including those identified in the "Risk Factors" and "Forward Looking Statements" sections and elsewhere in the Company's reports and filings made from time to time with the Securities and Exchange Commission. Many of these risks and uncertainties are beyond the control of the Company, and any one of which, or a combination of which, could materially and adversely affect the results of the Company's operations and its financial condition. We undertake no obligation to update information contained in this release.

For More Information:

Les Austin Vice President Finance and CFO Matrix Service Company 918/838-8822 laustin@matrixservice.com

Investors:

Trúc N. Nguyen VP, Investor Relations Stern & Co. 212/888-0044 tnguyen@sternco.com